Coventry University



DOCTOR OF PHILOSOPHY

Optimal energy allocation for households in generation-constrained off-grid microgrids

Palencia, Gene Fe

Award date: 2022

Awarding institution: Coventry University University of San Carlos

Link to publication

General rights Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- · Users may download and print one copy of this thesis for personal non-commercial research or study
- This thesis cannot be reproduced or quoted extensively from without first obtaining permission from the copyright holder(s)
- You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

OPTIMAL ENERGY ALLOCATION FOR HOUSEHOLDS IN GENERATION-CONSTRAINED OFF-GRID MICROGRIDS

GENE FE P. PALENCIA



A thesis submitted in partial fulfilment of the University's requirements for the Degree of Doctor of Philosophy.

August 2020 – version 2.0



Gene Fe P. Palencia: *Optimal energy allocation for households in generation-constrained off-grid microgrids* © August 2020

ABSTRACT

For microgrids with limited generation capacity, allocating a daily equal energy budget to each household is one way of ensuring that all households are provided with sufficient energy to be used in a day without compromising the daily operation of the power system. With the same daily energy quota, households are given freedom on how to spend energy according to their preferences and priorities. Issues in this type of energy management scheme include 1) power outage in households that use up all their energy allowance before the scheduled replenishment and 2) unused energy allocation turned to waste from households that are unable to consume the energy allowance. Energy waste in terms of unused energy allocation of some households can be beneficial to other households. The unused energy can be distributed to other households that experience a power outage and need more energy than the allocated. One approach to solving the above issues is to frame the problem as an optimisation problem that aims to minimise the energy wastage and maximise energy availability.

This research proposes an optimal energy allocation for households connected to generation-constrained microgrids. The proposed optimal energy allocation scheme has two main parts. First, the ideal energy utilisation of each household is predicted using a multi-layer perceptron (MLP); secondly, the optimal energy allocations for the households based on their predicted utilisations are derived using Karush-Kuhn-Tucker (KKT) optimality conditions. From the application of the KKT conditions, a methodology for optimal energy allocation that is adaptive to each household is proposed. The approach is optimal as it minimises the energy wastage/deficit while maximising energy availability to households, and adaptive because it uses the household's historical data and demographic information.

To support the development of the MLP-based forecast model, this thesis implemented an energy monitoring system called Philippines Micro-Off-Grids (PMOG) system to gather the actual historical energy usage data in representative households from select villages in Cebu, Philippines which have microgrids with limited generation capacity. In the Philippines, there are 40 million people without access to electricity [Och13] and microgrids are used to provide electricity access to villages that are not accessible by the traditional grid. There are three villages selected with two of them being off-grid communities and one being grid-tied community. The energy data from PMOG system serves as the baseline data for the development of the forecast model and the optimal allocation scheme. A survey is also conducted to gather the household demographic information that affects their daily energy consumption.

This thesis presents an experimental method to determine the best combination of hidden layers and neurons of the neural network along with the input delay window in shaping the input variables that allows the forecast model to generate the lowest possible root mean squared error (RMSE). Since the optimal energy allocation is dependent on the accuracy of MLP-based load forecast model, the right combination of those design parameters of the neural network together with the delay window used in shaping the inputs is crucial. These parameters are considered as the main factors affecting the performance of the neural network in forecasting.

Experimental results show that as the households' demographic information is included as input variables secondary to the historical energy data and weather information, the performance of the neural network improves significantly. The RMSE decreases from 92 W to 81 W, which represents a 12 % decrease for a neural network with three hidden layers, 20 neurons and seven delays.

Given the limited generation capacity of the microgrid, the objective is to minimise the squared difference between the ideal utilisation of the household (which is estimated by the MLP-based forecast model) and the (calculated) allocated energy. Results from the data from the select villages show an aggregated unused or deficit energy per household (for a day) from the existing equal allocation can be reduced from 0.24 kWh to 0.11 kWh using the proposed dynamic/adaptive allocation, which is about 54 % reduction in unused or deficit energy. For 288 days, a total of 44 % reduction of energy wastage is achieved with the proposed methodology when compared with equal allocation, that is 112 kWh using equal energy allocation, and 62 kWh using the proposed optimal energy allocation.

In summary, the proposed approach of allocating the daily energy allowance of the household which is a hybrid approach using an MLPbased forecast model and KKT optimality conditions minimises the unused energy and enables households to maximise their energy usage without compromising the minimum energy requirement of each household in villages powered by microgrids with limited generation capacity. By incorporating household profiles as inputs to the MLPbased forecast model, prediction accuracy was improved by 12% in terms of RMSE. From my experiments, employing MLP-based forecast model ensures better forecasting performance than other techniques such as Autoregressive Integrated Moving Average (ARIMA), Radial Basis Function Network (RBFN) and Gaussian Process Regression (GPR). The overall average accuracy for the MLP-based forecast model is 91% with the highest accuracy of 93% for House 5 predictions, and the lowest is 91% for House 3.

This approach is expected to work on households with similar profiles connected to any off-grid power systems. Optimising the daily energy quota will enable the village to maximise the usage of the available energy with minimum wastage in terms of unused energy quota. This approach will also lead the village to have a better payment scheme based on their actual usage of electricity. To the memory of my beloved parents: my mother whose love is eternal and my father who instilled discipline in us. They are the greatest parents I could ever have wished for ...

I want to thank the following for helping me with my research project and PhD journey:

To my supervisory team, **Prof. James Brusey**, **Prof. Elena Gaura**, **and Dr Kojo Sarfo Gyamfi**, who are very supportive and understanding. Thank you all for the guidance and feedback throughout my PhD journey—also, thanks to **Dr Ahn Tran and Dr Ross Wilkins** who give so much advice during the first phase of my research.

To my colleagues and friends in Cogent Lab and Futures Institute, Alia P., James W., G. Chen, Ross D., Daniel B., Brandi J., Lorena M., Magdalena K., Zihao L., and Brandon B., thank you for the good laugh. All of you are brilliant. Cheers! To our staff and friends, Susan L., Kriti B., Nandor V., James B. and Leonardo D., thank you for the excellent talk. You guys are fantastic!

To my friend **Kim Ochea**, thank you so much for always being there whenever I need someone. You always make me smile and motivated. To my friends here in the UK and back home, thank you all for your constant support since day one. You put some spice on my PhD journey.

To my **family**, thank you for your support, love, and prayers. I love you all. To my **nieces and nephews**, thank you for the funny messages, calls and video chats. I am proud of you all. You are all amazing! My deepest gratitude to my sisters and best friends, **Liezel and Luchi**, who are always there for me. Thank you for being supportive of my academic expedition. To my parents, **Celestino S. Palencia, Jr. (Tatay "Aplod") and Lydia P. Palencia (Nanay Lydia)**, thank you. This PhD is for you. Wherever you are, I hope you are looking down on us with your beautiful smile.

I want to thank my funding institutions, **British Council, Philippines** and the **Department of Science and Technology - Science Education Institute (DOST-SEI)** of the Philippines for providing the financial aspect of this research through DOST – Newton PhD Scholarships. Also, I want to thank and acknowledge the support of my home institution, the **University of San Carlos** for allowing me to have a study leave for the entire duration of my PhD journey.

And last but not least, thanks to our almighty **God** for giving me this opportunity. I will not be where I am now if He has not allowed it. To God, be the glory!

CONTENTS

1	INT	RODUCTION	1
	1.1	Motivation of the research	1
	1.2	Research aim	4
	1.3	Research questions	5
	1.4	Structure of the thesis	7
2	LIT	ERATURE REVIEW	11
	2.1	Microgrid power systems	12
		2.1.1 Benefits and challenges of off-grid power systems	13
	2.2	Energy Management System (EMS)	15
	2.3	Significance of optimising usage	16
	-	2.3.1 Generation-constrained off-grid microgrids	18
	2.4	Energy management schemes	19
		2.4.1 Load shedding and load shifting	19
		2.4.2 Time-based operation	21
		2.4.3 Energy consumption quota	22
	2.5	Load forecasting techniques	24
		2.5.1 Time series forecasting	26
	2.6	ANN fundamentals	26
		2.6.1 Back-Propagation Neural Network	28
		2.6.2 Issues in designing neural network-based fore-	
		casting models	29
	2.7	Factors influencing load forecast model performance .	31
	2.8	User preferences	32
	2.9	Research data	33
	2.10	Chapter summary	34
3	MET	THODS	37
	3.1	Data Gathering	37
		3.1.1 Evaluation of the survey questionnaire and test-	
		ing of PMOG system before deployment	38
	3.2	Load forecast modelling	39
		3.2.1 Evaluation of the loaf forecast model	40
	3.3	Optimising energy allocation of the households	40
		3.3.1 Evaluation	41
	3.4	Chapter Summary	42
4	DAT	A COLLECTION AND PROCESSING	43
	4.1	Off-grid villages' profiles	44
		4.1.1 Village 1	45
		4.1.2 Village 2	45
		4.1.3 Village 3	47
	4.2	Development of survey questionnaire (SQ)	47
		4.2.1 Profiles	48
	4.3	Development of PMOG system	50
		4.3.1 PMOG system flow	50
	4.4	Deployment of the survey and PMOG systems	51
		4.4.1 Field survey: Household face-to-face interview	51

		4.4.2	PMOG system installation	52
	4.5	Data p	processing	55
	4.6	Descri	ption of data	58
		4.6.1	Energy data yield	59
		4.6.2	Description of survey data	63
	4.7	Energy	y usage of the households	67
	4.8	House	hold profiles	68
	4.9	Chapte	er summary	69
5	MUL	TILAY	ER PERCEPTRON-BASED FORECAST MODEL	71
	5.1	Findin	g the best combination of neural network param-	
		eters .	· · · · · · · · · · · · · · · · · · ·	73
		5.1.1	Steps in determining the best combination of	
			hidden layers HL, number of neurons N, and	
			delay D?	73
		5.1.2	Input shaping (delay D) and the effect of the	
			network size on the performance of the model	
			in forecasting	75
		5.1.3	Standard load forecasting using historical energy	
			data as input	77
	5.2	Improv	ving the performance of MLP-based forecast mode	1 80
		5.2.1	Adding temperature as input	80
		5.2.2	Adding weekdays as input	81
		5.2.3	Adding demographic information as input	81
		5.2.4	How does the number of hidden layers HL af-	
			fects the performance of the network in load	
			forecasting?	83
	5.3	Discus	ssion of results	89
	5.4	Evalua	ation and comparison of results	94
	5.5	Experi	imental results using other forecasting techniques	96
		5.5.1	Gaussian Process Regression	96
		5.5.2	Radial basis function neural (RBFN) networks	97
		5.5.3	Autoregressive integrating moving average	
			(ARIMA)	99
	5.6	Comp	arison of results	99
	5.7	Chapte	er summary	100
6	OPT	IMAL I	DAILY ENERGY ALLOCATION	103
	6.1	Proble	m description	104
		6.1.1	Solving the optimisation problem using KKT	
			approach	106
		6.1.2	Steps in solving the optimal energy allocation .	110
		6.1.3	Verification of the optimal solution	110
		6.1.4	Validating the optimisation results using the	
			forecasted results	112
		6.1.5	Energy threshold and its effect on the optimal	
			energy allocation	115
	6.2	Discus	ssion of results	117
		6.2.1	Optimal and adaptive energy allocation: con-	
			sumer profiles	118
		6.2.2	Energy efficiency under the limited generation	
			capacity	120

	6.3 Chapter sum	nmary	124
7	CONCLUSION		127
	7.1 Answers to 1	research questions	129
	7.2 Contributior	ns to knowledge	132
	7.3 Significance	of the study	133
	7.4 Future work		133
Α	DEPLOYMENT DO	DCUMENTS	135
	A.1 PMOG syste	em raspberry pi set up guide	135
	A.2 PMOG syste	ems installation manual	139
	A.3 Survey Ques	stionnaire	141
	A.4 Deployment	Technical Report	146
В	FIELD WORK: PM	IOG SYSTEMS INSTALLATION	147
С	RESEARCH DATA		151
D	ETHICS APPROVA	AL	153
	BIBLIOGRAPHY		159

LIST OF FIGURES

Figure 1.1	Structure of the thesis	8
Figure 2.1	Energy management system (EMS) [GG15]. Legend: LC - Local controller	16
Figure 2.2	Architecture of EMS control A.) centralised and B.) decentralised illustrated by Espín-Sarzosa	10
T '	et al. [EPBN20] <i>Legend: LC - Local controller</i>	17
Figure 2.3	an input layer, a hidden layer, and an output	
	layer	28
Figure 3.1	Flow of methods used in this research	38
Figure 4.1	Village 1 is located in a remote island in Lapu- lapu city, Philippines. Houses are typically	
	made of light materials such as bamboos,	16
Figure 4.2	Village 2 is a community whose houses are	40
1.901.0 4.2	built by the Red Cross, Philippines in cooper-	
	ation with the Local Government Unit (LGU)	
	of Daanbantayan, Cebu for the survivors of ty-	
	phoon Yolanda. The community has an off-grid	
	solar power that allocates 0.80 kWh as the daily	
	energy quota for each household.	47
Figure 4.3	Village 3 is a community built by Human	
	of the LCU of Deephantayan Coby The	
	community is donated to the survivors of	
	typhoon Yolanda. The community is connected	
	to the conventional grid.	48
Figure 4.4	PMOG System Architecture. The PMOG sys-	1
-	tem is designed to collect energy data on both	
	household and appliance levels using CT jaw	
	sensor and IAMs, respectively. Energy data is	
	stored locally and pushed through to the re-	
T .	mote server.	52
Figure 4.5	PMOG Data Flow. Each nousehold has 1 PMOG	
	sors sense the total energy consumption of the	
	households as they are connected to the main	
	panel and communicate with the local server	
	RPi for storage. Each IAM monitors the energy	
	of the assigned appliance into it and sends the	
	data to the local server through wireless trans-	
	mission	53

Figure 4.6	Face-to-face interview with the respondents in one of the villages. Participants are asked to rank and give scores to the 10 items as enu- merated in Section 4.2.1.3 according to their priorities.	54
Figure 4.7	PMOG systems deployed in the villages col- lecting energy data on both the off-grid power systems (generation side) and the households with individual appliance monitors (IAMs)	56
Figure 4.8	PMOG system node report interface. The web- page displays the newest transmitted data to the remote server. The data includes the date and time, device number, and PMOG number aside from the main data which are the tem- perature and power usage of the household.	57
Figure 4.9	PMOG system data yield. The daily yield of the data in % (y-axis) for each household (x-axis). The daily average of yield (%) is above 80% for all PMOG systems installed in the households.	60
Figure 4.10	Actual electricity usage pattern of house 1 from Village 1. House-level power usage (y-axis) varies as the usage of TV and electric fans change reflecting the power usage of each appliance over time (x-axis). Data: 1 day (5 hours) electricity usage.	61
Figure 4.11	Households energy usage pattern. House level power usage (y-axis) of the selected households in village 1 varies in time (x-axis) when the load is changed. Most of the households use electricity as soon as the power system is up. Some households maintain almost the same load throughout the operation of the power system (e.g., House 3); thus very few variations on power usage is reflected	61
Figure 4.12	Daily energy usage pattern. Power usage (y- axis) at the house level varies significantly for the entire day (x-axis) for each household. Most of the households peak their usage of electricity around 6:00 PM to 0:00 PM (dipper time)	62
Figure 4.13	Monthly income of the 3 villages compared with the poverty and food thresholds	65
Figure 4.14	Average energy usage of the households in Vil-	68
Figure 5 1	Load forecasting model development flow	72
Figure $= 2$	K-Fold Cross Validation	74 76
Figure 5.3	Input delay window concept where DW_1 is delay window 1, DW_2 is delay window 2 and DW_m is the total number of delay windows for	70
	the given input series.	77

Figure 5.4	RMSE (Wh) results where x-axis shows the de- lay D and y-axis is the RMSE (Wh) for each	
Figure 5.5	delay D showing variations of a k-fold validation. Time needed to train the model for each delay, D with the corresponding neurons per layer N for neural network with $HL = 1$. X-axis is the range of delay D and X-axis is the time in	. 78
	minutes m	84
Figure 5.6	Time needed to train the model for each delays, <i>D</i> with the corresponding neurons, <i>N</i> . X-axis is the range of D and Y-axis is the time in minutes,	
Figure 5.7	<i>m</i> . Time needed to train the model for each de- lays, D with the corresponding neurons, N with HL = 5. X-axis is the range of D and Y-axis is	85
Figure 5.8	the time in minutes, <i>m</i>	86
Figure 5.9	Training time for neural network with 20 neurons per layer N and varying hidden layer HL for each delay D with energy, temperature, weekdays and the demographic information	00
Figure 5.10	as inputs	88
Figure 5.11	Regression plot of the neural network for load	09
Figure 5.12	MLP-based forecast model performance. The actual energy usage is almost the same with the predicted values with the neural network's MAPE of 6.2 % and RMSE 93 Wh and standard deviation of on	74
Figure 5.13	Prediction results using MLP-based forecast	92
Figure 5 14	Prediction performance of GPR model	94
Figure 5.15	Daily predicted values of energy consumption	71
Figure 6.1	(y-axis)	99

Figure 6.2	ANOVA results for the 3 allocations (Equal,	
	proportion, and optimal), with 288 days data.	
	The difference between the means of the 3	
	groups are statistically significant with a p-	
	value of 2.24×10^{-38} and F–value is 88.9	122
Figure 6.3	Anova test results between the two allocations	
	with $N = 288$. The means of the two groups	
	have statistically significant differences with p-	
	value of 6.61×10^{-31}	123
Figure 6.4	Absolute percent error (APE) for each house-	
	hold using the proposed optimal energy al-	
	location. The MAPE for each household are	
	6.2, 8.0, 5.8 and 10.4 for H1, H2, H3 and H4, re-	
	spectively	123
Figure B.1	PMOG systems deployed in the villages col-	
	lecting energy data on both the off-grid power	
	systems (generation side) and the households	
	with individual appliance monitors (IAMs)	148
Figure B.2	PMOG system node report interface. The web-	
	page displays the newest transmitted data to	
	the remote server. The data include the date	
	and time, device number, and PMOG number	
	aside from the main data which are the tem-	
	perature and power usage of the household	150

LIST OF TABLES

Table 4.1	Village profiles: The 3 villages have different levels of access to electricity. Village 1's access to electricity is limited to a 5-hour window, Village	
	Village a is connected to the conventional grid	
Table 4.2	Summary of installed PMOG systems in Village	44
	1 and Village 2	54
Table 4.3	Descriptive summary for the households daily	
	energy usage	58
Table 4.4	Statistics of households temperature data	59
Table 4.5	Daily yield of PMOG systems	60
Table 4.6	Household composition: Age, household mem-	
	bers working and attending school (%)	63
Table 4.7	Characteristics of household head and spouse,	
	and average number of occupants per household.	64
Table 4.8	Monthly income (range) of the households	65
Table 4.9	Percentage (%) of Households, by occupation	5
. /	of the HHs head	66
Table 4.10	Household profiles	69
•	L	

Table 5.1	Average RMSE (Wh) and the standard devia-	
5	tion as D increases for each neurons of neural	
	network with 1 hidden layer	79
Table 5.2	Average RMSE (Wh) and its standard deviation	
5	as the delay D increases for each neurons of	
	neural network with 3 hidden laver	79
Table 5.3	Average RMSE (Wh) and its standard deviation	1)
	as the delay D increases for each neurons of	
	neural network with 5 hidden laver	80
Table = 4	Average RMSF (Wh) and its standard devia-	00
14010 9.4	tion as D increases for each neurons of neural	
	network with 1 hidden layer for energy and	
	temperature data as inputs	81
Table	Avorage RMSE (Wh) as D increases for each	01
Table 5.5	Average RWDE (WII) as D increases for each	
	for anorgy temperature and weekdays as inputs	80
Table = (Assessed as DMCE (M/b) as the delay D in grades	02
Table 5.6	Average RNISE (WR) as the delay D increases	
	for each neuron per layer N of neural network	
	with 1 hidden layer HL for energy, temperature	
	and weekdays and the demographic informa-	0
	tion as inputs	83
Table 5.7	Average RMSE (Wh) as D increases for each	
	number of neurons with 3 hidden layer for en-	
	ergy, temperature, weekdays and the demo-	
	graphic information as inputs	84
Table 5.8	Average RMSE (Wh) and standard deviation	
	of neural network with 5 hidden layer with	
	energy, temperature, weekdays and the demo-	
	graphic information as inputs for each neuron	
	(N) and delay (D)	86
Table 5.9	Training time as the delay D increases for each	
	neurons per layer N of neural network with 5	
	hidden layer for energy, temperature, weekdays	
	and the demographic information as inputs	87
Table 5.10	Performance of the NN as D increases for each	
	hidden layer for neural network with $N = 20$	
	and with energy, temperature, weekdays and	
	the demographic information as inputs	87
Table 5.11	MAPE and absolute percentage error between	
0	the actual and forecasted energy usage of the	
	households.	93
Table 5.12	Average forecasting accuracy of the MLP-based	10
2	load forecast model for households in village 2	94
Table 5.13	Comparing the performance of the proposed	21
	MLP-based load forecast model to the existing	
	models.	95
Table 5.14	RBFN RMSE Results	98
Table 5 15	Comparison of performance of the forecast model	00
10010 J.19	companyon of performance of the forecast model	フフ

Descriptive summary of the proposed optimal	
daily energy allocation (kWh) for each house-	
hold in Village 2	118
Multiple comparisons of the mean of the	
proposed optimal energy allocations for each	
household	119
Comparison of ideal energy usage and the en-	
ergy allocations (equal, proportion, and opti-	
mal) of the households	121
Results for the objective function J for 288 days	122
	Descriptive summary of the proposed optimal daily energy allocation (kWh) for each house- hold in Village 2

NOMENCLATURE

D	Number of delays or the window of the inputs considered in reshaping historical energy data, in days
Eg	Generated energy supply by the off-grid power system, kWh
E ^a	Optimal energy allocation for each household, kWh
Ei	Ideal energy usage as predicted by the ANN-based forecast model, kWh
E _{min}	Daily energy threshold for each household, kWh
Eι	Second smallest amount of energy after E _s using ANN-based forecast model, kWh
E _s	Smallest amount of energy predicted using ANN-based forecast model, kWh
E ^a	Smallest amount of energy allocated to a household, kWh
g(x)	Equality function
h(x)	Inequality function
HL	Number of hidden layers of neural network
j	Index of the households separated from households with s or l indices.
l	Index of the household with the second smallest ideal energy demand $E_{\mathfrak{i}}$
L	Langrangian function

minJ	Nonlinear optimisation problem
n	Total number of households
Ν	Number of neurons of neural network
P _n	Number of historical energy data, in days
μ	Complementary slackness
s	Index of the household with the smallest amount of energy ideal energy demand E_{i}

ACRONYMS

AUC	Area Under Curve
ANOVA	Analysis of Variance
ANN	Artificial Neural Network
ARIMA	AutoRegressive Integrated Moving Average
BEC	Building Energy Consumption
BPN	Backpropagation Neural Network
CBR	Case-Based Reasoning
DG	Distributed Generation units
ECQ	Energy Consumption Quota
EMCS	Energy Management Control System
EMS	Energy Management System
EmS	Energy Monitoring System

ES	Exponential Smoothing
FFNN	Feedforward Neural Network
HVAC	Heating, Ventilation, and Air Conditioning
KKT	Karush-Kuhn-Tucker approach
k-NN	k-Nearest Neighbour (k-NN)
LTLF	Long-Term Load Forecasting
MAPE	Mean Absolute Percentage Error
MDP	Markov Decision Process
MSE	Mean Squared Error
MTLF	Medium-Term Load Forecasting
MLP	Multi-Layer Perceptron
MA	Moving Average
PMOGS	Philippines Micro-Off-Grid System
RBFN	Radial Basis Function Neural Network
RMSE	Root Mean Squared Error
STLF	Short-Term Load Forecasting
SVM	Support Vector Machines

INTRODUCTION

Electricity access is one of the problems faced by many developing countries, such as the Philippines. According to the World Bank, there are 1.2 billion people in the world without access to electricity, and most of them are in Asia [DW17]. Areas with no or limited access to electricity are mostly located in rural or remote regions of the countries. In the Philippines, there are approximately 2.3 million households that do not have access to electricity [IRE17]. While conventional electricity providers aim to connect every village and community in the country, they can be restricted by accessibility and geographical locations for some island communities and remote villages. Remote areas where households are dispersed and have meagre income are often less prioritised or left with no access to electricity for years. Microgrid power systems have been utilised to address this issue [IRE17; Int14].

Microgrid power systems can provide electrification to villages or communities where traditional grids are not economically feasible. Microgrid power systems are small-scale versions of a traditional grid which can operate in both grid-tied (on-grid systems) or stand-alone (off-grid systems) modes of operation. Most microgrid power systems use renewable energy resources available locally to the community such as water for mini-hydropower systems, wind for wind turbines, solar energy for solar power systems and biomass or agricultural wastes for biomass power plants [IRE17; HA12; Ene12]. Back-up power sources such as batteries and diesel generators are used in case of system failures [Int14].

1.1 MOTIVATION OF THE RESEARCH

Access to electricity for remote areas in The Philippines has been a great challenge. Remote areas, outside the reach of the traditional grid providers, opt to have an off-grid power system to provide their electricity needs [Ene12; RS16]. In rural communities and villages, off-grid power systems are designed and implemented in order to accommodate the basic electricity needs of each household. In a typical Filipino household, basic electricity needs consist of a television, an electric fan or ceiling fan, lighting bulbs, and a radio [Phi13].

While off-grid power systems are expected to provide the basic electricity needs of each household all the time, they are also expected to

do so without having to restrict the amount of energy that can be used by each household [Uni12]. However, most off-grid power systems are operating with restrictions such as load shedding and load shifting, time-based operation and daily energy quota in providing electricity to the consumers because of its limited generation capacity [Ene12]. Load shedding allows the power system to offload some of its load when the threshold for safe operation is breached [Mar+16]. This approach is used to save the power system from system failure due to overloading. The consumers have no control over their electricity access. The power system shed off parts of its distribution whenever necessary [Xu+16]. Load shifting is to defer the operation time of the load when demand is higher than the available energy at the time. Shifting is done based on load priorities [EAR20]. The implementation of load shedding or load shifting as part of the energy management system of microgrids is too costly. This makes it impractical to implement in off-grid power systems. Time-based operation is used when the power system can not operate for 24 hours daily due to technical issues [ARP16]. To save the power system from breakdown, a time-based operation is employed. The system will operate in a specific period in a day. Only at that time, customers will have access to electricity [Mar+13].

Daily energy quota is one of the common energy management scheme employed in off-grid microgrids with limited generation capacity [Sun+16]. Dwellings are given a daily energy quota to be used in a day [ZXT12]. This scheme is employed to ensure that each household will have sufficient energy for their basic electricity needs. The advantage of this approach is that the households control their energy usage, but they can not use any more than to what is being provided. Restrictions on household appliances are applied to maintain the power systems operation and prevent systems failure due to overloading.

For example, consider the Red Cross village located in Daanbantayan, Cebu, Philippines, the village is powered by community-based off-grid solar power systems with an aggregate generation capacity of 119 kW, serve 128 households including a community livelihood centre, and a day-care centre [Han15]. The solar power systems of the village are not capable of supplying the estimated peak power demand of 2 kW for each household without energy usage restrictions to the households and without compromising the operation of the whole off-grid power system. These peak demands are estimated to occur at different times according to household profile. Hence, the energy management system of the solar power systems was then designed to allocate an everyday power allowance to each household equivalent to 0.8 kWh for 24-hour use at their own times, in such a way that the

3

total generation capacity is not exceeded. When the allocated energy is surpassed, the system would automatically cut off the connection of that household from the solar power system, and the energy quota would be restored the subsequent day. The solar power system was designed by Technician without Borders, a French non-government organisation, that helps install and implements the solar power systems in the village [Han15]. Restrictions on the power consumption of each household are imposed to cater to the needs of the whole community while ensuring a steady operation of the off-grid power system. Implementing such restrictions with fixed monthly fees is necessary to provide the basic electricity services to each household.

With an equal amount of energy allocated daily and considering the limited generation capacity of the off-grid power system, two main drawbacks can be highlighted:

1.) Some households use up the allocated daily quota and experience power outage but may require more energy than what is currently provided to them.

2.) Some households do not use all the allocated energy daily and pay the same fixed monthly tariffs as the households that use all their allocated energy.

The two issues above are based on the assumption that all households have equal energy requirements and that the basic electricity needs of the households do not change. This assumption is flawed since the household's electricity needs can vary through time. These variations depend on several factors, such as the household's size, total monthly income, household's head occupation, household's head education, and the number of children who are attending school and staying at home. Furthermore, the fact that some households experience power outage almost every day may indicate that those households require more energy than what is allocated and those households that do not use up all the energy allocation daily might require less energy than the daily quota. All household is required to pay a fixed amount of monthly fees even though their daily energy allocation is not fully utilised. Paying a fixed amount for unused energy is unfair.

Increasing the generation capacity can resolve the issues with the limited energy allocation to each household. However, this approach is only appropriate if the community has the necessary finances to pay the cost of increasing the capacity of the off-grid power systems. Since increasing the generation capacity is not feasible without the necessary finances, one way of resolving the above drawbacks is to adaptively allocate energy to different households based on their expected energy usage so that billing can be based on the actual amount of energy used. The adaptive energy allocation scheme considers not only the basic loads of the households but as well as the individual household profiles. This way, each household would have the freedom to control how they spend their energy without worrying about a shortage, or not using all their energy allocation. The unused allocated energy is considered to be a waste that could have been useful to other households that needs more energy.

All conditions mentioned above are similar to any households in remote areas connected to microgrids which the generation capacity is not designed with the demand of the households but with the available resources for the implementation power systems. In villages such as the Red Cross Village which are built for people who suffers natural calamities, the issues surrounding electricity access are similar [IRE17].

Hence, this research proposes a new method of allocating the daily energy allowance of household in off-grid villages such as the Red Cross village in Daanbantayan, Cebu. Specifically, this research investigates a way of optimising the allocation of energy proportional to the consumption of each household with the following considerations:

- Case 1: If the energy allocation is more than the basic needs for a given household, they pay for something they do not use. Therefore, they want an allocation commensurate with their usage.
- 2.) Case 2: If the allocation is less than the desired utilisation for a given household, they would be dissatisfied with the amount of energy they have, as they would experience a shortage every day. Therefore, they want to have an energy allocation that is within their usage.
- 3.) The allocations in cases 1 and 2 are not guaranteed to be same.
- 4.) Energy allocation is constrained within the limited generation capacity of the off-grid power system.

1.2 RESEARCH AIM

Given that there are remote communities that are powered by a limited supply of electricity from an off-grid power system, with no capability of increasing the generation capacity of the power system, this research aims to provide a solution in maximising the energy usage of the available energy from limited supply and satisfy the basic electricity needs of the households.

To achieve this aim, an optimised energy allocation is proposed based on the forecasted energy usage of the households with constraints from the limited generation capacity of the microgrid, and the minimum energy threshold for each household. A Multilayer perceptron (MLP)-based load forecast model is developed that uses the individual household profile along with the historical electricity usage and temperature as its inputs to predict the energy that is likely to be consumed by the household a day-ahead. The proposed optimal energy allocation is limited to a dataset gathered from a village in Daanbantayan, Cebu, Philippines. Amongst the forecast models, the most widely used forecasting model is the Artificial Neural Network (ANN) [BGK15; Dud16; Moo+19]. One of the advantages of neural networks is their ability to generate a general map between inputs and outputs and does not require a priori knowledge that is needed in conventional statistical and econometric modelling. When using a neural network, the input data does not need to satisfy assumptions that are required in other statistical methods. Neural network-based forecast model is known for being robust in handling real-world data.

Load forecasting is essential for efficient energy management operation. When the generation capacity of the system is not capable of providing the desired energy of the households, forecasting the energy consumption is important to ensure maximum usage of the available energy and also to provide the basic electricity needs of the households whilst maintaining a balanced operation of the power systems.

1.3 RESEARCH QUESTIONS

To achieve the above aim in the context of microgrid off-grid power systems, this study investigates the following research questions:

1. Can the household's daily energy consumption be forecast with reasonable accuracy? For this research, optimising the daily energy allowance of each household requires a forecast model that can estimate the next-day energy consumption of the household to minimise the energy wastage in terms of unused/deficit energy. When a household is using a lamp with a 10 W power rating, this lamp consumes 100 Wh energy for ten hours of use. When another household uses an electric fan with a 60 W power rating, this appliance will consume 120 Wh for two hours of use. The forecast model should predict as close as the expected usage of energy of each household to provide the households energy needs as accurate as possible. Since the performance of the load forecast model affects in optimising the daily energy al-

location of each household, a reasonable accuracy in forecasting is desired. The goal is to determine the combinations of network parameters such as the hidden layer HL, and neurons N and delay window D for historical data points that would produce the least root-mean-squared error (RMSE). The aim is to have RMSE that is equal or less than 120 Wh, a 15% tolerance from the current daily energy allocation of the selected village. The tolerance is derived as the minimum allowable error in daily energy allocation of the household. According to the work of Hsiao [Hsi15] and Moriano et al. [Mor+16], 20% or less is an acceptable error difference in load forecasting in terms of RMSE or 10% or less in terms of mean-absolute-per cent error (MAPE) [Mor+16]. Answers to this research question are presented in Chapter 5.

2. Can knowledge of consumer profiles aid in optimal and adaptive energy allocation?

Economic factors such as type of consumer, price of electricity, and demographic information of the households such as household income and the number of occupancies, are major factors affecting energy consumption [Hsi15; Che17; Har+15]. The performance of the load forecast model can be improved by incorporating consumer profiles as inputs. An MLP-based forecast model is developed with consumer profiles as inputs along with the historical energy usage, and temperature. Integrating the consumer profiles as inputs in the forecast model makes it possible to predict individually the daily energy consumption of the households using an adaptive model for the whole community. Chapter 6 presents how the consumer profiles to aid in achieving an optimal and adaptive energy allocation.

3. Can the energy allocation be optimised to improve energy efficiency under the limited generation capacity?

Energy efficiency can be improved by providing an optimal and adaptive energy allocation to each household through forecasting using ANN-based forecast model and calculating the optimal daily energy allocation using Karush-Kuhn-Tucker (KKT) conditions given the limited generation capacity of the off-grid power system while ensuring the basic electricity needs of the households. KKT approach is used as the constraints of the optimisation problem meets the requirements of the KKT conditions. In this research, energy efficiency is defined as the adaptive energy allocation to reduce energy wastage by households that use less energy, or to provide more energy to those households that have higher needs. Energy wastage refers to the unused energy allocation of some households that could be redistributed to other households that needs more energy than to what is being provided.

This research aims to allocate each household some energy that is as close as possible to their ideal utilisation. This measure of closeness can be defined by the Euclidean distance (a measure of how far the proposed energy allocation for the household deviates from their actual energy consumption). Consequently, for all households, the aim is to minimise the summation of all Euclidean distances, or equivalently, the squared Euclidean distances, under the constraint that the sum of all allocations do not exceed the total generation capacity of the solar system and each allocation is not less than the energy threshold for all households.

Section 6.2 of Chapter 5 answers this question by comparing the difference between optimal energy allocation and the ideal energy usage as predicted by ANN-based forecast model and the difference between fixed energy allocation and the ideal usage. With the proposed optimal energy allocation, the difference between the optimal energy allocation and the ideal usage of energy is minimised. With optimal allocation, the ideal electricity needs of each household are met, and the usage of the available energy generated by the off-grid system is maximised.

1.4 STRUCTURE OF THE THESIS

This chapter describes the research questions and aim of this research work and presents an overview on the proposed optimal energy allocation to each household that would allow dynamic energy allocations per day based on the forecasted energy consumption.

The rest of the thesis, as shown in Figure 1.1, is structured as follows. Chapter 2 presents the related literature on load forecasting techniques such as Artificial Neural Networks (ANNs), Gaussian Process Regression (GPR) and Autoregressive Integrated Moving Average (ARIMA) in time series. Approaches for optimising the energy allocation in remote communities with off-grid power systems to maximise the usage of generated electricity are also discussed. The gaps in the existing literature on energy allocation are identified and explained in the context of the off-grid communities.



Figure 1.1: Structure of the thesis

Chapter 3 presents the methods used in this research from data gathering to calculating the optimal energy allocation of the house-holds.

Chapter 4 details the design and the deployment of an energy monitoring system which is referred to as Philippine Micro-Off-Grid (PMOG) system. PMOG systems are used to gather the electricity consumption data of the selected households. In this chapter, the survey conducted to collect data about the off-grid communities is also discussed, as well as the design of the survey. The survey questionnaire was evaluated by conducting a pre-survey to check the reliability of the questionnaire. PMOG systems are evaluated by allowing the system to run for almost 2 months before the actual installation in the selected households.

Chapter 5 describes the load prediction model for each household based on the historical electricity usage and the profile of the individual household. In this chapter, the modelling of the MLP-based forecast model and its experimental results are presented. The performance of the forecast model is evaluated in terms of RMSE and MAPE. The results were also compared to the performance of other forecasting techniques such as ARIMA, RBFN, and GPR.

Chapter 6 presents the mathematical solution of the optimisation problem of the proposed optimal energy allocation based on the forecasted energy consumption of each household, as discussed in chapter 5. The methodology for this allocation is presented in this chapter. The results were evaluated by comparing to the other energy allocation scheme as presented in Section 6.2. Analysis of Variance (ANOVA) is used to determine if the means between the proposed optimal energy allocation and the other two techniques (equal allocation and allocation by ratio and proportion) are statistically significant.

Chapter 7 then concludes the work of the thesis and discusses possible directions for future work.

This chapter presents the theoretical background for load forecasting and the existing related literature for this work. Energy management systems for microgrids, energy allocation, and load forecasting are the focus of this chapter. Energy allocation is one of the many ways of addressing the limited generation capacity of the microgrids in remote communities. This chapter discusses the current works on different techniques for energy allocation and presents applications of different load forecasting techniques. The chapter also discusses the considerations in selecting the forecasting technique used in this research.

This chapter presents the gap in the literature that this research work is trying to address, and the existing literature on forecasting techniques and optimisation applied in energy management systems that made a good change in improving the performance of a power system.

This chapter discusses the following major topics that are relevant to the work of this thesis:

- Microgrids and its energy management systems
- Challenges in remote communities with generation-constraint microgrids
- Energy allocation schemes
- Load forecasting techniques

Off-grid microgrid systems with limited generation capacity distribute the available energy to the consumers through energy management schemes such as equal energy allocation daily, load shedding, load shifting, or time-based operation [IRE17]. With limited generation capacity, off-grid microgrids may not be able to supply the demand of the consumers [Ene12]. For most of the off-grid power systems, equal daily energy allocation is practical and the simplest to implement among others [Ene12]. However, this approach does not ensure maximum utilisation of the available energy, and with limited daily energy quota or allocation, households experience power outage before replenishment of the quota [Pal16]. On the other hand, some households do not use all the energy given to them daily but pay the same amount as other houses that use all of their allocated energy daily [Pal16]. These issues motivate this research.

2.1 MICROGRID POWER SYSTEMS

A microgrid consists of generation units, such as wind turbines, solar panels, mini-hydro plants, diesel generators, and electricity generators that use biomass, power lines for distribution and transmission, power control systems, and power loads [ZB16]. Microgrids are a compact type of electrical power infrastructure designed to have better efficiency, reliability, and integration of renewable energy sources than the traditional power grid[Kua+16]. A microgrid has two modes of operation: grid-connected or online and islanded or off-line (referred to as off-grid power systems) [KI06; HD14]. When operating in gridconnected mode, the microgrid provides support to the main grid by drawing power from the localised source, and provide the energy demand of the consumers whenever necessary. The main grid provides the deficit power in the microgrid when needed [HB12]. When a crisis occurs, and the main grid operation is interrupted, microgrids supply the power temporarily and operate in isolated mode to maintain the operation of the grid [Zhe+11]. In this operation, the microgrid must operate separately from the main grid and maintain the integrity of the process. The process refers to as the off-line mode of operation of the microgrid [Xu+17]. In an off-line mode of operation, the generated power must be in balance with the demand of the local loads [Oli+14]. Microgrids are typically implemented in areas where a conventional grid is not possible. These areas are usually in remote places far from the urban setting. Although, a microgrid is used as a backup power source to the traditional grid when power crisis occurs, providing electricity to remote areas with renewable energy resources gains more impact than as being an alternative source of energy for the main grid [IRE17].

Microgrids operate permanently in off-line or stand-alone mode when implemented in the remote areas where the conventional grid is not feasible because of economic or technical constraints. These microgrids are commonly referred to as the off-grid power systems that usually use renewable energy (RE) resources, such as water (minihydropower), sun (solar power), wind (wind energy), biomass and other RE resources available locally. Localised microgrids can lead to tremendous opportunities to increase power system efficiency, sustainability, and reliability [ZDM12].

For off-grid power systems, maintaining a balanced operation can be difficult, especially when the generation capacity of the system is limited [AR20]. Power systems are required to operate with a reliable operation between the generation and demand side. Power systems are in balanced operation when the generation side generates sufficient energy for the demand of the consumers [MSS20]. Energy management system (EMS) controls the operation of the power systems. The EMS monitors the status of each generation plant, and each consumer connected to the plant. A good EMS can determine the next load or energy consumption of each consumer and generates energy according to the demand [Kip+20].

For an off-grid power system with a limited generation capacity, to meet the demand of the consumers while maintaining proper operation of the power system, energy allocation approaches are deployed such as load shedding [SSP11; Gu+14; Moh+18], daily energy allocation (quota based) [ZXT12; LL18] and time-based operation [Hu+17; Shu+19; Xu17]. Section 2.4 discuss the details about each approach.

Some energy providers employed a smart EMS to control the operation of the system to avoid unnecessary operational cost due to excessive generation of energy over what is demanded by the consumers [LC16].

This research proposes a new scheme of allocating the daily energy allowance of each household connected to the off-grid power systems with a limited generation capacity to avoid the above problems.

2.1.1 Benefits and challenges of off-grid power systems

There are several benefits of having off-grid power systems than a conventional grid. As a localised grid, implementation of an off-grid power system is cheaper than a traditional grid as off-grid power system uses renewable energy resources that are available locally like water for a hydropower plant, and sun for the solar plant [IRE18]. Other implementation uses biomass such as rice hay and animal wastes in areas where biomass are abundant. Furthermore, localisation of microgrid can provide jobs to the community [Kir+o9].

According to the study of Meng et al. [Men15a], the following are the benefits of having microgrids

- 1) better power quality and reliability in case of a power outage;
- economic advantages for microgrids that uses renewable energy (RE) resources, such as the wind and solar energy which have low carbon emission;
- 3) minimum cost for transmission infrastructure (for localised RE)

These advantages can be classified as economic, environmental, and technical benefits [Rol11].

For remote areas, it is assumed that having access to electricity will lead to an increase in the economic condition of the users [IRE18].

The absence of a reliable energy source or electricity can hinder their economic progress [IRE17]. Access to energy or electricity can give to the people in remote communities a way to meet their basic needs such as home lighting [Int14; IRE17]. Moreover, if the generated power will be used by the farmers for their farm activities and small businesses in the local community to be more productive, economic progress is expected [Man+16].

The challenges include technology transfer to the community, generation capacity, and maintenance of the plant [Sap+18]. Maintaining a reliable operation of the microgrid can be a challenge when there is no available engineer in the community to do the job [BK07; FHC16]. Thus, technical transfer to local people is essential for the sustainability of the power system [Per+12; FHC16]. Issues such as stability of the frequency and voltages of the distribution network can also arise, especially in the islanded mode of operation. This issue is addressed by using a load shedding technique in managing the generation and the demand side [Sap+18]. The work of Schnitzer et al. presented several methods and recommendations based on actual practices employed in off-grid communities for sustaining the operation of off-grid power systems [Sch+14]. According to their study, the sustainability of the off-grid systems implemented in the remote areas can be achieved with good business model and cooperation between the provider, consumer and government that subsidised part of the cost of implementation and operation. This was also highlighted by the work of Frame et al. [Fra+11]. They argue that for the off-grid power system to be sustainable, the community needs to be involved in the design and implementation stages. The community must be educated on how to maintain the power systems on their own. Several case studies were presented to assessed their chosen methods. This comprises their proposed community approach for the sustainability of the power system. Aside from community involvement, another factor that is worthy of attention is the chosen technology to be implemented. Aberilla et al. [Abe+20] emphasised the importance of suitable technology to be installed in the community. Through a series of simulations using Homer, they found out that the house level PV power system combine with wind turbines at the community level with back up batteries is most likely to be a sustainable configuration for off-grid communities. Homer is a software that allows the design of power systems with vast choices of components with real-life characteristics [Abe+20].

The challenges on the implementation of the microgrid systems occur in the off-grid power system in remote areas. According to the study of Schafer et al. [SKN11], these are

- 1. Installed off-grid power system technology is not suitable to the state of the locality and does not satisfy the user's requirements.
- 2. Reliability of the power systems is not assured with issues on the implementation.
- 3. Design and management of the power system are not built with economical financing schemes for the local people.

Other studies such as Ahlborg and Sjostedt [AS15], Miller et al. [Mil+15], and Ulsrud et al. [Uls+15], have mentioned that sociotechnical design aspect of the project must be considered to have successful project implementations. Long-term sustainability can be achieved when this design aspect of the project is examined at the planning stage [IRE18; Uls+15].

2.2 ENERGY MANAGEMENT SYSTEM (EMS)

An energy management system (EMS) is considered to be an essential unit of power systems [ZDM12]. EMS is the control unit of the off-grid systems designed to ensure reliable operations 2.1. EMS ensures the balanced operation between the generation units and the demand side by synchronising the operation schedule of the distributed generation (DG) sources while managing the efficient use of the power loads [ZDM12; Shi+15]. EMS has two distinct approaches in managing and controlling the power system operations 2.2: one is called centralised, and the other is decentralised [Oli+14]. In centralised control and management, the system operates based on the information from a controller that determines the actions taken by all units at a time. This approach entails extensive communications between the controller and the controlled units to ensure a balanced operation [Shi+15]. In a decentralised approach, a controller is used to control each unit. This approach requires a local controller that process information from the local unit only and is isolated from other controllers [Gu+14]. A dynamic consensus algorithm-based distributed hierarchical control method ensures an accurate current sharing and voltage restoration for microgrids with distributed generation sources [Men+15]. With consensus algorithm, the DG units share information and communicate in the network. For off-grid microgrids, the EMS can have either of the two controls depending on the available renewable resources.

In planning an off-grid power system, all significant parameters such as the load capacity, the generation capacity, the expected load, the storage size, the distribution networks, and the EMS, must be taken into consideration to secure successful project implementations [SW12; DM13]. Projection of power load is crucial for a thorough power


Figure 2.1: Energy management system (EMS) [GG15]. Legend: LC - Local controller

system design. The power load must be appropriately established as well as the projected level of daily energy consumption and the factors that influence the variations of this level through time [KK15].

2.3 SIGNIFICANCE OF OPTIMISING THE USAGE OF THE ELECTRIC-ITY IN OFF-GRID COMMUNITIES

In optimising the distributed generation system, the general objective is to find the timetable and the commitment level for each generator and load to minimise the aggregated operation costs and maximise the power usage [ZDM12]. Aggregated operation cost includes the local generation resources and the cost of energy obtained from the power market for grid-tied microgrids. Several studies, such as Meng et al., [Men15b], Zelazo et al. [ZDM12], Shi et al. [Shi+15], and Meng and Zhao et al. [Men15a] conducts experiments and simulation on how to optimise the operation of the EMS.

Other problems for optimisation of EMS is the scheduling of the local sources to generate energy. To optimise the generation cost, Zakariazadeh et al. [ZJS14] adopts demand response programs that enable them to use the energy resources using the stochastic method efficiently. Various types of demand response were considered and participated in the program that represents consumers from residential, commercial and industrial zones. Meanwhile, the work of Rigo-Mariani et al. [RM+14] proves that having optimal scheduling for the next day minimises the generation cost. The confirmation comes from the results of their investigation for different procedures for the optimal power dispatch of a grid-tied microgrid.

On the other hand, the study of Shi et al. [Shi+15], the design of a distributed EMS for optimum operation of microgrids where emphasised addressing the issues on the distribution network and the





B.)

Figure 2.2: Architecture of EMS control A.) centralised and B.) decentralised illustrated by Espín-Sarzosa et al. [EPBN20] Legend: LC - Local controller

associated constraints. As pointed out in the study of Meng et al. [Men+16], and Nasirian et al. [Nas+14], the problem with most of the existing approaches with regards to the EMS of the microgrids (grid-connected or off-grid), the demand-supply matching algorithm considers the demand to be always equal to the generated supply. All generation sets and loads are assumed to be connected to one single bus distribution line and ignore the power distribution network, power flow, and system operational constraints. When those algorithms are applied, the system may not work correctly [Shi+15]. Other energy optimisation approaches include incentive-based and price-based demand response [Ima+18; Zha+18; PNS15].

For off-grid communities, the generation capacity of the off-grid power systems is often limited [IRE17]. Most often, the generated electricity is distributed according to their basic needs for lightings. The available energy is allocated to each household equally to maintain a reliable power system operation [Shi+15; Gup17]. However, some households need more electricity for other purposes such as entertainment (e.g., television and radio), and comfort (e.g., electric fans). This situation leads to a power shortage for some households. Therefore, to maximise the available generated electricity, optimisation of the electricity utilisation by providing sufficient electricity to households is necessary. That is to ensure that the basic electricity needs of the households are provided. This can be done by predicting load and allocating energy of the households based on their previous electricity usage. With accurate forecasting, wastage in terms of unused energy allocation can be reduced if not eliminated. The remaining energy can now be allocated to the households that need more energy, and this ensures that the generated electricity is used efficiently.

2.3.1 Generation-constrained off-grid microgrids

Off-grid microgrids operate off-line and not tied to the traditional grid. The available energy is limited with the installed generation capacity of the power system may it be from solar, hydro, wind or biomass source [Xu+16]. For this research, the off-grid microgrid of interest is the solar power system. For this system, the generation capacity is dictated by the solar panels total capacity of converting the solar source to electricity. The capacity of the off-grid microgrid is not always sufficient to supply the electricity needs of the village; hence the different energy management schemes such as load shedding, load shifting, time-based operation and energy consumption quota are important. The next section presents the details of these approaches.

2.4 ENERGY MANAGEMENT SCHEMES

There are three common energy allocation schemes employed in a microgrid. These are

- load shedding and load shifting,
- time-based operation or scheduling
- energy consumption quota

When the power system is not able to supply the electricity demand of its customers such as households, commercial buildings, hospitals, etc., the available energy should be managed in a way that the basic electricity needs of all customers is served [Kha+18]. Load shedding is used when a threshold of usage of the available energy is breached [Gu+14; D'A+17]. Identifying the critical load is the key to make this approach successful [AAM20]. Load shifting is used when there is a need to defer in time the load operation [EAR20]. Time-based operation is used when the microgrid is only allowed to operate in a specific period of the day due to technical or management issues [Maz+14]. Energy consumption quota is used when the available energy is not sufficient to meet the electricity demands of customers and that they are encouraged to take charge of their energy usage in a day [ZXT12].

2.4.1 Load shedding and load shifting

Load shedding is one of the techniques used by the energy providers to address issues of stability of frequency and voltages in the distribution network. The stabilisation is done by shedding some load to ensure that there is a right balance between the generation and consumer end [Sap+18; Bak+17]. Furthermore, the technique is used in a microgrid with limited generation capacity to cater to the demand of the consumer by shedding loads identified as less prioritised [Gu+14]. The process of load shedding involves disconnecting the load from the power system when the energy demands are higher than the available energy.

In load shedding, energy management control is actively monitoring the conditions and status of both the generation and demand side. When a crisis occurs, load shedding commences ensuring the balance of the operation of the power system. Maintaining the stability of the system is one of the crucial processes in operating the microgrid. Multiple crises can occur depending on the current conditions of the microgrid that are caused by either energy sources, distribution network, and control instrumentations.

Traditional load shedding approaches, under frequency load shedding (UFLS) and voltage load shedding (UVLS), are usually designed separately to protect the system independently from frequency and voltage instability, respectively [SSP11]. However, they are not equipped to handle mixed fluctuations from frequency and voltage. This instability, if not addressed properly, would result in power system breakdown [Tan+13]. The inflexibility of the traditional approaches fails to consider the other sources of the instability of the system as they are designed independently and separately implemented, which made strategies unfavourable [SSP11].

To improve the traditional load shedding approaches, the adaptive approach was introduced as presented in the work of Mohamad et al. [Moh+18], Terzija et al. [Tero6], Chin et al. [Chi+08], Hooshmand et al. [HM12], and Santos et al. [San+19]. These research studies employ either machine learning algorithms such as artificial neural networks and a centralised, hierarchical multi-agent scheme that coordinates different stages of the monitoring and decision-making process or a combination of the approaches. However, all these studies have only considered frequency information and not the voltage information in dealing with instability issues in power systems. Kanimozhi et al. [KSB14], and Tamilselvan and Jayabarathi [T]16] proposed new schemes of improving the stability of the power systems using voltage information combined with genetic algorithm (GA) and artificial neural network (ANN). As the techniques are working separately in dealing with the voltage and frequency instabilities, this motivates other researchers to address the issue by proposing techniques that deal with both the frequency and voltage information as presented in the works of Tang et al. [Tan+13], Hsu et al. [HCC11], and Giroletti et al. [GFS12].

Adopting load shedding help maintains the balance between the generation and demand side of the power systems but this lead to power blackout for some consumers that are off-loaded from the power systems [Xu+16]. Load shedding is good for the generation side of the power system but will have a negative impact on the off-loaded consumers. Load shedding as presented by Xu et al. [Xu+16] was proposed to address the issue with microgrid with limited generation capacity. In order to continue supply power to the critical load of the power system, a dynamic load shedding was proposed. They formulated the load shedding as a stochastic optimisation problem where uncertainties caused by the intermittent power resources and the loads are considered as constraints. They aim to maximise the

economic performance of the microgrid considering the limitations of the generation resources. Then they develop a model using a Markov decision process (MDP) to determine the optimal load shedding process. With their approach, they were able to maximise the economic factor of the power system by providing an optimal load shedding strategy for a microgrid with limited generation capacity.

2.4.2 Time-based operation

The EMS schedules the availability of the power to the consumers. The operation can be scheduled hourly or daily depending on the availability of the generated energy. This is done to maintain a well-balanced operation of the microgrid. For a traditional grid, scheduling is done to have an optimal system operation. This approach has been addressed extensively in the recent research work in the field as the most efficient means to optimally coordinate the controllable and uncontrollable resources [Ima+18; CMP14; Zha+16; MMS16b; Jia+15; GZ16].

Carpinelli et al. [Car+17] proposed a way of minimising imbalances in low-voltage microgrids by the scheduling of distributed resources. Their work is based on a multi-objective approach that considers the structure of the low-voltage microgrid systems which have inherent imbalances of lines, loads, and generation systems. They formulated their model to be a multi-objective optimisation problem. The objective functions are defined as dependent to power, line current and the positive and negative component of voltages in minimising the imbalance factor, cost of energy, peak shaving, losses, security margin and voltage deviation. The constraints relating to the technical limitations and operation requirements are all considered from the buses (such as load bus, distributed bus, and electric vehicle bus), and the microgrid constraints such as different phase-voltage magnitude, and the line phase-currents. Their experimental results show that using a multi-objective approach for the scheduling of unbalanced microgrids provide advantages in terms of savings and efficiency in system operations.

In most cases, scheduling the operation of microgrid has been a challenge. The work of Mazidi et al. [Maz+14] presents an efficient way of solving the operational scheduling problem of microgrids with different generation sources. The problem was formulated considering the various constraints related to microgrid operations such as allocating the reserve capacity, battery scheduling, and the uncertainty innate to the renewable resources like the wind and solar power generations. A Latin hypercube sampling method was employed to combine and generate the different scenarios relating to the two sources, which in this case were wind and solar power generations. Their results showed that a lower reserve requirement is achievable with the stochastic method they used compared with a conventional deterministic approach.

Mazidi et al. [MMS16a] present an effective and efficient day-ahead scheduling of smart distribution networks by incorporating priceresponsive customers in their method. They formulated an optimisation problem to maximise the profit while maintaining the customers' satisfaction. The optimisation considers the hourly sale prices offered to customers, purchase or sale transactions made to the market, a commitment of the distributed generation units, dispatch of battery energy storage systems and the planning of interruptible loads. The models were formulated as a Mixed Integer Linear Programming (MILP) optimisation problem. The model formulation covers both the demand and consumer side that benefits both sides. Mathematical models were formulated for both distribution network and demand response (DR) including price-based and incentive-based DR models. These formulation leads to the objective function to maximise the difference between the profit and cost of operations. Constrains relating to power balance, distribution network, and distributed generation units were considered. The optimisation problem was then solved using Karush-Kuhn-Tucker (KKT) conditions. According to the results of their implementations and case studies, higher profit is achievable while keeping their customers satisfied. Their method even results in the customer's motivation to use energy efficiently and have a reduced electricity bill.

Most of these approaches are focused on the generation side and not on the customers' side in scheduling the operations of the microgrid. Most of these approaches are made for microgrids that are gridconnected and not in stand-alone. There are very few published works addressing the time-based scheduling operation of the off-grid power systems [Mar+13; Mar+16]. Hence, this necessitates us to look into the possible method that can work in any stand-alone microgrid systems that allocate daily energy to each customer to maintain a stable operation.

2.4.3 Energy consumption quota

Another popular technique in dealing limited generation capacity of the microgrid is allocating energy consumption quota (ECQ) to each household daily [Sun+16]. This is to maintain a stable operation of the power system. In this setup, all households connected to the power system is given an equal energy allocation daily for a fixed amount of monthly fees. Some use energy quota to encourage energy savings in commercial buildings [ZXT12]. Another application of energy quota is influencing the energy consumption of commercial buildings. By giving them energy quota, commercial buildings will be encouraged to limit their energy consumption within the given quota [Xin+12].

For example, the study of Xin et al. [Xin+12] presented a method on how to calculate the energy consumption quota of different hotels in China as a way of promoting the development of building efficiency. The method covers from data collection to analysis, and calculation of energy consumption data using statistical analysis methods such as the mean index of total energy consumption (MITEC), mean of EUIs, quadratic average method, median, percentile method and mode. For data collection, they used a questionnaire to be field up by the building owners. The data gathered involved the basic building information such as building area, age, number of guestrooms and occupancy rates as well as the energy consumption of the hotels in recent years. The basic building information was considered as factors affecting energy consumption of hotels. Basically, they gathered the information through a questionnaire answered by building owners, and then the energy providers verified the energy consumption. The approach is susceptible to biases as answers, and the verification process is based on the memory of the respondents (owners and energy providers). The study does not mention that the information was verified from the archives of the energy providers. Hence, the validity and reliability of the data may be compromised. However, their study underpins the importance of actual historical energy data for the development of energy consumption quota of the hotels. A correlation was performed to determine the factors affecting building energy consumption (BEC). These factors refer to the basic building information they have gathered. For the analysis of the energy data, they used the statistical analysis, and this approach serves its purpose right for calculating the energy consumption quota of each hotel. This is because there are no constraints involve except for promoting the building efficiency in China. However, for off-grid microgrids with limited generation capacity, this approach may not be adequate to address the constraints.

All three energy management schemes have their advantages and disadvantages, as presented in this section. Regardless of the type of renewable energy resources, these approaches are applicable in ensuring balanced operation of off-grid microgrids. For most cases in off-grid communities, energy daily quota is the most common technique used for its simplicity in implementation [IRE18]. However, this approach can be improved by combining machine learning algorithm

such as a neural network to improve the energy management system of the microgrid [ZXT12].

Daily energy allocation is quite popular in off-grid power systems implemented in remote areas [Int14; IRE17; IRE18]. This approach can be improved by incorporating the load forecasting technique that would predict day-ahead daily consumption. By doing this, the energy allocation can be done optimally, and better energy utilisation of the generated energy by the off-grid microgrid with limited generation capacity can be achieved.

2.5 LOAD FORECASTING TECHNIQUES

Forecasting is considered an essential component of any power providers [Gup17]. Forecasting the load helps operation planning which can lead to reducing unnecessary power production. Currently, power providers gather the power usage of their customer to help them predict the next day average power usage [Chi+15]. The accuracy of predicting the power usage of any building or industry has improved a lot since the application of neural network for forecasting as presented in the study of Dudek et al. [Dud16]. With accurate load forecasting, efficient generation scheduling can be achieved [Gup17]. The main objectives of accurate load forecasting can be categorised as

- Scheduling the operation of the generation of the power system
- Securing the reliable operation of the power system

Moreover, the allocation of generation resources, the operational limitations, environmental and equipment constraints can be determined and properly addressed [ESMK11; DM13]. For example, in the case of hydropower generation units, the optimal release of water from the reservoir and the generation scheduling of power systems can be carried out based on the short-term load forecasting (STLF) [RK15]. Load forecasting is also used to ensure power systems operations regardless of the distributed generation units used to generate the power [Maz+14; Dud16; RK15]. Accurate load forecasting is a crucial tool to determine the optimal operational state of the power system [ESMK11; RK15]. Additionally, forecast data can be used to prepare the power system per future load state and corrective actions. All these objectives lead to saving the cost of operation of the power system [FA14; RK15].

Load forecasting is used by the energy provider to ensure stable operations of power systems and to predict the necessary power to be generated to meet the demand [Gup17]. Unnecessary expenditures in generating excessive power can be avoided when an accurate prediction of the load is performed [FA14]. Moreover, accurate load forecasting is a necessary tool to determine the optimal operation of power systems [Sol+15].

Predictors such as historical load data, time information (i.e. year, month, day of the month, and day of the week), and minute-dependent context features (e.g., temperature, humidity, wind speed, UV index, and time index) were identified as factors affecting energy consumption [Hsi15]. These factors are used as the inputs for the backpropagation neural network to predict the household's electricity consumption volume. A strong correlation between weather information and load demand is evident as presented in the paper of Hernandez et al., [Her+12] and Hsiao et al., [Hsi15]. A study claims that incorporating appliance usage patterns for non-intrusive load monitoring improves the performance of load identification and forecasting using non-intrusive load monitoring approach [Wel+17]. A long short-term memory (LSTM) recurrent neural network (RNN) based framework was used to address the issues such as uncertainty and high volatility in load forecasting for an individual household [Kon+17].

Depending on the available data, prediction can be made a day ahead, a week, a month, or even a year [Moo+19]. The prediction can be categorised as follows:

Short-Term Load Forecasting (STLF)

SLTF refers to forecasting the load in the range of minutes, hours, days or a week. A short period where the load is specifically determined to provide better accuracy in forecasting [TS16]. Short-term load forecasting is used for reliable and efficient energy management and is influenced by many factors, such as weather, house occupancy and household income [KRF13].

Medium-Term Load Forecasting (MTLF)

MTLF refers to forecasting the load in the range of several weeks, months and a year [Gup17]. This is done when the industry wants to know the next week or next month average power usage. The result may not be as accurate as it can be like the STLF, but the results give an idea of how much energy is needed to be generated for the next week/month. Medium-term load forecasting is used for efficient operation and maintenance of the power system [Bor+17].

Long-Term Load Forecasting (LTLF)

LTFL, as the name implies, is in the range of years. This can be done when there are lots of data for years (10 to 15 years worth of data) [TS16]. So the prediction is made for the entire year of average power usage. Long-term load forecasting is used for long-term power system planning according to the future energy demand and energy policy of the state [Chaoo].

2.5.1 *Time series forecasting*

Analysis of time series can be divided into two essential parts. The first part focuses on the derivation of the structure and the underlying trend of the observed data. The second part emphasises on the fitting of the model to perform future predictions [LLCo9].

Time series analysis is used in many applications, such as economic forecasting, yield projections, process and quality control, workload projections, and any other predictions that use history data with inherent correlations, trends, and complexity [PB17; Deb+17].

The general approach in analysing the time series is to divide the time-series into three elements [Dud16; Tay10; Bor+17]:

- Trend The general characteristics that the variable exhibits within the considered period without taking into consideration the seasonality and irregularities of the data [Dud16].
- Seasonality This refers to the cyclic variations of the variable concerning weather information. The seasonality consists of the effects that are steady and unwavering through time [Tay10].
- Residual This is the remaining part of the time series after the above two features are considered. Most of the time, this part of the time series is mostly unexplained. Sometimes, residuals can be large enough to conceal the trends and seasonality of the variable [Bor+17].

Model fitting of the time series is a complex and challenging process.Time series forecasting can be grouped into univariate and multivariate analysis [TMMo6]. Univariate time series analysis is a time series forecasting that has a single observation recorded sequentially over time while multivariate time series analysis deals with a group of time series variables that interacts with each other [TMMo6].

2.6 FUNDAMENTALS OF ARTIFICIAL NEURAL NETWORKS (ANNS)

ANN-based load forecasting models are used by many energy providers for decades now since the earliest published paper on electric load forecasting using an ANN [Par+91; MS97]. Energy is then generated according to the forecasted energy consumption using the ANN model. Many research works are published to show how ANN-based load forecasting models used to resolve issues in the energy management system, particularly the generation side []. However, there are few published research works for forecasting energy consumption for the benefit of consumers (demand side). Since the goal of this research is to provide sufficient energy to the households from a source with limited generation capacity, forecasting the energy consumption of the households is important.

Artificial neural networks (ANN) are modelled after the cognitive learning process and the neurological functions of the human brain. Neural networks are capable of predicting new observations from the existing observations after learning from data. A specific neural network architecture must be designed and built. This includes choosing a specific number of hidden layers, each consisting of a certain number of neurons. The training process commences once the number of hidden layers and neurons are established. An iterative process is applied by the neurons to the number of inputs to adjust the weights of the network to optimally predict the sample data on which the training is performed. After learning from the given existing data set, the neural network is now ready to forecast. In forecasting, the output of the network represents the pattern it has detected from the training data set [LBH15].

Neural networks are considered to be one of the best techniques in load forecasting. One of the desirable features of the neural network is its robustness and flexibility in handling real-world data that have both linear and nonlinear components. It can handle well both the linear and nonlinear aspects of the data in generalising when used as approximators [HSW89]. Good modelled neural network-based forecast model can make a precise prediction of new observation after learning from the existing data. A specific network architecture must be designed specifying the network's hidden layers and neurons. In the training process, the network is learning specific features of the data. The network developed and trained with the existing data can be used to predict new values from the pattern it has detected and learned from the training.

Figure 2.3 shows the basic architecture of neural network. Traditionally, the neural network is fed with an input vector (or matrix depending on the architecture of neural network) V_x and generates the output vector V_y in which the network's topology defines the relationship between them. For multilayer perceptron (MLP)-based model, 3-layer back propagation neural network is widely accepted to generate an estimation of any continuous function with sufficient mid-layer units.



Figure 2.3: The simplest architecture of neural network has an input layer, a hidden layer, and an output layer.

2.6.1 Back-Propagation Neural Network

Backpropagation (BP) learning algorithm is used to train the ANN. The BPN learning algorithm uses the gradient descent method to update the weights of the network [LBH15]. The BPN network consists of an input layer, a hidden layer, and an output layer that are connected by synaptic weights.

There are several issues with BPN algorithm that affects the performance of the network while training. These are as follows:

- Local minima The training algorithm aims to reduce the error to the global minimum value to achieve higher training performance of the network. However, sometimes the network can get stuck in the local minimum even though the weight values keep updating and adjusting. This results in poor performance in training the model. This problem can be avoided by applying a modified error function that harmonises the updates of weights between the hidden layer and the output layer [Bi+o5].
- Network paralysis The weights of the network can be varied and adjusted for large values of output where the derivative of the triggering function is small during the training process. This is to send back the error of the network according to its derivative during the training process. Network paralysis happens when the network comes to a standstill during the training process that may lead to erroneous network output due to poor learning. By using the rectified backpropagation neural network called Morbidity neuron Rectified Backprogation network (MRBP)[JS05].
 MRBP identifies and corrects the defective neurons making the network more adaptive and robust that can easily escape from a local minimum.

- Temporal instability For the training algorithm to learn, the neural network needs to imitate the whole training set without disruptions according to what it has already learned. When a backpropagation training algorithm fails to learn from complex systems, network learning is suffering from temporal instability. This happens when the network fails to remember what it has already learned while trying to learn something new during the training process [RK15]. This issue can be minimised using Self-Partitioning Neural Network (SPNN). SPNN measures the conflict amongst the datasets during the training process and split into smaller groups by partitioning. The partitioning is done in such a way that the conflict between groups is significantly reduced. Reducing the conflict between groups speeds up the training process of the network and decreases the network paralysis and temporal instability [RKS95].
- Generalisation and the over-fitting problem of the neural network
 In modelling, the ultimate goal of the training is to minimise the
 Mean Squared Error (MSE) of the network with its training set.
 When the output of the network is accurate or close to the target
 values of the data that are not included during the training, the
 network is considered to be well generalised. Several factors
 affect the generalisation of the network. The factors include the
 quality and size of the data and the architecture and complexity
 of the neural network [Baso7]. With more complicated models,
 the possibilities of over-fitting are high. An over-fitted model can
 describe the training data well but poorly predict when given
 with new dataset [Cha+16a].

2.6.2 Issues in designing neural network-based forecasting models

Unlike other forecasting models such as Auto-Regressive Integrated Moving Average (ARIMA), the selection of parameters, such as the Akaike Information Criterion (AIC) or the Bayesian Information Criterion (BIC), for a neural network is not straightforward. The approach usually starts with the simplest architecture and gradually explores more complex structures. Kaastraa and Boyd [KB96] presented a design process for ANN-based forecasting. They aim to provide an introductory guide for designing an artificial neural network for forecasting. They presented the 8-steps design procedure of the neural network for forecasting. The steps start from input variables selection until the implementation of the model. The procedure was implemented in forecasting financial and economic series. Raza and Khosravi [RK15] conducted a systematic literature review of Artificial Intelligence (AI) based on short-term load forecasting techniques to identify, evaluate and analyse the performance of AIbased load forecasting model. They have concluded that the accuracy of ANN-based forecast model is dependent on several parameters such as model architecture, input combination, activation functions and training algorithm of the network and other variables that affect the input of the forecasting model. A technique that can generate a diverse and effective neural network, a group-based chaos genetic algorithm was developed by Chen et al. [Chi+16] to improve the performance of STLF. They also developed an ANN-based nonlinear ensemble of partially connected NN predictors to enhance the accuracy of the STLF further.

In determining the design parameters of the artificial neural network, the procedure become complex for finding the right combinations of the network's hidden layers HL and neurons N, in which a trial and error approach is usually employed.

Palmer et al. [PJSo6] also presents a step-by-step methodology in designing ANN-based forecasting for tourism. They point out that there is no definite rule in choosing the number of hidden layers HL and neurons N to build a neural network that generates accurate prediction. They provided a trial and error approach for a specific range of hidden layers HL and observed the effects on the performance of the network. The models were assessed in terms of Root-Mean-Squared Error (RMSE) and Mean Absolute Percentage Error (MAPE) for performance accuracy using validation data. According to their experimental results, MLP architecture with eight inputs, one hidden and one output neurons performs best with the most accurate prediction with the validation data. For this research, both RMSE and MAPE is used to evaluate the performance of the proposed MLP-based forecast model.

Hunter el al. [Hun+12] also pointed out the difficulty of the researchers in deciding the proper sizes and architectures when using neural networks. Their work presented a comparative analysis of the performance of 3 neural networks, namely Multilayer Perceptron (MLP), Bridged Multilayer Perceptron (BMLP), and Fully Connected Cascade (FCC). They also presented a detailed comparison of learning algorithms used for training the neural networks. Amongst the learning algorithms they have used, they pointed out that the Levenberg Marquardt (LM) algorithm is the most efficient from their theoretical analysis and experimental results. LM algorithm is used in MatLab neural network toolbox. This research used MatLab neural network toolbox in building the load forecast model that utilises household demographic profile as inputs together with the historical energy data to forecast the next-day energy consumption of the households.

Given the advanced configurations of neural network for load forecasting, MLP, as one of the basic architecture of ANN, is commonly used. This is because MLP allows simple implementation of the network, and it can handle both linear and nonlinear features of realworld data. For this reason, MLP-based forecast model is preferred in forecasting the next-day energy consumption of the households in this research. The details are presented in Chapter 5.

2.7 FACTORS INFLUENCING LOAD FORECAST MODEL PERFOR-MANCE

In load forecasting, selecting the right inputs is one of the crucial processes. This process is conducted to determine what are the key factors affecting the energy consumption of households, buildings, or any other dwellings[Che17]. In determining factors affecting load forecasting, there are several parameters to be considered. According to the study of Hsiao [Hsi15], behavioural patterns of similarity and difference in electricity consumption are caused by external factors called context factors. These factors are time, weather, calendar days and economic indicator. Aside from historical data, these factors significantly affect load forecasting. The work of Bedir and Kara [BK17] used factor analysis to determine the factors influencing the behaviour of households towards electricity consumption. Then they used these factors in correlation analysis to determine the relationship between the behavioural factors and the household characteristics. These approaches made them identify what the drivers and behavioural profiles that significantly affects the way dutch dwellings consumes electricity are. The factors they identified are household appliances, presence (in terms of frequency of lighting usage in different rooms), household characteristics (this includes the household size, occupants age, monthly household income, education, working schedule), and type of dwellings. According to their results, there is a significant correlation between behavioural factors and electricity consumption. Also, recall from Section 2.4.3 that these factors are similar to the factors considered by the work of Xin et al. [Xin+12] as they determined the factors affecting BEC using the same method, the correlation analysis. Thus, these factors were considered important demographic information in the proposed load forecasting model used in this thesis.

Chae et al. [Cha+16b] use variable important analysis to select key features that affect electricity consumption of commercial buildings. The features include the day type indicator, time-of the day, HVAC set

temperature schedules, outdoor air dry-bulb temperature and outdoor humidity as the most critical factors. These factors can be categorised into subgroups such as historical data, economic factor and weather information [Che17].

In the study of Mandal et al. [Man+o6], it was pointed out that the performance of the forecast model is improved as the temperature is added as one of the input. Moon et al. [Moo+19] present an ANN-based model that considers factors affecting energy consumption such as calendar data, weather information and historical data.

For every subgroup of the factors affecting load forecasting, each needs corresponding data. Using actual data can achieve more realistic results in load forecasting as opposed to computer-generated data from a formulated function. The work of Hsiao [Hsi15] presents how the data being used for pattern analysis to determine the demand of the household for each trend and daily behaviour patterns of electricity usage from the actual data.

2.8 USER PREFERENCES

Another parameter that needs to be considered in forecasting the household's load in off-grid communities is user preferences. User preferences refer to the household's behaviour related activities that use electricity. Accuracy in load forecasting can be improved when the behavioural-related parameters such as daily frequency of usage of deferrable loads, and non-deferrable are being considered in the modelling and prediction. The studies of Hsiao et al. [Hsi15] and Kong et al. [Kon+17] reported that user behaviour towards energy usage affects the aggregated energy consumption of the consumers. Electricity usage can be determined by studying the behavioural electricity usage patterns of individual households. Behavioural patterns can be determined based on the actual occupant behaviour towards lighting and appliance use [BK17]. Mixed methods approach was used to determine the importance of occupant's behaviour towards electricity usage. Data were collected and analysed to determine the effect of individual comfort and household attributes on occupant's behaviour towards energy [GS+16]. Occupant's behaviour is one of the factors influencing energy usage. A study presented an approach to estimate potential energy savings using behavioural measures. The approach includes the profiling of the occupant's behaviour, then simulate and analyse the individual and integrated impacts on energy usage. The study concluded based on the simulation results that the behaviour measures can save energy significantly, and the main energy savings came from the energy savings on unoccupied rooms [SH17]. Individual residential loads are usually volatile to forecast because of the resident's various activities. An extended short-term memory-based deep learning forecasting framework with appliance consumption sequences has been proposed by Kong et al. [Kon+17] to address the issue.

2.9 RESEARCH DATA

Most of the data used in research can be taken from any repository that is available for the public. For microgrids, there are several data available for public consumption and research purposes, and the most popular ones are from GitHub [Tia18; Bye19], government domain (Department of Energy) and electric company that are available upon request. However, for energy data of households connected to generation-constrained microgrids, there are very few to none data available for public or research use. Also, as pointed out in section 2.4.3, historical data is vital in determining the ECQ of the buildings. Thus, this research developed and deployed an energy usage of the households. The energy monitoring system is designed to gather specific information from the households about their energy usage. Chapter 4 presents the development and deployment of the system.

Factors affecting energy consumption are also collected. As discussed in section 2.7, demographics of the households is gathered through face-to-face surveys. The survey is used as this is the simplest way of gathering information. Chapter 4 presents the development of the survey questionnaire and the deployment of the survey to the selected villages in Cebu, Philippines.

Using these collected data in the development of the load forecast model (detailed in Chapter 5) will give real-world results that are more reflective of the actual usage of the households. For this research, the data gathered is used for the development of the load forecast model and the proposed optimal energy allocation scheme. The process of gathering the data is presented in Chapter 4.

Chapter 4 presents the gathering and preprocessing of the data from the deployed PMOG systems in the households at the selected remote villages in Cebu, Philippines. The real-world data serves as the baseline data for the development of the MLP-based load forecast model, as discussed in Chapter 5.

2.10 CHAPTER SUMMARY

In this chapter, the importance of energy consumption quota in the context of an off-grid microgrid is presented. Energy consumption quota (ECQ) promotes building efficiency. For households in off-grid communities, it ensures the basic electricity needs of the household while using the available energy efficiently from a generation-constrained microgrid. The actual data from the households or buildings is the key to calculate the ECQ. The method used by Xin et al. [Xin+12] in data collection can be improved by employing actual energy monitoring systems. The approach of calculating the ECQ can be improved by using machine learning and optimisation techniques that can handle defined constraints.

The need for optimising the energy management system of a power system is discussed. Different schemes of the energy management system (EMS), such as load shedding, time-based operation, and energy allocation are presented and how they are being employed to maintain the balanced operation of the power systems (Section 2.4). For off-grid microgrids, energy allocation is commonly used to cater and provide the basic electricity needs of the connected households.

In optimising the operation of the power system, load forecasting is incorporated into the energy management control system (EMCS). MLP-based load forecasting is one of the popular techniques along with ARIMA, GPR and other forecasting techniques such as RBFN. The fundamental architecture of the neural network is presented as well as its robustness in dealing both linear and nonlinear features of data which makes the neural network as a good approximator (Section 2.6). Furthermore, applications of neural networks on load forecasting are discussed.

Finally, factors affecting load forecasting are also presented and discussed (Section 2.7) as well as the importance of gathering an actual energy usage (Section 2.9) for the development of MLP-based forecast model in Chapter (5) and the proposed optimal energy allocation in Chapter (6) for households connected in off-grid microgrids with limited generation capacity.

Within the presented literature, studies about optimal energy allocation in the context of off-grid microgrids are limited. Although there are several studies (Section 2.4) in optimising the energy management system, these approaches are not implemented in an off-grid microgrid with limited generation capacity. In providing an optimal solution for an optimisation problem, real-world research data is crucial. Thus, this thesis focuses on the following areas:

- In addressing the importance of historical energy data in determining the ECQ, an energy monitoring system (EmS) is developed. PMOG system is built using readily available components and sensors to gather energy data. For this research, gathering real-world data of load consumption from the selected villages that will serve as the baseline data for the load forecasting model. The deployment of the PMOG system to gather historical energy data is discussed in Chapter 4.
- 2. Development of MLP-based load forecasting model that includes household demographic profile as inputs to achieve better prediction performance of the network. The need for forecasting the next-day energy consumption is crucial in calculating the optimal energy allocation, hence MLP-based load forecast model is proposed. The process of finding the best combination of parameters of the neural network is presented in Chapter 5.
- 3. An optimal energy allocation based on forecasted load consumption considering the limited generation capacity of the microgrid and maintaining the basic electricity needs of the households. In the context of off-grid power systems, ECQ ensures sufficient energy for the households while maintaining a balanced operation of the power systems and efficiently use the available energy. The analytical approach of allocating daily energy allowance optimally to each household is presented in Chapter 6.

METHODS

This chapter describes the methods used in this research for the data gathering as discussed in Chapter 4, developing the forecast model presented in Chapter 5, and optimising the daily energy allocation of the households as shown in Chapter 6. Figure 3.1 shows the flow of work of this research from data gathering to allocating of daily energy quota using the proposed scheme.

3.1 DATA GATHERING

Data gathering is divided into two parts. First is the collection of demographic profile through field survey, and second is the collection of the energy usage using the PMOG system.

As discussed in Section 2.7 and 2.9, real-world data is essential in developing models that would describe the characteristics of the household's energy consumption [Che17]. These data include the factors affecting their energy consumption and drivers of using energy [BK17]. Since data are important in developing the load forecast model, the factors influencing the behaviour of the households towards energy, as identified by surveying the existing literature, are classified as demographic information. A survey is conducted to collect this data.

To do the first part of the data gathering, we need a survey questionnaire. The survey questionnaire serves as a guide while conducting the survey to the participants. The development of the survey and its deployment is discussed in Chapter 4, Section 4.2 and Section 4.4, respectively.

A survey was done with the developed survey questionnaire with the aim of gathering the demographic information of the households. A face-to-face survey was employed as this approach is the most suitable for the selected villages that are located in remote areas. Another approach, such as online survey is not doable as the people in the villages have no access to the Internet.

For the second part of the data gathering, a PMOG system was developed and installed to the selected households of the two villages. PMOG system is an energy monitoring system (EmS) that is designed to gather the energy consumption of the households in two levels – household level and appliance level, using wireless sensors. The



Figure 3.1: Flow of methods used in this research.

development and the deployment of the system are presented in Chapter 4, Section 4.3 and Section 4.4.

For the historical data, energy monitoring systems called Philippines-Micro-Off-Grid (PMOG) systems were installed in the selected households to monitor and gather their electricity usage. PMOG systems were developed and tested in the laboratory before the deployment.

3.1.1 Evaluation of the survey questionnaire and testing of PMOG system before deployment

The survey questionnaire was tested by conducting a pre-survey in the laboratory with colleagues act as participants. This is done to ensure that the questionnaire is readable, understandable and each question written serves with a purpose for the research as a whole.

For the PMOG system, the testing was done by allowing the system to run for two months prior to the deployment to test its reliability. The output is being recorded and monitored in the remote server. When data gathering is interrupted, the systems were checked for errors. If code bugs caused the errors, the system code is adjusted to make sure that the error will not happen again if external factors like power blackout cause the interruption, the system will just stop recording. This will appear as blank or N/As in the data for the duration of the blackout. No data or N/As will appear in the records whenever the system is experiencing a blackout or power interruption. Chapter 4, Section 4.3.1 presents the details of the PMOG system flow and operations.

3.2 LOAD FORECAST MODELLING

A Multi-layer Perceptron (MLP) based model is employed for load forecasting in this research. MLP network is used because of its simplicity and straightforward approach in forecasting and can perform best while dealing with the nonlinearities of the data. Four forecasting techniques were considered in the development stage of the forecast model for this research. These four techniques are MLP, ARIMA, RBFN and GPR. Both MLP and RBFN can handle well with nonlinear features of the data, and both ARIMA and GPR are best suitable in handling linearities.

The MLP-based forecast model is developed by using the Matlab tool. Functions are available to use with an option of designing the network by determining the best combinations of the network's layer and neurons, as well as the number of the historical electricity usage of the households. Different simulation and experiments were conducted to determine the right combinations of the networks' hidden layers HL, neurons N, and delay window D. Simulations were conducted to observe the influence of these three parameters in the performance of the network in forecasting. The model is trained using the energy data from PMOG system and demographic information from the survey data. During the training stage of the model, several case scenarios involving a number of inputs for the network were considered—all for the purpose of finding best performance of the model in forecasting in terms of RMSE. The goal is to find the lowest possible RMSE given the inputs and the chosen architecture of the neural network. The process of determining the forecast model includes the input shaping, as discussed in Section 5.1.2.

In the development of the forecast model, the following inputs were considered:

1.) Historical energy data (PMOG data) such as power consumption of the household

- 2.) Household's temperature (indoor)
- 3.) Calendar days such as weekdays

4.) Household profiles from the survey data (demographics, and monthly energy tariff)

The steps on how to design a neural network for load forecasting is presented in Chapter 5 as well as the details of the simulation and its results.

The final model with the lowest RMSE is chosen through a grid search where all possible combinations of inputs and neural architectures are considered.

3.2.1 Evaluation of the loaf forecast model

The forecast model was validated using k-fold validation with k = 10 in all input combinations used to determine the best combinations of the three networks parameters. This is presented in Section 5.1.1 The MLP-based forecast model is evaluated by comparing its performance to three other techniques such as Gaussian Process Regression (GPR), Autoregressive Integrated Moving Average (ARIMA), and Radial Basis Function Network (RBFN) (see Section 5.4). Furthermore, the performance of the proposed load forecast model used in this thesis is compared with two other works that also used neural network-based models in load forecasting as detailed in Chapter 5 Section 5.15.

3.3 OPTIMISING ENERGY ALLOCATION OF THE HOUSEHOLDS

The optimisation of the energy allocation of the households consider in this research takes two-steps process. The first step is the forecasting of the next-day energy consumption based on their historical usage and households profile. The second step is the calculation of the optimal energy allocation, considering the known constraints.

The first step is done using the proposed forecast model developed using MLP that uses the historical energy data and the household profiles. The second step is done by solving the optimisation problem using the Karush-Kuhn-Tucker (KKT) approach, given the constraints of the microgrid generation capacity and the minimum energy level for each household. The optimisation process aims to minimise the difference between the forecasted values and the calculated optimal energy allocation of each household. At the same time, it is ensuring that the total optimal energy allocation does not exceed to the generated energy by the microgrid and that the optimal energy allocation of each household should not be lesser than the minimum energy required by the households. This is to ensure maximum utilisation of the available energy while ensuring that households that need more energy than what is allocated will get sufficient energy. There will be no wasted energy allocation in terms of excess of energy allocation or unused energy allocation. The formulation of the optimisation problem is presented in Chapter 6, Section 6.1

KKT approach is one of the optimisation techniques that is used in solving an optimisation problem that subjects to constraints written

in the form of inequalities. The formulated optimisation problem described in Chapter 6 meets the requirements of KKT conditions that involves inequality equation in the optimisation conditions allows us to use the KKT approach. These conditions are the constraints written in the form of equality and inequality. The two constraints meet the requirement of KKT conditions; hence the approach is used to solve the optimisation problem. The details of solving the optimisation problem is presented in Chapter 5, Section 6.1.1.

3.3.1 Evaluation

Using the available data from PMOG system, the approach is evaluated by carrying sample calculations on what would be the optimal energy allocation using the proposed method and compared with the other existing methods such as the equal daily allocation and ratio and proportion approach. Chapter 6, Section 6.1.4

Analysis of Variance (ANOVA) is employed to determine the difference between the average energy allocation of each household. This is to show that the difference between each households energy allocation is statistically different. ANOVA is the most common and easy to implement in comparing the averages of the two or more groups to determine whether the difference between the averages are statistically significant. The null hypothesis is tested by comparing the means of the different groups, that is

$$H_0 = \phi_1 = \phi_2 = \phi_3 = \phi_n \tag{3.1}$$

where ϕ is the means and n is the number of groups. The alternate hypothesis H_A is accepted that the group means are statistically different from each other when the result of ANOVA test returns statistically significant.

The proposed optimal energy allocation is validated using the values forecasted by MLP-based forecast model. It is important that the results of the calculation of the optimal energy allocation should meet the stated constraints of the optimisation problem. The detailed validation process is presented in Chapter 6, Section 6.1.4.

In this research, three methods in daily energy allocation are considered together with the proposed optimal energy allocation that uses load forecasting and KKT conditions to be compared using ANOVA. The detailed discussion is presented in Chapter 6, Section 6.2.

3.4 CHAPTER SUMMARY

This chapter presents the methods used in the proposed optimal energy allocation of households connected in off-grid microgrids with limited generation capacity. The methods cover the data gathering to the development of the forecast model and calculating the optimal energy allocation of each household.

In collecting the necessary data for the development of the forecast model and the solving the optimisation problem, a survey questionnaire and PMOG system were developed and deployed. For forecasting the next-day energy consumption of the household, an MLPbased forecast model is proposed that uses both historical energy data and households profiles. A mathematical approach is proposed in determining the optimal energy allocation using KKT approach incorporated with the forecasted values by the MLP-based forecast model.

In this chapter, a two-step process of finding the optimal energy allocation is proposed using MLP-based forecast model and KKT approach.

4

DATA COLLECTION AND PROCESSING

Given this research project is done with the support of British Council, Philippines in partnership with the Department of Science and Technology-Science Education Institute to positively impact the chosen local communities, this chapter presents the selected villages and the data gathered from these villages. The data collected serves as the baseline of this research in the development of the load forecast model presented in Chapter 5 and the optimal energy allocation shown in Chapter 6. The problem this research stated in Chapter 1 is a realworld problem faced by the people in off-grid villages. For this reason, the profiles of the villages are also presented in this chapter.

This chapter also presents how the data collection process using the energy monitoring system called PMOG system and the survey questionnaire. The demographic information collected includes the factors affecting their energy consumption as discussed in Section 2.7 that falls to the three subgroups, namely historical data, economic factor, and weather information. Furthermore, this chapter presents the foundation of the proposed adaptive energy allocation, given the collected data from the selected households.

To perform load forecasting and calculate the optimal daily energy allocation for households connected to generation-constrained microgrids, the following data are needed:

- Historical data such as power usage, weather information such as temperature, and monthly tariff to identify patterns and trends in the usage.
- Demographic information such as the household's total monthly income, education and employment of the household head, total number of household's occupants, number of children, and the number of working members of the household, to understand the influence of socio-economic factors on household consumption.

A survey and PMOG system was deployed to gather demographic and the historical energy data of the households, respectively.

The historical energy data is necessary for the development of the load forecast model, and the proposed scheme for optimally allocating the available energy adaptively while meeting the basic needs of each household. In forecasting the energy usage, the demographic

Table 4.1: Village profiles: The 3 villages have different levels of access to
electricity. Village 1's access to electricity is limited to a 5-hour
window, Village 2's is constrained with daily energy quota, and
Village 3 is connected to the conventional grid.

Village	Location	Electricity	Water	Total HH	Surveyed
	(Philippines)	service	service	with electricity	HH
Village 1	Pangan-an	Yes	No	204	152
	Island,	(5-h daily)			
	Cebu	Off-grid			
Village 2	Paypay,	Yes	Yes	128	121
	Daanbantayan,	(o.8 kWh daily)	(4-h daily)		
	Cebu	Off-grid			
Village 3	Aguho,	Yes	Yes	67	50
	Daanbantayan,	(unlimited)			
	Cebu	Grid			

information of each household is also essential, second to the historical energy data, since it helps the load forecast model to identify the different households according to their profile [Hsi15]. Both historical energy data and the household profile are therefore considered to be essential factors affecting the load forecasting [Gup17]. The monthly tariff can be used as inputs to identify the electricity usage patterns of the households. The corresponding monthly tariffs of each household is reflective of how much electricity they have used for the month. To understand the community's energy usage as a whole, community profile in terms of electricity and water services is presented in the following section.

4.1 OFF-GRID VILLAGES' PROFILES

Three villages were selected in the province of Cebu, Philippines as the communities of interest for this research.

These three off-grid communities are located in remote areas of the province of Cebu that are powered by stand-alone power systems or connected to the traditional grid. Table 4.1 presents the profile of the villages surveyed for this study. The villages were selected according to their electricity access – limited access (off-grid) and unlimited access (traditional grid). A total of 366 sample households (HH) were surveyed, 195 from Village 1 which 152 households are connected to the off-grid power system and 43 of which have no electricity access, 121 from Village 2, and 50 from Village 3.

4.1.1 Village 1

Village 1 is a small island located in Pangan-an, Lapu-lapu city, Cebu, Philippines, with a household total of 405 with an average of 5 members for each family (Figure 4.1). Out of 405 households, only 204 households are connected to the off-grid power system (diesel generators) and only 152 of which participated in the survey along with other 43 households that are not connected to the off-grid power system. Diesel generators are scheduled to operate around 6:00 PM to 11:00 PM daily, and the community has electricity access only at this time. In exceptional cases, the power system can operate in day time upon request with corresponding fees for diesel and other operating expenses. The Pangan-an Island Cooperative for Community Development (PICCD) manages the operation of the power system in the village.

The village used to be serviced by an off-grid solar power plant with a generation capacity of 45 kW. The solar plant was composed of 504 solar modules (90 W), 118 lead-acid batteries with 12V 1800AH rating, charge controllers, inverters and a low voltage distribution system. However, the solar plant stopped operating sometime in 2011 on the recommendation of the Department of Energy Visayas Field Office (DOE VFO) after they had conducted an assessment. According to the assessment of the DOE VFO, the solar plant needed full replacement of the battery bank and 150 solar modules that would cost around PHP 6 million before the plant could go back to its regular operation. The management of PICCD decided to purchase diesel generators to continue to provide electricity to the community instead of doing an overhaul of the solar plant because of financial constraint. Without external help, the community cannot implement the recommendation of the DOE FVO as there is no funding available for them to avail and purchase the necessary components and devices for replacement.

Village 1 has no access to potable water supply within the island. Almost all households store rainwater for daily use. Rainwater is used for bathing, washing of clothes, cleaning and other household activities that require water, except for cooking for which they need to use potable water. Some households use rainwater for drinking after boiling and cooling it. The island sources its potable water from the nearest mainland.

4.1.2 Village 2

Village 2 is located in Paypay, Daanbantayan, in the southern part of Cebu province (Figure 4.2). In 2013, the super typhoon Yolanda

This item has been removed due to 3rd Party Copyright. The unabridged version of the thesis can be found in the Lanchester Library, Coventry University

Figure 4.1: Village 1 is located in a remote island in Lapu-lapu city, Philippines. Houses are typically made of light materials such as bamboos, plywood, and weaved coconut leaves.

devastated several villages, towns, and cities in the Philippines. Many people died, while others lost their homes and livelihoods. Village 2 is located in one of the towns affected severely by the typhoon, and where many people lost their houses. Aside from food and other relief goods, selected survivors of the super typhoon were provided with houses to help them recover faster. The Philippine Red Cross in cooperation with their partners and the local government unit of Daanbantayan, Cebu, built houses in a small community called Red Cross Village for the benefit of a selected few survivors of typhoon Yolanda. The village is composed of 128 households, a daycare centre, and a community livelihood centre. Five mini solar plants service the whole community. Three of the mini solar plants have 33 kW generation capacity, and two have 10 kW that aggregates to 119 kW. With this generation capacity, each household is allocated with 0.8 kWh of energy for daily use (4.1). Each household is installed with energy usage indicator that the households can check the level of their energy usage. The indicator has three levels with different colours. Green for "go" status that means the remaining energy is still high. Orange for "warning" status that the available energy is low and the allocated energy is at a critical level. Red is for "stop" status that means the allocated energy is at an empty level and signifies that the households consume all their allocated energy for the day. Once the red level is breached, the energy management system automatically cut-off the connection of the household to the off-grid power system. The power system is using batteries that serve as energy storage during the day and back-up source during the night. The allocation is done to control the energy usage of the households according to

This item has been removed due to 3rd Party Copyright. The unabridged version of the thesis can be found in the Lanchester Library, Coventry University

Figure 4.2: Village 2 is a community whose houses are built by the Red Cross, Philippines in cooperation with the Local Government Unit (LGU) of Daanbantayan, Cebu for the survivors of typhoon Yolanda. The community has an off-grid solar power that allocates 0.80 kWh as the daily energy quota for each household.

the generation capacity of the available solar plants. The remaining energy is then used to supply the community livelihood centre and the daycare centre.

All households are connected to the local water supply that operates 4-hours daily, 2-hours in the morning (randomly selected) that runs between 6:00 AM to 10:00 and 2 hours in the afternoon that runs between 2:00 PM to 6:00 PM.

4.1.3 Village 3

Village 3 is located in Aguho, Daanbantayan, Cebu, Philippines (Figure 4.3). The community is composed of 67 households and a community livelihood centre. Out of 67 households, 65 are occupied, and five households are not connected to the grid. The community was built by Human Habitat Philippines, a non-government organisation with the aide of their building partners and the local government. This community is another project of the non-government organisation for typhoon Yolanda survivors of the town.

The community is connected to a local water supply operating 24-hours daily.

4.2 DEVELOPMENT OF SURVEY QUESTIONNAIRE (SQ)

In this study, collected data includes energy usage of households in terms of the number of appliances and frequency of use each day, This item has been removed due to 3rd Party Copyright. The unabridged version of the thesis can be found in the Lanchester Library, Coventry University

Figure 4.3: Village 3 is a community built by Human Philippines in coordination with the help of the LGU of Daanbantayan, Cebu. The community is donated to the survivors of typhoon Yolanda. The community is connected to the conventional grid.

apart from their demographic information. Thus, a personal interview or face-to-face survey with a questionnaire was employed.

Aside from gathering the demographic information of the households, the survey is used to:

- determine the household energy usage pattern based on the number of appliances and frequency of usage and
- 2) understand their energy needs in the context of the available services such as water and electricity in their villages.

The questionnaire was designed to have three sections that covered the following profiles of the selected villages: Demographics, Energy, and User-empowerment.

4.2.1 Profiles

4.2.1.1 Demographic Profile

educational attainment,

The Demographic Profile (DP) section includes questions that gather information, such as their:

- role in the family, house occupancy type,
- age, occupation,
- gender,

 number of children in school,

- total number of people living in the household,
- total monthly income of the household.
- total number of people working, and

The demographic information is considered important in determining the different factors that affect the energy consumption of the household. These information were selected based on the works of Hsiao [Hsi15], Chae et al. [Cha+16b], Mandal et al. [Man+06], and Moon et al. [Moo+19] as discussed in Chapter 2.

4.2.1.2 Energy Consumption Profile

The Energy Consumption Profile (ECP) section covers information, such as:

- the list of household appliances,
- the frequency of use,
- monthly electric bill, and
- the list of people who stay at home at night and day with their corresponding ages.

An open-ended question was also included in the ECP section to determine whether their electricity usage would change when there is no power interruption or quota for day use.

4.2.1.3 User-empowerment Profile

The last part of the questionnaire is the User-empowerment Profile (UEP) section. This section allows the participants to rank ten enumerated items according to their priorities. The 10 items are:

•	water,	•	community livelihood pro-
•	sanitation,		gram,

- electricity, health facilities,
- housing,education/school,
- roads,
- employment/job,
 flood/typhoon protection.

These items are considered to be the basic needs of the people according to Maslow's hierarchy of needs [Mas43]. These items also represent the two levels (low and high) of the needs of the people.

Water, sanitation, electricity, housing, and roads represent the low-level needs, and the items represent the high-level needs: employment/job, community livelihood program, health facilities, education/school, and flood/typhoon protection.

In User Empowerment Profile section (UEP), participants awareness level on the impact of their energy use on the environment is also surveyed, as well as their motivation for saving energy. The data from this section is used in comparing the three villages concerning their needs in Section 4.6.2.

The survey questionnaire is attached as appendix A.3 of this thesis.

4.3 DEVELOPMENT OF PMOG SYSTEM

For collecting the energy data of the households, an Energy Monitoring System (EmS) is developed and deployed. The EmS used in this research is based on an existing monitoring system available in the Cogent Laboratory of Coventry University called Cogent-House [Cog15]. The EmS is a modified version of Cogent-House that caters for the requirements of this research, which is to gather the electricity usage of the households in the remote communities powered by an off-grid power system, with limited generation capacity. The Cogent-House system is designed to run individually from a group of sensors communicating with a centralised server, and sending the data by hopping from one local server to another. For this research, the EmS is designed to collect the total household's energy usage using a Current Transformer (CT) jaw sensor of the Current Cost Envi energy monitor, and the appliance energy usage using Individual Appliance Monitors (IAMs). This modified version of the EMS is referred to as the Philippines micro-off-grid (PMOG) system.

Apart from energy usage (house and appliance levels), the PMOG system is enabled to gather environmental data such as the temperature inside the households. The PMOG systems can store and transmit data to a remote server, where it can be accessed and viewed through a web browser. Full details of PMOG system deployment are attached in this thesis as appendix A.4.

4.3.1 PMOG system flow

The PMOG system is composed of the Envi energy monitor and IAMs, gateway (Raspberry Pi (RPi), and internet dongle) as shown in Figure 4.4. The PMOG system uses an RPi 2, Model B, with 1 GB RAM and external internet dongle as its local server devices and internet gateway. RPi is installed with a micro Secure Digital (SD)

card which serves as the storage unit of the system and the house of the codes. The codes govern the system's operations from logging, storing, sending, and displaying data to the webpage intended for the PMOG (see appendix C for the RPi set up guide). The webpage displays power consumption and temperature of the household for a given date and time. The Current Cost Envi energy monitor and IAMs are both off-the-shelf devices. The Current Cost Envi energy monitor consists of two integral parts: 1) the function display monitor with built-in receiver and 2) the CT jaw sensor with energy transmitter [Cur]. CT jaw sensor is clamped to the electricity mainline of the house to monitor the household level energy consumption without altering the existing electricity circuits connections to contact the probe to the electrical lines physically. The Envi energy monitor gathers data from the generated energy of the microgrid, and the total power usage of the selected households in the community. IAMs monitor the appliance level power consumption of each appliance plugged into it. An external antenna is attached to the 4G modem internet dongle to enhance the signal for data transmission. Sensors and the local server communicate wirelessly through the Envi energy monitor. The Envi display monitor displays all current readings of both CT jaw and IAM sensors. All data recorded in the local server are transmitted to the remote server using the internet dongle. The current clamps and IAMs take power reading every 6 s and the PMOG system stores the data temporarily in the local server and push all the recorded data to the remote server every hour (Figure 4.5). PMOG system gathers electricity reading from the household and appliance level every 6 seconds and update the remote server every hour. The PMOG system was deployed without altering the existing set up of the microgrid on both the generation and utilisation side during installation.

4.4 DEPLOYMENT OF THE SURVEY AND PMOG SYSTEMS

The deployment of both the survey and PMOG system were carried out with the help of the University of San Carlos (USC) in Cebu City, Philippines. The survey and the installation of the PMOG to the selected households were conducted between August – September 2016. This research project has undergone the Ethical approval process of Coventry University.

4.4.1 Field survey: Household face-to-face interview

The survey was done in the three remote villages in the Philippines. The survey took 40 - 50 minutes for each respondent. The survey


Figure 4.4: PMOG System Architecture. The PMOG system is designed to collect energy data on both household and appliance levels using CT jaw sensor and IAMs, respectively. Energy data is stored locally and pushed through to the remote server.

questionnaire was written in English, so orientation was conducted for the interviewees to familiarise the questions and get acquainted with the local language translations. Orientation was done to ensure that the interviewees knew how to explain the questions from the survey questionnaire to the respondent when needed, as not all respondents were expected to understand English.

All answers were recorded in English or Cebuano – the local dialect, according to the preference of the interviewer.

4.4.2 PMOG system installation

There were 10 PMOG systems installed in the selected households and microgrid at the two villages; Village 1 and Village 2. These two villages are both powered by an off-grid power system with limited generation capacity. Each village had 5 PMOG systems, 1 for monitoring the generation side and 4 for the households representing the household profiles as discussed in Section 4.8. The maximum number of IAMs installed in each household is 3 for television, electric fan, and DVD players, respectively. Full details of the installation is shown in Table 4.2.

Households were selected after the survey was done where the demographic information was already gathered. The number of households with the same profile was considered in the selection process. Parameters such as number of appliances, number of occupancies and



Figure 4.5: PMOG Data Flow. Each household has 1 PMOG system, as shown in Figure 4.4. CT jaw sensors sense the total energy consumption of the households as they are connected to the main panel and communicate with the local server RPi for storage. Each IAM monitors the energy of the assigned appliance into it and sends the data to the local server through wireless transmission.

This item has been removed due to 3rd Party Copyright. The unabridged version of the thesis can be found in the Lanchester Library, Coventry University

Figure 4.6: Face-to-face interview with the respondents in one of the villages. Participants are asked to rank and give scores to the 10 items as enumerated in Section 4.2.1.3 according to their priorities.

					11 1	т.	1.11
		No.	Appliance	occupants	installed	Income	bill
						(PhP)	(PhP)
Village 1	PMOG1	0	Generator		02-Sep-16		
	PMOG2	0	House 1	7	01-Sep-6	10000	270
		1	TV		01-Sep-16	(£164)	(£ 4.5)
	PMOG3	0	House 2	4	02-Sep-16	14000	525
		1	TV		02-Sep-16	(£230)	(£8.6)
		2	Fan		02-Sep-16		
	PMOG ₄	0	House 3	10	02-Sep-16	17800	800
		1	TV		02-Sep-16	(£292)	(£13.1)
	PMOG ₅	0	House 4	11	02-Sep-16	6500	560
		1	TV		02-Sep-16	(£107)	(£9.2)
VIillage 2	PMOG6	0	Solar Plant		21-Sep-16		
	PMOG ₇	0	House 5	3	21-Sep-16	6500	100
		1	Fan		21-Sep-16	(£107)	(£1.6)
		2	TV		21-Sep-16		
		3	DVD player		21-Sep-16		
	PMOG8	0	House 6	9	21-Sep-16	8000	100
		1	TV		21-Sep-16	(£131)	(£1.6)
		2	Fan		21-Sep-16		
		3	DVD player		21-Sep-16		
	PMOG9	0	House 7	5	22-Sep-16	10000	100
		1	TV		22-Sep-16	(£164)	(£1.6)
		2	Fan	5	22-Sep-16		
	PMOG10	0	House 8	4	22-Sep-16	2500	100
		1	TV		22-Sep-16	(£41)	(£1.6)

 Table 4.2: Summary of installed PMOG systems in Village 1 and Village 2

 Location
 Server
 Device
 Dwelling/
 No. of
 Date
 Monthly Monthly

 No.
 Appliance
 occupants
 installed
 Income
 bill

total household's monthly income were considered in the selection process. The number of households with the same count of those three parameters was tallied. Then a household was selected in each profile to represent them. Figure 4.7 shows the actual PMOG system installed in households. The details of PMOG system installation are attached to this thesis as appendix A.2

The webpage, which is running from the remote server, updates the displayed data every 20 minutes after each hour. When the webpage is not updated, PMOG system may need to be checked to confirm its status.

Figure 4.8 is the actual webpage of the PMOG monitoring system. The webpages display sender, device number, server time for both local and remotes servers, and the readings from IAMs, CT jaw sensors and built-in temperature sensors. Appliances monitored by the IAMS were assigned to a specific channel display in the Envi display monitor in which the devices number is displayed on the webpage. The Envi display monitor can display individual energy consumption of the appliances through IAMs with channel numbers from 1 to 9 with channel o reserved for the CT jaw sensor. The transmitted data can be retrieved from the remote server designated to the PMOG system by accessing the server from any computer connected to the internet. The data is in the form of .csv file that can be viewed with Microsoft Excel. It includes the date and time, electricity usage (power (W)), temperature (degree C), device number, and the server number of the nodes.

The technical report of the deployment is available online for public consumption (http://cogentee.coventry.ac.uk/ gene/).

4.5 DATA PROCESSING

From the six seconds interval, the data is preprocessed to convert the raw data into the targeted interval of forecasting, which is a day ahead load forecasting. The data used for this research is collected from September 2016 to December 2018.

The PMOG data is processed to clean the data and eliminate the possible extreme values recorded from the CT jaw sensor (household level monitoring) and the IAMs (appliances level monitoring).

Table 4.3 shows the summary statistics of the energy data from the PMOG system. With 5 hours operation daily for Village 1, the data points are much lesser when compared with Village 2 which households have 24 hours access to electricity depending on their usage of the given energy quota. For households in Village 2, the energy quota can last for 24 hours when used accordingly. Although



PMOG system installed at the power plant (generation side)



Figure 4.7: PMOG systems deployed in the villages collecting energy data on both the off-grid power systems (generation side) and the households with individual appliance monitors (IAMs).

© M	cogentee.co	ventry.ac.uk	Č	7
🖗 McAfee 📀				
es Status - genepalencia@gmail.com - Gmail			cogentee.coventry.ac.uk/PMOG/	+

Node Report Interface

Sender	Device	Server Time (UTC)	Server Time (Local)	Temp	Power_Usage (W)
PMOG10	0	2016-10-14 10:01:52	2016-10-14 11:01:52	27.80	19.00
PMOG1	0	2016-10-14 10:04:02	2016-10-14 11:04:02	26.00	8930.00
PMOG2	0	2016-10-14 10:04:33	2016-10-14 11:04:33	35.40	31.00
PMOG2	1	2016-10-14 10:04:33	2016-10-14 11:04:33	35.40	9.00
PMOG3	0	2016-10-04 09:57:59	2016-10-04 10:57:59	33.80	76.00
PMOG3	1	2016-10-04 09:57:56	2016-10-04 10:57:56	33.80	0.00
PMOG3	2	2016-10-04 09:57:57	2016-10-04 10:57:57	33.80	0.00
PMOG4	0	2016-10-14 10:04:10	2016-10-14 11:04:10	28.20	183.00
PMOG4	1	2016-10-14 10:04:07	2016-10-14 11:04:07	28.20	5.00
PMOG5	0	2016-10-13 14:04:21	2016-10-13 15:04:21	31.30	31.00
PMOG6	0	2016-10-14 10:03:14	2016-10-14 11:03:14	27.10	1296.00
PMOG7	0	2016-10-14 10:06:34	2016-10-14 11:06:34	27.80	60.00
PMOG7	1	2016-10-12 14:24:59	2016-10-12 15:24:59	28.40	0.00
PMOG8	0	2016-10-08 16:35:12	2016-10-08 17:35:12	28.10	7.00
PMOG9	0	2016-10-14 04:56:11	2016-10-14 05:56:11	28.10	2.00

last loaded: 2016-10-14 10:26:52

Figure 4.8: PMOG system node report interface. The webpage displays the newest transmitted data to the remote server. The data includes the date and time, device number, and PMOG number aside from the main data which are the temperature and power usage of the household.

some households experience power outage as their demand is much higher to what is provided. If households use their energy quota in 5 hours, the number of data points will be similar to that of Village 1. Energy data shows that households from Village 1 have much higher energy usage than households from Village 2, as shown in Table 4.3. This can be attributed to the available energy of each village. Village 1 is not limited to how much electricity they use within the 5 hours operation while Village 2 is limited with the daily quota.

It can be observed that House 3 from village 1 has the highest average power usage of 261.8 W, and House 8 of village 2 has the lowest average power usage of 13.4 W. To confirm that the tabulated average from the PMOG system reflects the expected power usage of each household, reference is made to the survey data. During the survey, the households were asked to list all their appliances and their frequency of use every day or per week. According to our survey data, household 8 has listed a television, and two ceiling fans as their main load aside from 3 light bulbs used during the night. From this list, an average of at least 50 W of electricity usage is expected per day when the television is used. However, from the energy data gathered, the television has not been used since the TV is broken (confirmed with the survey data). This explains why household 8 has an average power usage of 13.4 W instead of the expected average power usage of at least 50 W.

Most of the households have the same electricity load, which include television, electric fan or ceiling fan, light bulbs, and DVD players

i vuinder of					1 000	LI USUE	50, (11)			
data points	Mean	SD	Median	MAD	Min	Max	Range	Skew	Kurtosis	Std. Error
9288	78.2	33.0	93.5	16.6	1.3	146.2	144.9	-1.0	-0.4	0.34
9116	112.5	29.1	116.8	19.0	0.0	610.2	610.2	-0.1	10.6	0.30
7667	261.8	67.3	273.6	80.2	0.0	767.3	767.3	0.4	1.6	0.77
9369	80.5	70.5	75.9	52.2	6.5	718.7	712.2	3.6	19.3	0.73
68657	55.0	53.6	35.7	43.9	0.0	452.5	452.5	1.2	1.3	0.20
82514	22.9	32.0	7.6	7.1	1.0	254.5	253.5	1.8	2.0	0.11
58692	33.0	40.6	14.5	13.1	0.0	378.4	378.4	1.6	1.4	0.17
61136	13.4	11.3	11.0	10.4	0.0	122.9	122.9	1.9	6.6	0.05
	data points 9288 9116 7667 9369 68657 82514 58692 61136	data points Mean 9288 78.2 9116 112.5 7667 261.8 9369 80.5 68657 55.0 82514 22.9 58692 33.0 61136 13.4	Addite points Mean SD 9288 78.2 33.0 9116 112.5 29.1 7667 261.8 67.3 9369 80.5 70.5 68657 55.0 53.6 82514 22.9 32.0 58692 33.0 40.6 61136 13.4 11.3	data points Mean SD Median 9288 78.2 33.0 93.5 9116 112.5 29.1 116.8 7667 261.8 67.3 273.6 9369 80.5 70.5 75.9 68657 55.0 53.6 35.7 82514 22.9 32.0 7.6 58692 33.0 40.6 14.5 61136 13.4 11.3 11.0	data points Mean SD Median MAD 9288 78.2 33.0 93.5 16.6 9116 112.5 29.1 116.8 19.0 7667 261.8 67.3 273.6 80.2 9369 80.5 70.5 75.9 52.2 68657 55.0 53.6 35.7 43.9 82514 22.9 32.0 7.6 7.1 58692 33.0 40.6 14.5 13.1 61136 13.4 11.3 11.0 10.4	Addition Mean SD Median MAD Min 9288 78.2 33.0 93.5 16.6 1.3 9116 112.5 29.1 116.8 19.0 0.0 7667 261.8 67.3 273.6 80.2 0.0 9369 80.5 70.5 75.9 52.2 6.5 68657 55.0 53.6 35.7 43.9 0.0 82514 22.9 32.0 7.6 7.1 1.0 58692 33.0 40.6 14.5 13.1 0.0 61136 13.4 11.3 11.0 10.4 0.0	data points Mean SD Median MAD Min Max 9288 78.2 33.0 93.5 16.6 1.3 146.2 9116 112.5 29.1 116.8 19.0 0.0 610.2 7667 261.8 67.3 273.6 80.2 0.0 767.3 9369 80.5 70.5 75.9 52.2 6.5 718.7 68657 55.0 53.6 35.7 43.9 0.0 452.5 82514 22.9 32.0 7.6 7.1 1.0 254.5 58692 33.0 40.6 14.5 13.1 0.0 378.4 61136 13.4 11.3 11.0 10.4 0.0 122.9	Addition of the second seco	Addition of the second seco	Addition of the second seco

Table 4.3: Descriptive summary for the households daily energy usage Dwelling Number of Power usage. (W)

Legend: SD - Standard Deviation, MAD - Median Absolute Deviation

with speakers. Their electricity usage varies in time of use, frequency of use, and availability of the electricity.

The data is processed by eliminating all extreme values or spikes reading of the PMOG system. PMOG reading beyond 1000 W for house level monitoring is suppressed and 200 W for appliance level. The thresholds are approximate values calculated from the total power rating for the available appliances of the households. The elimination of power surges or spikes are done before the daily average of the power usage is performed.

Daily average power consumption is calculated by getting the area under a curve. A trapezoidal method is used to estimate the daily average power usage. The generalised form of the trapezoidal method in getting the area under a curve in integrating a sine function with N + 1 evenly spaced points is expressed as

$$\int_{p}^{q} f(x)dx = \frac{q-p}{2N} \sum_{n=1}^{N} \left(f(X_{n}) + f(X_{n+1}) \right)$$
(4.1)

where the scalar value of $\frac{q-p}{2N}$ is the spacing between each point. The trapezoidal method is readily available as a Matlab function. This function is then used in getting the average power usage per day.

The calculated daily average power consumption is used as the input data of the forecast model in Chapter 4. The aim is to forecast the next day power usage and used it as one of the basis in determining the optimal energy allocation of the household.

4.6 DESCRIPTION OF DATA

The preprocessing of the dataset is done using R tool, free software for computing and one of the popular programming languages used in

Dwelling	Number of					Temperature, (°C)					
	data points	Mean	SD	Median	MAD	Min	Max	Range	Skew	Kurtosis	Std. Error
House 1	9288	34.0	5.2	35.9	4.9	21.6	39.4	17.8	-0.6	-1.2	0.05
House 2	9116	35.0	3.7	35.9	4.1	23.9	39.4	15.5	-0.6	-0.8	0.04
House 3	7667	30.2	1.4	30.1	1.1	25.4	39.4	14.0	2.5	13.5	0.02
House 4	9369	35.4	4.0	36.8	3.6	22.8	39.4	16.6	-0.9	-0.5	0.04
House 5	68657	28.8	2.2	28.4	2.1	23.6	37.3	13.7	0.8	0.3	0.01
House 6	82514	28.0	2.6	27.5	2.4	21.9	37.3	15.4	0.8	0.2	0.01
House 7	58692	29.5	1.9	29.5	2.0	24.6	39.4	14.8	0.2	-0.4	0.01
House 8	61136	29.6	2.4	29.3	2.4	23.4	39.4	16.0	0.5	-0.2	0.01

Table 4.4: Statistics of households temperature data

Legend: SD - Standard Deviation, MAD - Median Absolute Deviation

the field of Data Science [Rto]. Tables 4.3 and 4.4 present the statistics of the household's energy consumption and the temperature data. The energy consumption and temperature data of the households were collected from September 2016 to December 2018. For the forecasting model, the dataset was divided into 70:15:15 ratio for training, test, and validation purposes. Both energy consumption and temperature data were used as inputs for the forecast model, as presented in Chapter 5. All data are stored and available for public consumption at Cogentee repository (url: http://cogentee.coventry.ac.uk/ gene/researchdata/) [Pal19].

4.6.1 Energy data yield

The average data yield every day is determined to check if the data points collected are adequate to represent the daily energy usage of the households. Figure 4.9 shows the average data yield for each PMOG system installed in selected households at village 1 and village 2. The average yield of the data collected daily is above 80 %, as shown in Table 4.5.

For any system that collects data, missing values are often one of the issues. There are many possible reasons as to why missing values occur. This issue can be attributed to the system's malfunction (i.e. short circuit, system breakdown), interrupted operation (i.e. systems are turned off), loss of contact (i.e. lossy cable), or no internet connection for wireless transmissions. All these are difficult to address within the system itself to avoid all the possible causes of missing data. In this study, to ensure that daily energy usage of the household is represented well by the data points from the PMOG system, daily data points must at least have 80% yield. This is equivalent to 4 hours of electricity usage in a day for Village 1 and 19.2 hours for Village 2. For Village 1, the average daily yield is above 90% and for Village 2 is above 80% as shown in Table 4.5. A data yield of at least 80%



Figure 4.9: PMOG system data yield. The daily yield of the data in % (y-axis) for each household (x-axis). The daily average of yield (%) is above 80% for all PMOG systems installed in the households.

Ν	Daily yield							
	mean	std. dev	min	max	median			
307	95.17	14.93	22.55	100.00	100.00			
315	94.18	17.33	10.94	100.00	100.00			
232	94.94	18.12	2.99	100.00	100.00			
314	95.90	14.06	22.55	100.00	100.00			
266	94.23	16.40	8.33	100.00	100.00			
314	95.81	14.72	2.78	100.00	100.00			
263	81.52	23.00	0.35	100.00	89.58			
246	89.18	24.97	0.35	100.00	100.00			
	N 307 315 232 314 266 314 263 246	N mean 307 95.17 315 94.18 232 94.94 314 95.90 266 94.23 314 95.81 263 81.52 246 89.18	N D mean std. dev 307 95.17 14.93 315 94.18 17.33 232 94.94 18.12 314 95.90 14.06 266 94.23 16.40 314 95.81 14.72 263 81.52 23.00 246 89.18 24.97	N Description mean std. des min 307 95.17 14.93 22.55 315 94.18 17.33 10.94 232 94.94 18.12 2.99 314 95.90 14.06 22.55 366 94.23 16.40 8.33 314 95.81 14.72 2.78 263 81.52 23.00 0.35 246 89.18 24.97 0.35	N Description mean std. dev min max 307 95.17 14.93 22.55 100.00 315 94.18 17.33 10.94 100.00 312 94.94 18.12 2.99 100.00 314 95.90 14.06 22.55 100.00 314 95.91 14.06 22.55 100.00 314 95.81 14.72 2.78 100.00 314 95.81 14.72 2.78 100.00 314 95.81 23.00 0.35 100.00 314 81.52 23.00 0.35 100.00			

Table 4.5: Daily yield of PMOG systems

is considered necessary to build the forecasting model to maintain the integrity of the data as logged by the PMOG system for each household that is monitored.

Figure 4.10 shows the actual electricity usage of one of the households from Village 1 over 5 hours that the off-grid power system is operating. It can be observed that the household immediately uses electricity as soon as the power system is up around 6:00 PM. Power level variations reflect certain changes in electricity usage of the households. From Figure 4.10, the house-level power lever changes as soon as the TV is turned on and off. When the electric fan is turned on, total power usage increase reflecting the power consumption by the fan on house level.

According to the survey data [Pal19], households usually use televisions during dinner time (around 7:00 PM) until the last TV program they like which is generally around 9:30 PM to 10:30 PM daily. This can be observed that the power usage in this hour is high, as shown



Figure 4.10: Actual electricity usage pattern of house 1 from Village 1. Houselevel power usage (y-axis) varies as the usage of TV and electric fans change reflecting the power usage of each appliance over time (x-axis). Data: 1 day (5 hours) electricity usage.



Figure 4.11: Households energy usage pattern. House level power usage (y-axis) of the selected households in village 1 varies in time (x-axis) when the load is changed. Most of the households use electricity as soon as the power system is up. Some households maintain almost the same load throughout the operation of the power system (e.g., House 3); thus very few variations on power usage is reflected.

in Figure 4.10 Otherwise households just use light bulbs, electric fans and phone chargers until the operation cuts down.

Figure 4.11 shows the house-level power usage of the selected households in village 1. The variations in electricity depend on how the household uses their appliances at any time. The house level power usages reflect the general characteristics of the electricity usage of the household. According to our survey, people in village 1 use electricity as much as they can during the operation time of the off-grid system. From figure 4.11, it can be observed that some households register power usage higher than 100 W just before the cut off time. According to the survey data, correspondents used electricity by watching television until the end of operation of the microgrid. This explains the 100 W usage of power right before the cut-off time.



Figure 4.12: Daily energy usage pattern. Power usage (y-axis) at the house level varies significantly for the entire day (x-axis) for each household. Most of the households peak their usage of electricity around 6:00 PM to 9:00 PM (dinner time)

For Village 2, households have an energy quota every day, which is 0.8 kWh. Households are given the responsibility to use their energy quota wisely if they want the 0.8 kWh energy to last for 24 hours. As shown in Figure 4.12, each household reflects their power usage at different times of the day. Figure 4.12 shows high power usage happens between 6:00 AM and 10:00 in the evening.

Figures 4.11 and 4.12 show an actual electricity usage of the households in Village 1 and Village 2, respectively for a given day. A similar usage can be observed on the households in Village 1 from 7:00 PM up to 9:00 PM where their electricity usage peaks. For households in Village 2, most of them peak their electricity usage from 6:00 PM to 9:00 PM. Since Village 1 has electricity available for them from 6:00 PM until 11:00 PM only, all households utilise the available power for the entire duration of the operation of the microgrid, showing a constant electricity usage of the households. However, for Village 2, electricity can be available for 24 hours depending on their usage as they are given a specific energy quota daily. From Figure 4.12, households have several peaks of electricity usage spread in 24 hours compared with households from Village 4.11 that depicts a stable usage from the beginning of the operation of the microgrid.

Both villages have a fixed amount of monthly fees for the corresponding energy allowance. Hence the impact of the monthly bill may have minimal influence on how they use their energy allowance. The electricity usage pattern of Village 1 is perceived to be almost constant for the entire duration of operation of the microgrid however Village 2 presents interesting electricity characteristics as the usage varies for 24 hours depending on their usage of the given energy quota. According to the survey data, 46 % of the households experience a power outage,

Household composition	Village 1	Village 2	Village 3
Age			
less than 5 years old	9.81	13.31	15.63
5 to 18 years old	34.97	36.53	38.67
more than 19 years old	55.23	50.15	45.70
Attending school	31.48	35.45	37.11
Working (with regular income)	5.55	7.89	5.86

Table 4.6: Household composition: Age, household members working and attending school (%).

and 54 % are satisfied with the existing quota and do not use all up the given energy quota daily.

4.6.2 Description of survey data

Here, survey data are the responses recorded from the conducted face-to-face interviews with the households in the three villages based on the survey questionnaire. As mentioned in Chapter 2, the survey questionnaire gathers the demographic profiles of the households. In this section, the tabulated responses of the correspondents are presented.

4.6.2.1 *The respondents*

All respondents are identified to be mothers, fathers, or head of the households. This is to ensure that the responses to the questionnaires reflect the household's conditions and status on energy usage. Table 4.6 shows the breakdown of the age of all the participants and their household members, as well as the number of family members working and attending school.

Village 3 has the highest percentage of children attending school with 37.11 %, followed by Village 2 (35.45 %), and Village 1 (31.48 %), as shown in Table 4.6. These percentage values indicate that more than 1/3 of the total population of the villages are children and attending school. Village 1 has the lowest portion of the household members with ages less than 5. These children stay at home with their mother or father attending to them. The number of family members with regular work and sources of income, aside from the household head and spouse, is few with only 5.55 % in Village 1, 7.89 % in Village 2, and 5.86 % in Village 3. Most of the adults are working on a contractual basis with no regular source of income.

Characteristics	Village 1	Village 2	Village 3
Average age (years)			
Household head	44.69	42.19	41.27
Spouse	44.85	40.34	38.93
Average level of education (completed years)			
Household head	7.92	9.27	8.31
Spouse	7.41	8.93	8.09
Gender of household head (%)			
Female	28.48	16.53	6.67
Male	71.52	83.47	93.33
Average number of occupants per household	6	7	7

Table 4.7: Characteristics of household head and spouse, and average number of occupants per household.

4.6.2.2 Characteristics of household head and spouse and average number of occupants per household

AS shown in Table 4.7, overall, household heads across all three villages finished at least primary level of education (average education years is 8.50). In Village 2, household heads have average education years of 9.27, the highest of the three villages, indicating 3rd year level in high school, followed by Village 3 with 8.31 and Village 1 with 7.92.

Across all villages, female heads are only 17.22 % while male heads are 82.78 %. Village 1 has a relatively high female head of 28.48 %, followed by Village 2 with 16.53 %, and Village 3 has 6.67 % (the lowest). On the other hand, the average number of household occupants for both Village 1 and Village 2 is 7, 1 point higher than Village 1, which has 6.

4.6.2.3 Households income and household head occupations

The majority of the household heads in the selected three villages earn less than the poverty threshold and some even less than the food thresholds (Table 4.8 and Figure 4.13). Both food and poverty thresholds are set by the Philippine Nutrition and Food Research Institute. The food threshold is defined as the minimum monthly income needed to meet the monthly food needs and the nutritional requirement for a Filipino family with five members. The poverty threshold is an expanded food threshold to include non-food needs such as clothing, housing, education, transportation, and health expenses.

As shown in Table 4.7, the average number of occupancy per household for the three villages is higher than 5. From the given thresholds shown in Figure 4.13, there are more than half of the surveyed households in all three villages living with a monthly income under the Food threshold. Table 4.8 shows the breakdown of the monthly income

Table 4.8: Monthly income (range) of the households										
Income Range	Households (%)									
(PhP)	Village 1	Village 2	Village 3							
Less than the food threshold	68.88	59.84	60.47							
(6,300.00), approx. £87.5										
Within the food and poverty	10.71	24.59	27.91							
threshold (6,301.00 ~ 9,100.00),										
approx. £87.51 ~ £125)										
Above the thresholds	19.90	13.93	16.28							
(9,101.00), approx. £125.1 and up										



Figure 4.13: Monthly income of the 3 villages compared with the poverty and food thresholds.

	10	icentage ()	(0)
Occupation	Village 1	Village 2	Village 3
Manager, supervisor, official, professional	0.66	4.96	
Technician or assoc. professional	6.58	4.96	6.67
Clerk	1.32	2.48	2.22
Service, shop, or market sales worker	17.11	14.88	
Farmer, forester, fisher	45.39	19.01	48.89
Trader	11.84	0.83	
Plant or machine operator or assembler	0.66	2.48	
Laborer or unskilled worker	5.92	41.32	33.33
Housewife	3.29	5.79	2.22
Shellcraft/Special occupation	7.24	3.31	6.67

 Table 4.9: Percentage (%) of Households, by occupation of the HHs head

 Percentage (%)

of the households in each village. There are 68.88 % from Village 1, 59.84 % from Village 2, and 60.47 % from Village 3.

Poverty is common to most of the families in the communities who make a living from agriculture, construction, fishing and any other related activities as shown in Table 4.9. The average income of the three communities, as shown in Figure 4.13, is below the national minimum monthly income, which is PHP 9,064 (£145) and is referred to as the poverty threshold. There are approximately 82 % of the respondents have a monthly income below the poverty threshold. There are 15 % of them earning between PHP 9100 (£150) and PHP 20,000 (£325), and only three % of the households have monthly income more than PHP 20,000 (£325).

The household heads commonly work as farmers, fishers, foresters, labourers, and unskilled workers (Table 4.9). Since most of the household heads completed high school education only (Table 4.7), the occupations they engage in are limited to jobs that do not require a college degree. The income from this type of job is minimum wage, and sometimes even less than the minimum wage mandated by the law.

4.6.2.4 Households electricity experience

One of the open-ended questions in the survey questionnaire is "If you will have 24-hour access to energy, what would change in your energy consumption?" in the pretext that the energy available is unlimited. This means that for Village 1, electricity will be available 24 hours daily and not 5 hours and for Village 2, there will be no quota. Their electricity usage will be similar to the households from Village 3 that are connected to the traditional grid.

For Village 1, 96% of the respondents when given to have 24 hours access to electricity want to use it during the day as the temperature

in the island can be too warm. With electricity, they can use some electric fan to cool down. Other reasons include to be able to use a refrigerator for the food storage, to watch television any time of the day for entertainment purposes, and want to use electricity for business purposes. The other 4% of the respondents are quite happy with the 5 hours of electricity access.

For Village 2, 83 % of the respondents want to have 24 hours access with no quota so that they can use more electricity for food storage, entertainment, business, and for their children's education. They believe that 24 hours of access can improve the study habits of their children that go to school. For the other 17 %, the existing energy scheme is sufficient for their electricity needs; however if the electricity will be available 24 hours similar to the traditional grid, they may change their electricity usage. Currently, there are 46% of the households in Village 2 that experience electricity outage with the given quota and wanted to have more energy when possible. Other households are quite satisfied as they do not use all up their daily quota, these comprise the 54% of households connected to the microgrid but express willingness to change their electricity usage when the access has no limitation.

Village 3 enjoys electricity access 24 hours as they are connected to the traditional grid, and the only factor that limits them to use electricity is their capability to pay. Their income dictates on how much electricity they use.

4.7 ENERGY USAGE OF THE HOUSEHOLDS

In this section, the average energy usage of each household in Village 2 is investigated as a case study for this research. The data from Village 1 was dropped as the electricity usage of the households is almost constant for the entire 6 hours operation of the microgrid. What makes the data from Village 2 interesting is the fact that households use different electricity in 24 hours depending on their available energy quota daily. The households are given the responsibility to utilise the given quota that would serve them best for 24 hours. It is hypothesised that some households are using energy twice more than the other households. Moreover, these households are expected to experience an energy shortage. From the survey data that 46 % of the household experience electricity outage daily as their energy quota is not sufficient for their needs and there are 54 % of the households that used all up their energy quota daily as they used less electricity than what is given. Hence the motivation of this work can be simplified from this reality of Village 2 that the excess energy quota of the other households can be used by other households that need more energy.



Figure 4.14: Average energy usage of the households in Village 2

To determine the actual energy usage of the households, the average energy usage of the households are determined from the available energy data of PMOG systems. Figure 4.14 shows the average energy usage of households. It can be observed that house 5 has the highest average energy usage and house 8 has the lowest. House 5 also presents the characteristics that it uses beyond the quota given. At first, it was thought that the PMOG monitoring system has some faulty reading the electricity usage of the households. However, it was found out that the household is allowed to have more than the quota being the president of their association in the village. The electricity usage of House 5 is also reflective of the demand of households with a similar profile. On the other hand, House 8 electricity usage is too little when compared to House 5. House 8 is one of the households that do not use all up their given quota daily. According to the survey data, House 8 uses electricity for lightings and ceiling fan only. The light bulbs used for lightings are rated 10W, and the ceiling fan is 15W. House 6 and House 7 can be considered an average user that most of the time, use their energy quota efficiently. Sometimes they experience a power outage, but most of the time they do not use all up their energy quota.

4.8 HOUSEHOLD PROFILES

This section identifies the features of the household profile that would be used as the consumer profiles for the development of the load forecast model in Chapter 5. The forecast model is to predict the next day energy usage of the households. The data needs to represent the 24 hours usage fo the households. Since Village 1 operates for 5 hours only, this does not meet the required 24 hours usage; hence the data was dropped. The features are selected from the survey data that are considered to be significant factors of influencing households use of

	Tuble 4.10. Household promes											
Profile	Occupants		Average		Family members							
		Monthly No. of		Freq. of use	Working		Home					
		income	appliances	(hr/day)		School						
А	1 ~ 3	4987	4	23	1	0	1					
В	$4 \sim 6$	7056	4	23	1	2	2					
С	$7\sim 8$	7030	4	26	2	3	2					
D	9 up	6529	3	17	3	5	3					

Table 4.10: Household profiles

energy. From Section 5.2.3, demographic information survey data, and from the literature review, the following factors (shown in Table 4.10) are selected to be the main features of the household affecting their energy usage. There are four household profiles from grouping the number of occupants of the household. The first group has one to three occupants, the second group has four to six occupants, the third group has seven to nine occupants, and the last group has ten or more occupants. These groups represent the size of the family members occupying a house. According to the study of Hu et al. [Hu+17], the family size and the number of appliances are major influencers of energy consumption for Chinese urban households based on the results of their online survey for 4964 households. Other features as discussed in Chapter 2 such as the monthly income, number of appliances and its frequency of usage, number of family members working, schooling and staying at home were selected from the survey data. Table 4.10 shows the full details of the household profiles considered as inputs aside from the historical energy data, temperature and the weekdays of the forecast model developed in Chapter 5. Households are then selected to represent each profile and to cover the variability of the households in Village 2. House 5 is to represent profile A, House 6 is for profile D, House 7 is for profile C, and House 8 is for profile B.

These data are then used for the training of the MLP-based neural network presented in Chapter 5 to enable the model to predict the energy consumption of each household.

4.9 CHAPTER SUMMARY

In this chapter, the process of gathering the data and initial analysis of the data is presented. All the necessary data needed for the development of a load forecast model is successfully collected using the developed survey questionnaire and the PMOG systems installed in the selected households. The processing of historical energy data is presented. The processing of the data is done to eliminate the extreme values that may affect the real energy consumption of the households. The extreme values were eliminated by applying a threshold on both house level and appliance level monitoring based on the power rating of the available appliances of the households.

The PMOG data from Village 1 and Village 2 is scrutinised and studied well to deduced the electricity usage of each village that is connected to the microgrid. Both villages are powered with different microgrids that are both operating offline. Village 1 is powered by a diesel generator that operates 5 hours daily during night time while Village 2 is powered by a solar power plant that provides energy quota daily to the households. This energy quota allows the households to use the energy for 24 hours depending on their usage. Some households use all up the given energy daily, and experience power outage while other households are satisfied with the existing energy management scheme and do not use all up the given energy quota. There are 46% of the households in Village 2 that experience power outage and 54% do not use all up the given quota. The excess or unused energy quota was given to those households are considered energy wastage that could be used by other households that experience the power outage. Given this scenario, the need to optimally allocate the available energy is considered. This is explained in details in Chapter 6.

The data is also processed to create the necessary input matrix needed for the neural network and the other techniques considered in this research such as RBFN, GPR, and ARIMA.

With the real-world data gathered from the conducted survey and the installed PMOG system to the households, this research is achieving realistic results on the proposed optimal daily energy allocation for households connected to microgrids with limited generation capacity as discussed in Chapter 6. The proposed optimal energy allocation is derived by calculating the optimal value in the formulated optimisation problem considering the constraints posed by the power systems and the number of households connected to the microgrid. The details of this optimisation problem are presented in Chapter 6. The needs of having an optimal energy allocation are also presented in Section 4.5 where some households consume more energy than the others. Also, in this chapter, the consumer profiles are identified as discussed in Section 4.8.

The next chapter presents the process of development of the load forecast model.

5

MULTILAYER PERCEPTRON-BASED FORECAST MODEL

Chapter 4 presents the process of gathering energy and survey data. These data are used for the development of energy forecast model and consequently for the development of optimal energy allocation scheme of daily energy allowance of each household in generation-constrained microgrids, as shown in Chapter 6. Generation-constrained microgrids are microgrids with limited generation capacity. This type of microgrids uses energy management restrictions such as energy consumption quota to provide the basic electricity needs of the households while maintaining a balanced operation. Microgrids that can supply the electricity needs of the households without restrictions are considered to have unconstrained generation capacity.

This chapter presents the development of the energy forecast model based on artificial neural networks (ANN), specifically the multilayer perceptron (MLP). When using the neural network, choosing the proper neural network sizes can be complicated as these affect the performance of the network. Hidden layers and number of neurons influence the networks' forecasting performance in terms of rootmean-square error (RMSE) along with the number of delay window considered in the input stage. The process of selecting values for these three parameters is discussed first. The process includes on how the number of hidden layers HL, number of neurons N, and the number of delay window D affect the performance of the MLP-based forecast model in predicting the next day energy usage of the households. This procedure is done to all input considered in the modelling of MLPbased forecast model. MLP is preferred for being robust in handling real-world data compared to the other techniques mentioned and does not require a priori knowledge as most statistical-based approaches. The data used in developing the forecast model is from village 2, where the data reflects the 24-hours energy usage of the households.

This chapter aims to find the answers for the research question,

RQ1: Can the household's daily energy consumption be forecast with reasonable accuracy?

To answer the above research question, experiments were conducted using four forecasting techniques namely MLP, Radial Basis Function Neural Network (RBFN), Gaussian Progress Regression (GPR), and



Figure 5.1: Load forecasting model development flow.

Autoregressive Integrated Moving Average (ARIMA). For MLP-based forecast model, simulations were conducted to determine the combination of hidden layer HL, neurons N, and delay window D that generates the least values of RMSE. The RMSE value should meet the threshold set for the forecast model that is 15 % of the actual energy usage. The threshold is set based on the acceptable RMSE as laid out in the work of Hsiao [Hsi15] and Moriano et al. [Mor+16], 20%. All forecast models were developed using the same data as with the MLP-based model for comparison purposes. The performance of the proposed MLP-based forecast model is also compared to existing works of Marnerides and Din [DM17] and Ryu et al. [RNK16].

All forecast models are trained and validated using the data gathered from the survey and the PMOG monitoring systems, as presented in Chapter 4. Figure 5.1, shows the flow of the development of the load forecasting model developed based on the MLP neural network. Gathering the data is described in chapter 4, as well as the pre-processing of data to eliminate unreasonable readings from the PMOG systems. The succeeding sections described the process to determine the proper size of the neural network for parameters such as hidden layers and number of neurons and the appropriate number of historical data delay window D. The results were graphed and tabulated to compare the RMSE results. At the end of the experiment, the combination with the least RMSE values is chosen for the MLP-based forecast model.

5.1 FINDING THE BEST COMBINATION OF NEURAL NETWORK PARAMETERS

It has been established in Section 2.6.2 that the design parameters of the neural network affect the performance of the network in forecasting. Particularly, design parameters such as hidden layers HL, number of neurons N, and delay D, as well as the training data, affect the forecast model performance and generality [Cha+16b]. In this regard, the best combination of the three parameters must be determined to improve the forecast accuracy based on the available data for training. For the neural networks, the selection of the network size is not a straightforward process. The traditional approach usually starts with the simplest architecture and gradually explores complex structures. This approach is followed in the development of the MLP-based forecast model.

The neural network MATLAB toolbox is used in the development of the model. The toolbox's *train* function is used to train the network using the Levenberg-Marquardt algorithm. To identify the values of hidden layers HL and the number of neurons N that would generate the lowest RMSE possible; a heuristic approach is employed. The three variables were considered in the simulations of the model using MATLAB toolbox. A reasonable range for hidden layers HL, number of neurons N and delay D is considered to avoid over-fitting that may lead to erroneous load forecasting.

The process of selecting the values of hidden layers HL, the number of neurons N, and delay D to generate the best performing forecast model is presented in the next section.

5.1.1 Steps in determining the best combination of hidden layers HL, number of neurons N, and delay D?

Following the guide on the design procedure of the neural network for forecasting from the work of Palmer et al. [PJSo6], below are considered important when designing and developing the network for the MLP-based forecast model.

 Determining the number of outputs: Any MLP neural network model can have one or more outputs that could correspond to the desired outcome from the model. It could represent the hourly, daily, weekly, monthly, or annual energy consumption. In this work, a single-output model with the output being the next day energy consumption (kWh) of the households is used.

- 2.) **Identifying the number of inputs:** Determining the inputs before the development of the model is important. The inputs are selected depending on the available data. In this study, the inputs for forecasting the daily energy usage of each household include the historical energy data, weather variables such as temperature, and calendar such as days of the week.
- 3.) **Identifying the number of hidden layers (HL):** There are no existing common rules in determining the suitable number of hidden layers of the MLP. In this work, a heuristic approach is used to identify the appropriate number of hidden layers with the best performance in terms of root mean square error (RMSE). From the selected range of the HL, more than 100 combinations were used during the simulations.
- 4.) **Identifying the number of neurons per layer (N):** Same with the hidden layers, there are no common methods in determining the number of neurons per layer that would guarantee a good performance of the network in forecasting. A heuristic approach is also employed in identifying the number of neurons for the MLP with the ultimate goal of having the best performance in predicting the energy consumption of the households.
- 5.) **Identifying the number of delays, (D):** The number of delays refers to the window of the inputs to be considered in reshaping the historical energy data to form the input matrix. Determining the right number to be used in the network together with the hidden layers HL and the neurons N, requires more than 100 simulations from the selected range of 1–30 days of equivalent energy usage.

The simulation is done using a heuristic approach, and then the selection is made from the tabulated results where the model with the smallest RMSE is chosen. In this research, the desired level of accuracy corresponds to an RMSE of less than 15% of the 0.8 kWh (existing allocation), which is equivalent to an energy of 0.12 kWh or 120 Wh. This amount is considered to be the allowable error for forecasting the energy usage of the households. RMSE is a measure of the difference between the predicted and the actual value over the number of observed data as described by equation 5.1. This error represents the average difference between the predicted values and the actual values. The 15% error is the maximum error allowable based on the previous usage of the households as gathered from the PMOG

systems. Thus, the 15% error or less is desirable for forecasting the next day usage of the households.

$$RMSE = \sqrt{\sum_{i=1}^{n} \frac{(\hat{E}_{i} - E_{i})^{2}}{n}}$$
(5.1)

where \hat{E}_1 , \hat{E}_2 , \hat{E}_3 , ... \hat{E}_n is the predicted values, E_1 , E_2 , E_3 , ... E_n is the actual values, and n is the number of observations.

The number of neurons was simulated in the range of five to thirty with 5-unit steps. This range was chosen to observe the performance of the network through its RMSE. The range is considered to be reasonable to avoid the complexity of the network and overfitting due to a large number of neurons in the network. On the other hand, the delay was set in the range of 1 to 21 with 2-unit step. The neural network given with knowledge of a household's current energy usage, the future usage (state) may depend on the previous consumption (states) only to a small degree, i.e. the behaviour is Markovian up to a certain point. Therefore, few days historical data should suffice to have an acceptable accuracy in forecasting. For the number of hidden layers, the network was trained with hidden layers ranging from one up to five hidden layers. This range was chosen to have a possible model that is shallow (with 1 to 2 hidden layers) and not so deep (with 4 to 5 hidden layers) neural network.

To validate the forecast model a k-fold validation was performed with k equal to 10 in all the combinations of inputs used in determining the best combinations of the three parameters that would allow the neural network to provide the lowest RMSE. This is to make sure that each observation from the original data has the chance to appear in training and testing datasets [Koh95]. The validation method works, as shown in Figure 5.2 where I is the number of iterations.

5.1.2 Input shaping (delay D) and the effect of the network size on the performance of the model in forecasting

This section aims to investigate the effects of the number of energy historical data points considered as input window or delay (D) to the performance of the neural network in load forecasting.

The performance of the network can be improved or worsen depending on the number of delay window D. If the value of delay D is too small, there may be not sufficient information to make an accurate prediction, and if the value of D is too large, there may be over-fitting.

To avoid over-fitting, simulations were conducted. The number of delay window D is varied to observe the effect of delay D on



Figure 5.2: K-Fold Cross Validation

the performance of the network and to determine the best value of delay D, which generates the lowest RMSE. The concept of delay D is shown in Figure 5.3. For this research, the experiment is done by considering the past points of historical data to predict the next day energy consumption. This also serves as the input matrix of the MLP-based forecast model.

For example, if N = 10 and D = 5, so

Energy data: $P_1, P_2, P_3, P_4, P_5, P_6, P_7, P_8, P_9, P_{10}$ Input: Output: $P_1, P_2, P_3, P_4, P_5 \rightarrow P_6$ $P_2, P_3, P_4, P_5, P_6 \rightarrow P_7$ $P_3, P_4, P_5, P_6, P_7 \rightarrow P_8$ $P_4, P_5, P_6, P_7, P_8 \rightarrow P_9$ $P_5, P_6, P_7, P_8, P_9 \rightarrow P_{10}$

Suppose the energy consumption for 5th of May 2020 is to be predicted with D equal to 5, then the historical data from 30th of April 2020 to 4th of May 2020 will be taken into consideration.

RMSE is used as the performance metric of the MLP-based forecast model [CD14]. RMSE is the measure of difference between the actual energy consumption (E_{α}) and the forecasted values (E_i). Hence, smaller value of RMSE is preferred. The lower the RMSE, the better the performance of the model to forecast. There are several sets of input tried during the simulation to observe the effects of the input to the performance of the neural network. Since the off-grid system is using batteries, the generation side is considered constant and hence



Figure 5.3: Input delay window concept where DW_1 is delay window 1, DW_2 is delay window 2 and DW_m is the total number of delay windows for the given input series.

the variations caused by the solar panels and other factors that affect the generation of the electricity is neglected. Below is the list of inputs (and its combinations) used in developing the forecast model:

1.) Historical energy data (PMOG data) such as power consumption of the household

2.) Household's temperature (indoor)

3.) Calendar days such as weekdays

4.) Household profiles from the survey data (demographics, and monthly energy tariff)

5.1.3 Standard load forecasting using historical energy data as input

This section aims to investigate how the delay D affect the performance of the neural network in forecasting. Simulations were conducted using the daily average energy consumption of the households as inputs. It assumes that the number of D significantly affects the performance of the forecasting model. Figure 5.4 shows the results of the simulation with the network's number of neurons are 15, and the hidden layer is one. It can be observed that the performance of the network varies as the delay D increases. At some point, it may seem that the RMSE values decrease as the delay D increases; however, RMSE standard variations are also changing randomly. The lowest average RMSE for



Figure 5.4: RMSE (Wh) results where x-axis shows the delay D and y-axis is the RMSE (Wh) for each delay D showing variations of a k-fold validation.

this network is 0.102 kWh or 102 Wh with a standard deviation of 9 for D = 21. Table 5.1, shows the performance of the network for each neuron considered in this part of the experiment. Across all the values neurons together with values for Figure 5.4, the lowest RMSE is 0.099 kWh or 99 Wh with a standard deviation of 6 for a network with one hidden layer, 30 neurons and historical energy data of 11.

To observe further the effects of the hidden layers and neurons along with the number of historical energy data considered as inputs, the number of hidden layers is increased to 3 and then 5. Tables 5.2 and 5.3 show the RMSE results as the number of hidden layers is increased with historical energy data as the inputs. For a neural network with three hidden layers, the lowest RMSE is 87 Wh with a standard deviation of 22 for a network with 30 neurons N and the delay D is 21. On the other hand, for a network with five hidden layers, the lowest RMSE is 85 Wh with a standard deviation of 15 for a network that 30 neurons and 21 delays. For both networks, the network's performance is at its best when the number of neurons is 30, and D is 21. Both values are extreme values considered in the simulations. Same trends are observed for the values of RMSE that is decreasing as the values of both neurons, and D are increasing with big variations as reflected in the standard deviations of each RMSE corresponding to each neuron N and delay D.

Since the goal is to find the best combination of neurons per layer N, hidden layer HL, and the delay D that generates the lowest RMSE

	Number of Neurons, N											
	5		10		15		20		25		30	
Delay, D	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD
1	116	6	116	6	116	7	116	6	115	5	115	7
3	108	9	109	7	108	10	107	9	108	7	107	8
5	108	10	107	7	106	7	106	8	105	8	103	10
7	106	8	105	9	106	8	102	9	103	7	104	9
9	105	7	105	6	105	5	106	5	102	8	102	6
11	107	7	103	6	104	6	102	7	104	7	99	6
13	106	10	103	7	105	8	104	8	100	9	103	9
15	104	10	107	9	106	11	100	6	101	9	103	8
17	103	11	106	8	106	10	101	11	106	6	99	12
19	105	11	105	11	103	8	99	9	104	10	105	11
21	106	10	104	12	102	9	101	12	99	12	103	13

Table 5.1: Average RMSE (Wh) and the standard deviation as D increases for each neurons of neural network with 1 hidden layer

 Table 5.2: Average RMSE (Wh) and its standard deviation as the delay D increases for each neurons of neural network with 3 hidden layer

Number of Neurons, N Delay, D RMSE SD RMSE SD RMSE SD RMSE SD RMSE SD RMSE SD 107 11 106 12 104 12 105 10 102 10 101 11 102 10 98 10 97 13 95 13 97 10 94 11 87 18 100 11 103 11 103 11 91 13 106 14 104 11 101 12 84 24 97 14 86 18 96 13 96 11 105 11 100 14 107 12 104 11 100 12 91 14 95 15 107 10 105 12 104 10 94 13 102 12 87 22

	Number of Neurons, N												
	5		10		15		20		25		30		
Delay, D	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	
1	116	6	116	7	115	6	116	10	115	4	115	5	
3	108	10	108	8	104	11	106	9	106	11	104	8	
5	107	6	104	9	104	9	103	11	99	13	101	12	
7	105	7	106	8	104	9	100	11	103	7	103	11	
9	106	6	101	8	99	7	99	8	100	7	104	9	
11	106	8	101	9	98	9	98	5	100	10	96	9	
13	105	8	101	9	98	10	97	8	102	12	94	13	
15	105	10	101	11	92	15	98	14	97	12	95	12	
17	102	12	102	9	102	12	92	14	98	12	96	15	
19	105	9	99	13	92	13	98	8	100	14	94	10	
21	101	12	105	10	94	13	100	12	96	14	85	15	

Table 5.3: Average RMSE (Wh) and its standard deviation as the delay D increases for each neurons of neural network with 5 hidden layer

possible, this process is repeated for all input (temperature, weekdays, and households profiles) and then the acceptable RMSE is chosen.

Note: From this point onwards, the results presented are from MLP-based forecast model with one hidden layer unless otherwise specified. A neural network with one hidden layer is widely accepted in forecasting as discussed in Section 2.6.2.

5.2 IMPROVING THE PERFORMANCE OF MLP-BASED FORECAST MODEL

As mentioned in Chapter 2, factors affecting load forecasting include weather information such as temperature, calendar days such as days of the week, and household demographics such as income, number of household occupants, number of working people and number of children still in school. The next section presents the performance of the neural network, as these inputs are added.

5.2.1 Adding temperature as input

With the goal of improving the performance of the forecast model, the temperature inside the houses is added as input. Table 5.4 shows the RMSE results with energy and temperature data as inputs for a neural network with one hidden layer. The lowest average RMSE is 86 Wh with a standard deviation of 8 for a neural network with N = 25 and D = 19. The smallest standard deviation is 4, with average RMSE of 94 Wh for a neural network with N = 10 and D = 21. This is the overall performance of the model for all households in village 2,

Table 5.4: Average RMSE (Wh) and its standard deviation as D increases for each neurons of neural network with 1 hidden layer for energy and temperature data as inputs

	Number of Neurons, N													
	5		10		15		20		25		30			
Delay, D	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD		
1	102	7	103	6	102	7	100	6	101	5	102	7		
3	98	7	99	9	98	9	98	7	97	6	100	9		
5	96	8	95	8	96	6	97	7	98	8	93	10		
7	97	11	96	7	96	8	95	7	95	7	94	8		
9	96	6	98	6	94	7	95	8	94	12	96	7		
11	95	9	95	8	93	8	91	9	92	6	91	8		
13	94	10	94	10	92	11	93	9	91	14	91	14		
15	92	10	96	11	92	10	95	8	90	9	91	11		
17	96	10	92	10	91	11	95	11	93	8	91	11		
19	96	10	95	9	93	9	95	10	86	8	91	10		
21	95	5	94	4	94	5	93	5	91	8	92	7		

considering the temperature inside all houses as input. Results show that the performance of the forecast model continues to improve as the number of input increased.

5.2.2 Adding weekdays as input

To improve further the performance of the forecast model, the number of weekdays is added as the input of the neural network. Table 5.5 shows the performance of the network for each neuron and delay. The lowest average RMSE is with 88 Wh with a standard deviation of 15 for a neural network with N = 30 and D = 15. The smallest standard deviation is 3, with an average RMSE of 94 Wh for a neural network with N = 5 and D = 21.

5.2.3 Adding demographic information as input

This section aims to evaluate the effect of adding the demographic information of each household or the consumer profiles as inputs to the neural network's performance in load forecasting.

An assumption was made that MLP-based load forecast model can perform better when demographic information of the households are considered as inputs.

The demographic information included in the model are:

- HHs total monthly income,
- number of occupants,

	Number of Neurons, N												
	5		10		15		20		25		30		
Delay, D	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	
1	102	6	105	4	102	8	104	5	102	6	102	5	
3	99	8	99	8	97	7	96	5	98	6	93	8	
5	96	6	96	10	94	7	96	8	96	11	93	9	
7	96	6	97	8	95	8	96	8	93	10	93	9	
9	96	6	96	6	95	6	93	8	92	8	91	10	
11	96	7	96	6	93	8	92	11	93	11	88	13	
13	96	10	95	10	94	11	93	9	91	14	91	12	
15	95	7	95	7	94	10	92	11	92	11	88	15	
17	95	10	93	10	93	8	94	8	92	8	91	8	
19	95	8	93	9	90	6	92	8	91	9	92	9	
21	94	5	94	6	93	7	90	5	91	5	89	10	

Table 5.5: Average RMSE (Wh) as D increases for each neurons of neural network with 1 hidden layer for energy, temperature and weekdays as inputs

...

. .

.. .

- number of appliances,
- frequency of using the appliances in a day (in terms of total hours),
- number of children with age less than five years old,
- number of children who are in school, and
- number of HHs member who is working.

The household's total monthly income is selected to be one of the input as income represents the economic status of the household. The number of occupants affects on how the household spend their energy. It is assumed that the larger the family, the more electricity is used. The total number of hours used of the appliance each day gives an idea to the model that this is how long they spend the energy given to them. The children with age less than five are expected to be at home at all time and having the kids at home, means the households need to have some entertainment for them and may use electricity. The number of working members tells the model how many people are inside the house during the day and how many people are away. All this demographic information is considered to be affecting directly or indirectly to energy consumption by using electricity through home appliances as the socio-economic factor in the load forecast model.

By integrating this information, the model has become adaptive to each household.

Table 5.6 shows the average RMSE and the standard deviation for the neural network with HL = 1 for each number of neurons N and delay D. The lowest average RMSE is 85 Wh with a standard deviation

Table 5.6: Average RMSE (Wh) as the delay D increases for each neuron per layer N of neural network with 1 hidden layer HL for energy, temperature and weekdays and the demographic information as inputs

	Number of Neurons, N													
	5		10		15		20		25		30			
Delay, D	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD		
1	102	5	102	7	100	7	102	8	101	8	100	7		
3	97	6	96	7	95	5	95	6	95	6	95	6		
5	95	8	95	8	95	6	93	8	93	9	93	11		
7	94	7	94	6	93	7	93	8	92	10	90	10		
9	95	7	95	6	91	9	92	10	91	7	91	11		
11	94	6	94	7	92	6	94	8	87	10	90	12		
13	93	9	93	12	92	10	89	11	92	14	88	12		
15	95	7	92	9	91	9	90	10	88	12	92	10		
17	94	12	92	9	88	10	91	11	92	9	86	8		
19	95	11	93	8	90	8	91	9	91	7	90	15		
21	96	5	92	7	93	7	88	7	85	12	87	14		

of 12 for a neural network with N = 25 and D = 21. The smallest standard deviation is 5, with average RMSE of 95 Wh for a neural network with N = 15 and D = 3.

The performance of the model is assessed not only in terms of RMSE but also with the time needed to train the network. Training time for a neural network with one hidden layer HL for the range of neurons per layer N from 5 to 30 and delay D from 1 to 21 is shown in Figure 5.5. Training time increases as the number of neurons per layer N and delays D increases.

5.2.4 How does the number of hidden layers HL affects the performance of the network in load forecasting?

In this section, the effect of the number of hidden layers to the performance of the neural network in load forecasting is evaluated. To see if the RMSE results vary significantly with the number of hidden layers HL and with all the inputs considered, hidden layer HL was increased from 1 to 3 and then 5. Table 5.7 shows the network's performance with three hidden layers in terms of average RMSE and standard deviation for each number of neurons N, and delay D. The lowest average RMSE is 83 Wh with a standard deviation of 8 for a neural network with N = 25 and D = 21. The smallest standard deviation is 5 for a network with N = 5 for both D = 3 and D = 21 with corresponding average RMSE of 96 Wh and 94 Wh, respectively.



Figure 5.5: Time needed to train the model for each delay, D with the corresponding neurons per layer N for neural network with HL = 1. X-axis is the range of delay D and Y-axis is the time in minutes m.

Table 5.7: Average RMSE (Wh) as D increases for each number of neurons with 3 hidden layer for energy, temperature, weekdays and the demographic information as inputs

	-	_				-								
	Number of Neurons, N													
	5		10		15		20		25		30			
Delay, D	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD		
1	103	8	101	6	101	8	99	8	101	6	99	7		
3	96	5	96	7	94	6	90	9	91	9	92	9		
5	93	11	94	10	95	7	93	11	85	13	89	9		
7	96	7	91	10	93	10	88	11	89	10	87	15		
9	96	6	96	7	93	10	85	18	83	20	88	13		
11	93	9	93	12	91	8	93	12	85	17	83	17		
13	94	7	92	11	89	16	89	13	87	22	85	16		
15	96	11	93	10	88	13	84	13	83	16	83	20		
17	94	9	92	9	86	11	85	13	86	7	89	10		
19	95	9	92	11	94	13	88	16	86	15	84	17		
21	94	5	91	8	88	9	93	8	83	8	83	20		



Figure 5.6: Time needed to train the model for each delays, D with the corresponding neurons, N. X-axis is the range of D and Y-axis is the time in minutes, m.

Figure 5.6 the time needed to train the network for each value of neuron for the range of delay D for a network with three hidden layers HL.

For a network with five hidden layers, the tabulated RMSE results are shown in Table 5.8. The lowest RMSE is 80 Wh with a standard deviation of 12 for a neural network with 20 neurons and 21 delays.

Figure 5.7 and Table 5.9 show the time needed to train the network with five hidden layers for each N and range of D. When HL was raised from 3 to 5, training time has increased accordingly.

To understand the effect of hidden layers HL to the performance of the network, the following simulations were conducted. For these simulations, we used HL value from one to five for a neural network with 20 neurons.

As the number of hidden layers of the neural network for the MLP-based forecast model was increased, training time increases proportionally with the number of hidden layers as well as with the number of neurons and the delays.

It was observed from the simulation results that the smallest average RMSE does not have the least standard deviation that makes the selection of the combination of the hidden layers HL, neurons per layer N and delay D even more difficult. The goal is to have the smallest RMSE as possible with the least standard deviation to have a minimum error in prediction using the MLP-based forecast model. From the gathered results of the experiments, there was no instance that the lowest av-

Table 5.8: Average RMSE (Wh) and standard deviation of neural network with 5 hidden layer with energy, temperature, weekdays and the demographic information as inputs for each neuron (N) and delay (D)

	Number of Neurons, N												
	5		10		15		20		25		30		
Delay, D	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	RMSE	SD	
1	103	8	101	7	101	7	100	7	99	9	97	7	
3	97	9	95	9	96	7	94	6	92	6	94	8	
5	91	10	95	9	90	9	93	10	89	14	89	10	
7	94	9	93	9	89	11	94	7	87	13	86	13	
9	97	7	95	9	93	9	92	7	90	11	88	15	
11	92	7	96	9	89	16	90	13	94	12	94	10	
13	95	7	91	9	86	16	85	14	88	13	92	12	
15	93	8	90	13	94	14	85	16	82	20	86	11	
17	95	9	91	10	87	11	89	9	91	11	92	14	
19	95	9	93	12	89	10	88	9	84	11	93	9	
21	94	6	97	4	94	9	80	12	81	21	95	7	



Figure 5.7: Time needed to train the model for each delays, D with the corresponding neurons, N with HL = 5. X-axis is the range of D and Y-axis is the time in minutes, *m*.

Table 5.9: Training time as the delay D increases for each neurons per layer N of neural network with 5 hidden layer for energy, temperature, weekdays and the demographic information as inputs

Training time in minutes, T_m													
	Number of neurons, N												
Delay, D	5	10	15	20	25	30							
1	0.04	0.09	0.29	0.85	1.78	4.42							
3	0.08	0.18	0.62	1.75	3.73	8.85							
5	0.12	0.27	0.96	2.70	5.82	13.50							
7	0.16	0.38	1.34	3.77	8.04	18.52							
9	0.20	0.50	1.73	4.89	10.34	23.83							
11	0.24	0.62	2.20	6.11	12.82	29.24							
13	0.28	0.76	2.67	7.37	15.36	35.03							
15	0.33	0.91	3.17	8.73	18.04	41.06							
17	0.37	1.07	3.70	10.20	20.86	47.56							
19	0.42	1.24	4.31	11.79	23.81	54.50							
21	0.48	1.43	4.90	13.58	27.00	62.18							

Table 5.10: Performance of the NN as D increases for each hidden layer for neural network with N = 20 and with energy, temperature, weekdays and the demographic information as inputs

Number of hidden layer(s), HL

	1		2		3		4		5	
Delay, D	RMSE	SD								
1	100	9	100	7	102	8	100	8	100	8
3	93	6	95	8	92	11	93	6	94	7
5	94	7	90	9	92	11	91	10	94	8
7	91	9	94	10	89	12	88	8	87	10
9	92	9	91	11	91	12	92	9	83	16
11	91	7	85	11	96	17	84	12	85	12
13	93	12	92	11	91	8	87	17	92	10
15	88	11	91	12	89	10	78	23	93	9
17	93	8	87	11	87	16	84	8	90	10
19	91	7	88	7	81	11	87	13	85	13
21	89	7	88	15	90	11	88	9	95	4


Figure 5.8: RMSE (Wh) values for neural network with 3 hidden layers HL and 20 neurons per layer N for each delay D with energy, temperature, weekdays and the demographic information as inputs.



Figure 5.9: Training time for neural network with 20 neurons per layer N and varying hidden layer HL for each delay D with energy, temperature, weekdays and the demographic information as inputs.



Figure 5.10: RMSE (Wh) values for neural network with 3 hidden layers HL and 20 neurons per layer N for each delay D with energy, temperature, weekdays and the demographic information as inputs.

erage RMSE has the least standard deviation. Results show that the lowest average RMSE sometimes has the largest standard deviation. This made the network undesirable as the forecast model needed to be precise as it can be in forecasting the next-day consumption of the households, hence lowest RMSE with small standard deviation is preferred. Note that the data used for training the forecast model is from September 2016 to December 2018 with 70% for training, 15% for testing and 15% for validation purposes. Finding the best combination of the three parameters which generates the smallest average RMSE with minimum standard deviation needs to be examined thoroughly to ensure the accuracy of the prediction.

5.3 DISCUSSION OF RESULTS

Recall that delay D represents the delay window as presented in Section **??**. From the input shaping and determining the best number of delay D to provide low average RMSE, results show that the delay D affects the performance of the model directly. However, this relationship between the delay D and the RMSE is nonlinear as shown in Figure 5.4 and Figure 5.8. And as for the number of neurons and hidden layers, the effect of these two hyper-parameters to the performance of the neural network in load forecasting is also nonlinear and can be observed in Figure 5.4 and Figure 5.8 as well as from Table 5.1 to Table 5.10.

From the lowest RMSE value in Table 5.1 which is 99 Wh with a standard deviation of 6 (for a neural network with 30 neurons N per layer and delay D of 11), with historical energy data as input, lowest RMSE value goes to 86 Wh with a standard deviation of 8 for a network with 25 neurons N per layer and delay D of 19 (Table 5.4) when households temperature was added as inputs. When calendar information such as weekdays was added as inputs, the lowest RMSE values is 88 Wh with a standard deviation of 13 for a network with 30 neurons per layer and delay D of 11 (Table 5.5). And when the demographic information of the household are added as inputs, the lowest RMSE value is 85 Wh with a standard deviation of 12 from a network with 25 neurons per layer and delay D of 21 (Table 5.6). All these networks have one hidden layer only. From these results, an RMSE of 85 Wh was achieved when the demographic information was added.

To continue to observe the effects of the number of hidden layers HL values is increased from 1 to 3 with the four inputs considered. Results are shown in Section 5.2.4. The performance is improved as the possible lowest RMSE is 83 Wh with a standard deviation of 8 from a network with 25 neurons per layer and delay D of 21. For a neural network with five hidden layers, the lowest RMSE value is 85 Wh with a standard deviation of 15 with 30 neurons per layer and delay D of 21 with energy data as inputs. However, adding the temperature, weekdays and demographic information of the households, the lowest RMSE is down to 80 Wh with a standard deviation of 12 from a neural network with ten neurons per layer and delay D of 21. It can be observed that the average RMSE is decreasing when the number of delay D is less than or equal to 9. This is true for all inputs as seen in Figure 5.4 and Figure 5.8 however RMSE values seem to be unstable when delay D is higher than 10. Several RMSE outliers occur for the range of delay D from 11 to 21. Furthermore, in this range, the standard deviation is bigger than with the lower values of delay D.

The lowest RMSE with the lowest standard deviation is desired as this model can forecast better with small variations in the performance of the network. However, considering the results, the lowest RMSE does not have the lowest standard deviation, which is the ideal preference, and the lowest standard deviation does not have the lowest RMSE. To aid in deciding the proper sizes of the network, the time parameter is considered. The training time is compared for each hidden layer with different neurons per layer N for the range of delay window D. Figures 5.5, 5.6, and 5.7, shows the training time needed

to train networks with 1, 3, and 5 hidden layers respectively for each value of neurons per layer and the range of delay D inputs.

It can be observed that the training time takes longer as the delay D increases as well as the values of neurons N per layer and hidden layers HL increases.

Therefore, to maintain a good performance in forecasting, a neural network with 20 neurons per layer N, three hidden layers HL and seven delays D is chosen to be the best network for a load forecast model to generate the lowest RMSE of 81 with a standard deviation of 9 and training time of 24 seconds. Simulation results show that the RMSE values are smaller with demographic information of the households added as input than without it. For the same number of hidden layer HL, neurons N per layer and window delay D, without demographic profile as input, the lowest RMSE result is 128 Wh with a standard deviation of \pm 14.1 and the % error is 20.11 %. When the demographic profile is added, the RMSE value decreased to 81 Wh with a standard deviation of \pm 9 and % error of 13.58 %. The % error improved by 32.47 %. The RMSE of 81 Wh is an acceptable value for this research that is below 15% of the allowable error. As stated earlier, the goal is to develop a forecast model that would generate an RMSE less than 120 Wh or 15 % of the actual energy usage.

Figure 5.11 shows the regression plot of the model that validates the network's performance in forecasting, which exhibits the output of the model for the given data sets for training, validation and test. For the proposed MLP-based model, the regression fit is fairly acceptable with R equal to 0.92, 0.91, and 0.90 for training, validation and test, respectively, with R equal to 0.92 for overall response.

Figure 5.12 shows the actual energy usage and the predicted values using the proposed model from the test data. The data used in this experiment is from Village 2, dated from September 2016 up to December 2018. The performance of the model in terms of RMSE is 93 Wh with a standard deviation of 9. The mean absolute per cent error (MAPE) is 6.2 %. Table 5.11 shows the MAPE and absolute per cent error between the actual energy usage and forecasted energy usage.

The main reason why household profiles are taken into considerations in developing the load forecast model is not only to improve the performance of the network but to make the model adaptive that would enable the model to do village-wide forecast using a single model rather than having one model for each household. This model is expected to work with similar household profiles of Village 2 in forecasting their next-day energy consumption.

Table 5.12 and Figure 5.13 show the performance of the model to forecast for each household load. The highest accuracy is 93.22% for



Figure 5.11: Regression plot of the neural network for load forecast model.



Figure 5.12: MLP-based forecast model performance. The actual energy usage is almost the same with the predicted values with the neural network's MAPE of 6.2 % and RMSE 93 Wh and standard deviation of 90.

No.	Actual	Predicted	Absolute percent	MAPE
			error %	
1	0.3695	0.3747	0.5255	6.2
2	0.8000	0.7449	5.5130	
3	0.5977	0.6702	7.2436	
4	0.6515	0.6264	2.5083	
5	0.6392	0.7330	9.3832	
6	0.6629	0.6560	0.6902	
7	0.7351	0.6446	9.0556	
8	0.5010	0.6003	9.9298	
9	0.7613	0.6771	8.4213	
10	0.5304	0.6131	8.2791	
11	0.6752	0.7380	6.2812	
12	0.8000	0.7630	3.6984	
13	0.7423	0.6950	4.7251	
14	0.7899	0.7436	4.6375	
15	0.8000	0.7480	5.1960	
16	0.8000	0.7198	8.0227	
17	0.7107	0.7575	4.6820	
18	0.8000	0.7771	2.2945	
19	0.7858	0.7228	6.3018	
20	0.2027	0.2854	8.2730	

Table 5.11: MAPE and absolute percentage error between the actual and forecasted energy usage of the households.



Table 5.12: Average forecasting accuracy of the MLP-based load forecast model for households in village 2



Figure 5.13: Prediction results using MLP-based forecast model

forecasting the energy usage of House 5 H5, and the lowest is 88.6% for House 6 H6. The overall accuracy is 91.3% for forecasting the energy usage of all households.

5.4 EVALUATION AND COMPARISON OF RESULTS

The performance of the proposed MLP-based load forecast model is compared to the existing models using the same data from village 2. Table 5.13 shows the comparison of the neural network architecture and the performance of the model in terms of RMSE.

The lowest RMSE is 77 Wh from the forecast model of Ryu et al. [RNK16] that uses DNN with four hidden layers and 50 neurons. However, this model has a standard deviation of 30, the highest standard deviation amongst the models. This model can forecast the next day energy consumption of the households more accurately than the others. However, the variations are more significant compare to other

Reference	Input	NN	Hidden	Neurons	RMSE	Std.	Training
	Variables		layer(s)			Deviation	Time (s)
Din and	Calendar data,	RNN	1	20	105	7	62
Marnerides [DM17]	weather conditions,						
and Din	historical energy data						
Ryu et al. [RNK16]	Calendar data,	DNN	4	50	77	30	1411
	weather conditions,						
	historical energy data						
	weather conditions						
This work	Calendar data,	MLP	3	20	81	9	24
	weather conditions,						
	historical energy data,						
	household's profile						

Table 5.13: Comparing the performance of the proposed MLP-based load forecast model to the existing models.

models with lower standard deviation. Moreover, this model takes 1411 s or 23.5 minutes to train the network to learn. A very long time compared to other MLP-based models presented. The work of Din and Marnerides [DM17] is similar to this work. The differences are on the neural network architecture, the former uses RNN with one hidden layer, and the latter uses MLP with three hidden layers, and the input variables considered. This work includes a household's profile aside from the three input variables mentioned in the work of Din and Marnerides. The proposed model offers RMSE of 81 Wh with a standard deviation of 9 compared to the model from the work of Din and Marnerides that generates an RMSE of 105 Wh with a standard deviation of 7. For the training time, the proposed MLP-based forecast model needs 24 s to train the neural network compared to the work of Din and Marnerides that needs 62 s. The training time takes longer as the number of input, and the NN hidden layers increases.

The performance of the network varies with respect to the number of the hidden layers HL, neurons per layer N, and the delay D. Since these parameters influence the performance of the network to forecast the energy usage significantly, the network is trained and evaluated with different data to avoid overfitting and to ensure that the model is working not only with the trained data but also with the new data.

It can be observed that this process can be beneficial in handling big data without performing a specific pre-processing to exclude outliers and still able to get good and acceptable results in forecasting. Since the approach does not require a special process on the data before using it as inputs of the network, makes this method easy to implement.

The MLP-based forecast model presented in this chapter estimated the next day energy consumption of the household within the accepted tolerance, which is 12 % of the current fixed energy allowance that is equivalent to 120 Wh. Since the model is using both historical energy

data and household profiles, the model generally estimates the energy consumption of each household. The model is to predict the next-day energy consumption of the households as the inputs were indexed for each profile. The model is then trained to learn the difference between households by the given index for each profile. Hence, when the data is similar to one of the profiles, the model is able to recognise the pattern and predict the next day energy consumption accordingly. Indexing the data through the profiles of the households enables the proposed model to predict the next-day energy consumption of each household without the need of creating another model for each household. For this study, the total number of households profiles is 4, as presented in Chapter 4 Section 4.8. The performance of the model in predicting the next-day usage of the households can be improved further when the number of profiles is increased. Practically, the proposed forecast model with household profile as one of the inputs can predict each household's energy consumption without having special features. The proposed model also shows better performance in terms of RMSE when compared with the other two existing works, as presented in Table 5.13 that use neural networks.

5.5 EXPERIMENTAL RESULTS USING OTHER FORECASTING TECH-NIQUES

To assess the overall performance of the MLP-based forecast model, the results were compared with three other existing forecasting techniques such as radial basis function neural network (RBFN), Gaussian progress regression (GPR) and autoregressive integrating moving average (ARIMA).

5.5.1 Gaussian Process Regression

For comparison purposes, a Gaussian process regression (GPR) model is also developed to forecast a next-day energy usage of the households in an off-grid community. GPR models are non-parametric probabilistic models.

Matlab GPR tool is used to develop the model for forecasting the next-day energy consumption of the households. The algorithm used in this tool includes estimating the important parameters from the given data such as the covariance function $k(x_i, j_i \div \theta)$, noise covariance, σ^2 , vector coefficient of fixed basis function, β . The function 'fitrgp' makes use of the values of the aforementioned parameters to determine the kernel parameters. The vector value of the kernel



Figure 5.14: Prediction performance of GPR model

parameters argument consists of all the initial values of the data's standard deviation σ_f and the characteristic length scale σ_l .

The performance of the model in terms of RMSE is 97 Wh with a standard deviation of 4.

5.5.2 Radial basis function neural (RBFN) networks

Another model is developed for comparison purposes; in this section, the RBFN model is presented. RBFN is one of the simplest forms of neural networks. It consists of 3 layers only: the input layer, a hidden layer, and the output layer. The network has one hidden layer that connects all the neurons from the input layer and the output based on the specific weights specified by the network. Building the RBFN network follows a specific procedure like the feedforward neural networks. The four essential parameters in designing or building the RBFN network considered in this chapter are the initial centroid, the number of neurons, β , and delay.

Training the RBFN network is a bit tricky since in each simulation, the initial values for centroids, neurons, and the beta β need to be initialised. Determining the right combinations of these three hyper-parameters are critical as they dictate the performance of the network in forecasting. Table 5.14 shows the performance of RBFN in terms of RMSE. The lowest RMSE recorded in this simulation is 0.169 kWh with a beta β of 0.001, neurons per layer N of 10, centroid C of 110, and delay D of 3.

$(N = 10, \beta = 0.001)$						
Initial	Delay,	RMSE (kWH)	Delay,	RMSE (kWh)		
Centroids, C	d	(Lowest)	d	(Highest)		
20	1	0.224	7	0.275		
30	1	0.235	13	0.271		
40	1	0.212	7	0.271		
50	1	0.181	9	0.270		
60	1	0.181	15	0.261		
70	2	0.171	8	0.253		
80	1	0.172	14	0.249		
90	3	0.171	13	0.233		
100	2	0.171	12	0.231		
110	3	0.169	14	0.219		
120	3	0.172	14	0.217		
130	3	0.171	12	0.214		
150	3	0.171	13	0.207		
200	4	0.171	14	0.196		

Table 5.14: RBFN RMSE Results

The performance of the RBFN network is presented in Table 5.14. It can be observed that as the values of beta β are varied and keeping the values for centroids and neurons, the RMSE results either increase or decrease. The beta β controls the range of the variation of the radial function and thus affects the results in a symmetrical manner. The performance of the network depends on the values of the beta β . When the number of neurons is varied while keeping the values of beta β , and centroids C, constant, the performance of the network also varies. Depending on the number of the delay window D, the RMSE is either increased or decreased or both. When the number of centroids C is varied while keeping the values for neurons per layer N, beta β , and the delay D constant, the performance of the network is either increase and decreased. This result is the same when the number of neurons per layer and the beta β are varied. Since the performance of the network does not convey a specific pattern when the parameters are varied, selecting the right combination takes time until all the combinations were considered and simulated. For the RBFN model, the lowest RMSE is selected and compared with the performance of the proposed MLP-based forecast model (Table 5.15).



Figure 5.15: Daily predicted values of energy consumption (y-axis)

Techniques	Network size,	RMSE (Wh)	SD
	delays, and model		
MLP	N=20, HL=3, D =7	81	9
RBFN	C=10, D=3, N=10, β=0.001	169	20
GPR	D=7	103	15
ARIMA	p=1, d=0, q=0	120	20

Table 5.15: Comparison of performance of the forecast model

5.5.3 Autoregressive integrating moving average (ARIMA)

Another forecasting technique that is popular in load forecasting is ARIMA. The output is determined by averaging the entire data given to the model. Figure 5.15 shows the prediction performance of the ARIMA-based load forecasting with RMSE of 119.9 Wh and a standard deviation of 20.

5.6 COMPARISON OF RESULTS

The performance of the proposed model is comparable with the existing model using the available data. From the results shown in Table 5.15, MLP-based model is outperforming the three other techniques in terms of RMSE.

When using MLP-based model forecasting, the model is expected to have an error around 72 Wh to 90 Wh from the actual usage as represented by the RMSE results of 81 Wh with nine a standard deviation. The performance of the other three techniques are comparable however to have an optimal energy allocation, the performance of the forecast model is important thus the lowest RMSE model is preferred which in this case, the MLP-based forecast model.

5.7 CHAPTER SUMMARY

In this chapter, the development of the MLP-based forecast model is presented. The procedure on selecting the value for the three parameters of the network namely the hidden layer HL, the neurons N) and the delay D – the number of the historical data points taken as inputs of the neural network is discussed. The right combination of these three parameters affects the performance of the model directly to forecast the next day energy usage of the households. Thus, in this chapter, a step-by-step process of finding the right combination of the hidden layer HL, number of neurons per layer N and delay D of the neural networks is presented. The range considered in the selection is also explained in details. Careful selection of the values for the three parameters aid the model to have a better prediction as these parameters directly affect the performance of the forecasting model. The model was trained and validated using the data gathered from using PMOG systems, as presented in Chapter 4.

The performance of the MLP-based forecast model was assessed in terms of RMSE and MAPE. Results show that as the input of the network is changed, the RMSE results also change accordingly. The lowest possible RMSE for MLP-based forecast model with historical data as input is 99 Wh, when temperature and weekdays were added as inputs, the lowest values of RMSE are 86 Wh and 88 Wh, respectively. This value goes down to 85 Wh when demographic information was added as input. This shows that the performance of MLP-based forecast model is improved when the demographic information of the households was included as inputs. The final MLP-based forecast model used in this research has 20 neurons, three hidden layers, and seven delays that has 81 Wh RMSE with a standard deviation of 9. This means that the highest possible error would be 89 Wh and the lowest possible error is 72 Wh. This model gives an overall accuracy of 91.3 % with 93.2 % accuracy for House 5 as the highest.

It was hypothesised that the energy consumption of the households could be forecast with reasonable accuracy. From the given results that fall within the threshold set as the allowable error, the hypothesis was answered yes. From the experiment results, the next-day energy consumption of the households can be forecasted within the threshold using MLP-based forecast model employing households profiles as input aside from the historical data, temperature, and weekdays. The RMSE of the proposed model was improved from 128 Wh without household profile as input to 81 Wh with a household profile. The % error is improved by 32.5 %.

The performance of the proposed model is compared with three other existing forecasting techniques such as RBFN, GPR and ARIMA. Compared to the other three models, MLP-based forecast model outperforms RBFN, GPR and ARIMA in terms of RMSE as shown in Table 5.15. The proposed MLP-based forecast model was also compared to two existing works that use the neural network in forecasting. Table 5.13 shows that the proposed MLP-based forecast model outperforms the two other forecast models with the difference in the input sets. The performance of the proposed MLP-based forecast model is better in terms of RMSE and training time.

The forecast model's performance has improved as the consumer profiles were added as inputs together with the historical data. The model's RMSE is less 15%, which successfully achieved the goal. Therefore, the next day energy consumption of the household is forecasted with reasonable accuracy using MLP-based forecast model.

OPTIMAL DAILY ENERGY ALLOCATION BASED ON MLP-BASED FORECAST MODEL

This chapter presents the mathematical framework of determining the optimal energy allocation of each household in a community based on their forecasted energy demand. The purpose of determining the optimal energy allocation is to ensure that the utilisation of the available energy is maximised while minimising the energy wastage.

An optimisation is used when a certain variable needs to be minimised or maximised based on known constraints. These constraints are the guiding parameters of the optimisation process. Using Karush-Kuhn-Tucker (KKT) conditions, the function can be minimised when the conditions are met. KKT is one of the optimisation techniques that allow a function to be evaluated with constraints in the forms of equalities and inequalities as shown in the following equations as the general form of optimisation problem:

$$\min_{\mathsf{E}\in\mathbb{R}^n} \quad \mathsf{f}(\mathsf{E}) \tag{6.1}$$

subject to

$$h_i(E) \quad \leqslant \quad 0, i = 1, 2, ..., m \tag{6.2}$$

$$l_j(E) = 0, j = 1, 2, ..., r$$
 (6.3)

where

f(E) is the objective function to be minimised,

 $h_i(E)$ and $l_i(E)$ are the constraints.

The KKT conditions are expressed in the following equations:

$$0 \in \partial f(E) + \sum_{i=1}^{m} \mu_i \partial h_i(E) + \sum_{j=1}^{r} \upsilon_j \partial l_j(E)$$
 (6.4)

$$\begin{array}{rcl} \mu_{i}h_{i}(E) &=& 0 \ \forall \ i & (6.5) \\ h_{i}(E) &\leqslant& 0, \ \ l_{j}(E) &=& 0 \ \forall \ i, \ j & (6.6) \end{array}$$

(6.6)

$$\mu_i \geqslant 0 \forall i$$
 (6.7)

where equation 6.4 is the called stationary, equation 6.5 is the complementary slackness, equation 6.6 is the primal feasibility, and equation 6.7 is the dual feasibility. These are the KKT conditions that are vital in solving the optimisation problem. In order to be able to use

the KKT approach in solving the optimisation problems, the above conditions must be met. Complementary slackness is the 2nd condition that deals with both primal and dual feasibilities. Complementary slackness states that if the dual variable is greater than zero, then the primal constraint is zero. It also states that if the primal constraint is greater than zero, then the dual variable is zero.

In this research, the aggregated supply from the solar-powered offgrid power systems is considered as constant since the system is using batteries to ensure uninterrupted power source to the households and for simplicity of the calculations. The summation of the proposed optimal energy allocation must be equal to the generated energy by the solar power systems. This is one of the constraints of the optimisation as expressed in equation 6.9. Another constraint considered in the formulation of the optimisation problem is that the proposed energy allocation should be higher or equal to the calculated energy threshold of the households as presented in equation 6.10. The energy threshold ensures that the optimal energy allocation is sufficient for the electricity needs of the households in a day. The constraints for this optimisation problems meet the KKT conditions. Thus KKT approach is used in solving the optimisation problem.

This chapter intends to provide answers to the Research Questions (RQs) 2 and 3:

RQ2: Can knowledge of consumer profiles aid in optimal and adaptive energy allocation?

RQ3: Can the energy allocation be optimised to improve the energy efficiency under the limited generation capacity?

Note that for RQ 2, as presented in Chapter 5, knowledge of consumer profiles are used in forecasting the household's energy usage, and the predicted energy usage is used in determining the optimal daily energy allocation of each household. Note also that there is only one load forecast model used in predicting the households next-day energy consumption. The model is adaptive to each household since it uses the historical energy data and the demographic profiles. The optimisation problem is formulated from the issues raised due to the limited generation capacity and the power outage experienced by households as presented in Chapter 4.

6.1 **PROBLEM DESCRIPTION**

In order to address the above research questions, an optimisation problem is defined. As discussed in chapter 1, two main issues arise within the current existing energy allocation, whereby each household has equal daily energy quota. These issues are re-stated below:

1.) Some households use up the allocated daily quota and experience power outage but may require more energy than what is currently provided to them.

2.) Some households do not use all the allocated energy daily and pay the same fixed monthly tariffs as the households that use all their allocated energy.

The excess energy quotas of the households in scenario 2 can be allocated to the households in scenario 1 and minimise the power outage duration if not eliminated, without increasing the generation capacity of the microgrid.

To address the above issues, a new approach to allocating the daily energy quotas of the household is proposed. This approach involves forecasting the ideal energy usage of the households daily from the historical data first before allocating their optimal energy quotas. The method utilises the forecasted household energy demand from Chapter 4. The approach also considers the limited generation capacity of the microgrid and the basic electricity needs of the households. The proposed approach of allocating the daily energy quota of each household will enable new payment scheme based on their actual usage.

Given the limited generation capacity of the power system, the goal is to minimise the difference between the forecasted energy and the allocated energy. This is to make sure that the allocated energy is as close as possible to the ideal energy demand of the households.

Thus, the optimisation problem is defined to minimise the cost function J as given in equation 6.8 subject to two constraints which are the generation capacity and the minimum threshold energy level for each household.

Note that the ideal demand is the forecasted energy consumption based on the household's historical usage and the households individual profile.

The nonlinear optimisation problem is expressed in the differentiable standard form in equation 6.8,

min J =
$$\sum_{i=1}^{n} (E_i^a - E_i)^2$$
 (6.8)

subject to

$$\sum_{i=1}^{n} E_i^a = E_G \tag{6.9}$$

$$E_{i}^{a} \geq E_{\min}, \forall i \in \{1, 2, ..., n\}$$
(6.10)

where

 E_i^a is the calculated optimal energy allocation for each household, which is the optimisation variable,

 E_i is the ideal energy usage as predicted by the MLP-based forecast model,

 E_G is the generated supply by the off-grid power system,

n is the total number of households, and

 E_{min} is the daily energy threshold for each household.

Equation (6.8) is the objective function, while equations (6.9) and (6.10) are the equality and inequality constraints (Note that other power losses are neglected). The presence of the inequality constraint allows the optimisation problem to be solved using Karush-Kuhn-Tucker (KKT) conditions.

6.1.1 Solving the optimisation problem using KKT approach

In this section, the mathematical analysis of the optimal energy allocation is presented. The methodology for obtaining the optimal energy allocation for each household is derived by solving the optimisation problem presented in the previous section considering the given constraints. The optimal solution is derived by solving the equations analytically using the first-order optimality conditions. Using KKT methods, the equality and inequality equations are expressed in the standard forms g(x) = 0 and $h(x) \leq 0$, respectively. So, equations 6.9 and 6.10 can be written as

$$\sum_{i=1}^{n} E_{i}^{a} - E_{G} = 0$$
 (6.11)

$$E_{\min} - E_i^a \leq 0, \forall i \in \{1, 2, ..., n\}$$
(6.12)

Given the minimisation problem in equation (6.8), the Langrangian function, L is defined by combining the three equations, (6.8), (6.11), and (6.12) in the form of

$$L = f(x) + \sum_{i} \lambda_i g_i(x) + \sum_{j} \mu_j h_j(x)$$
(6.13)

which results in:

$$L = \sum_{i=1}^{n} (E_{i}^{a} - E_{1})^{2} + \lambda (\sum_{i=1}^{n} E_{i}^{a} - E_{G}) + \sum_{i=1}^{n} \mu_{i} (E_{min} - E_{i}^{a})$$
(6.14)

where

$$\mu_{i} \ge 0, \ \forall \ i \ \in \ \{1, 2, ..., n\}$$
(6.15)

The optimality conditions are obtained by taking the partial derivative of the Lagrangian (equation 6.14) with respect to E_i^a as follows:

$$\frac{\partial L}{\partial E_{i}^{a}} = 0$$

$$2(E_{i}^{a} - E_{i}) + \lambda - \mu_{i} = 0$$
(6.16)

and the partial derivative of the Langrangian with respect to λ ,

$$\frac{\partial L}{\partial \lambda} = 0$$

$$\sum_{i=1}^{n} E_{i}^{\alpha} - E_{G} = 0$$
(6.17)

The complementary slackness is given as follows:

$$\mu_{i}(E_{\min} - E_{i}^{a}) = 0, \ \forall \ i \in \{1, 2, ..., n\}$$
(6.18)

which implies that either

$$E_{\min} < E_i^a \text{ and } \mu_i = 0 \tag{6.19}$$

or

$$E_{\min} = E_i^a \text{ and } \mu_i > 0 \tag{6.20}$$

In order to solve the systems of equations given in equation 6.16 and 6.17, let

$$s := \arg \min_i E_i$$

That is, s is the index used to identify the households with the smallest amount of predicted ideal demand and (E_{min}) is the minimum energy to be allocated to the households without compromising their basic

electricity needs.

From equation (6.14), for i = s, the following is obtained:

(6.22)

The expression for the optimal energy allocation for each household is obtained as follows:

$$\forall j = 1, 2, ..., n; j \neq s,$$
$$2(E_j^{\alpha} - E_j) + \lambda - \mu_j = 0 \tag{6}$$

Substituting equation 6.21 into 6.22 gives:

$$2(\mathsf{E}_j^{\mathfrak{a}}-\mathsf{E}_j)+\mu_s+2(\mathsf{E}_s-\mathsf{E}_s^{\mathfrak{a}})-\mu_j=0$$

so that,

$$E_{j}^{a} = \frac{\mu_{j} - \mu_{s}}{2} - (E_{s} - E_{s}^{a}) + E_{j}$$
(6.23)

On the other hand, from equation 6.17, the aggregate optimal energy allocation, $\sum_{i=1}^{n} E_i^a$ can now be expressed in terms of s and j as follows:

$$\sum_{i=1}^{n} E_{i}^{a} = E_{s}^{a} + \sum_{j=1, j \neq s}^{n} E_{j}^{a} = E_{G}$$
(6.24)

Then E_s^a by substituting equation 6.23 into 6.24, the following is obtained:

$$E_{s}^{a} + \sum_{j=1, j \neq s}^{n} \left(\frac{\mu_{j} - \mu_{s}}{2} - (E_{s} - E_{s}^{a}) + E_{j} \right) = E_{G} \quad (6.25)$$

$$E_{s}^{a} + \sum_{j=1, j \neq s}^{n} \frac{\mu_{j} - \mu_{s}}{2} - \sum_{j=1, j \neq s}^{n} (E_{s} - E_{s}^{a}) + \sum_{j=1, j \neq s}^{n} E_{j} = E_{G} \quad (6.26)$$

$$E_{s}^{a} + \sum_{j=1, j \neq s}^{n} \frac{\mu_{j} - \mu_{s}}{2} - (n-1)(E_{s} - E_{s}^{a}) + \sum_{j=1, j \neq s}^{n} E_{j} = E_{G} \quad (6.27)$$

So,

$$E_{s}^{a} = \frac{E_{G} - \sum_{j=1, j \neq s}^{n} E_{j} - \sum_{j=1, j \neq s}^{n} \frac{\mu_{j} - \mu_{s}}{2} + (n-1)E_{s}}{n}$$
(6.28)

or expressed as

$$E_{s}^{a} = \frac{1}{n} \left(E_{G} - \sum_{j=1, j \neq s}^{n} E_{j} - \sum_{j=1, j \neq s}^{n} \frac{\mu_{j} - \mu_{s}}{2} + (n-1)E_{s} \right)$$
(6.29)

From equation 6.18, the complementary slackness,

$$\mu_i(\mathsf{E}_{\min} - \mathsf{E}^a_i) = \mathbf{0}, \ \forall i \tag{6.30}$$

for i = s,

$$\mu_s(\mathsf{E}_{\min} - \mathsf{E}_s^{\mathfrak{a}}) = 0 \tag{6.31}$$

In order that the minimum energy, E_{min} is allocated to the household with the smallest ideal energy demand, let

$$\mathsf{E}_{\min} = \mathsf{E}^a_{\mathsf{i}} \tag{6.32}$$

This implies from equation 6.19 and 6.20, the complimentary slackness, that

$$\mu_s \neq 0 \tag{6.33}$$

This also implies that the minimum energy allocation to any household does not fall below E_s^a , i.e., what is allocated to the household with the smallest ideal energy demand E_s amongst all households. Moreover, since the remaining households must be allocated with energy higher than the household with the smallest ideal demand, it follows that

$$E_{j}^{a} > E_{s}^{a}, \forall j \in \{1, 2, ..., n\}, j \neq s$$
 (6.34)

resulting in

$$E_{j}^{a} > E_{min}, \forall j \in \{1, 2, ..., n\}, j \neq s$$
 (6.35)

since $E_s^a = E_{min}$. Therefore, $\mu_j = 0$, $\forall j \in \{1, 2, ..., n\}, j \neq s$ as given in equation 6.19

Thus, to find the optimal energy allocations for all other households $\forall j \in \{1, 2, ..., n\}, j \neq s$, let $\mu_j = 0$, and from equation 6.29

$$E_{s}^{a} = \frac{1}{n} \left(E_{G} - \sum_{j=1, j \neq s}^{n} E_{j} - \sum_{j=1, j \neq s}^{n} \frac{\mu_{j} - \mu_{s}}{2} + (n-1)E_{s} \right)$$

becomes

$$E_s^a = \frac{1}{n} \left(E_G - \sum_{j=1, j \neq s}^n E_j + \frac{(n-1)\mu_s}{2} + (n-1)E_s \right)$$
(6.36)

Since $E_s^a = E_{min}$, equation 6.36 can be rewritten as:

$$E_{\min} = \frac{1}{n} \left(E_G - \sum_{j=1, j \neq s}^n E_j + \frac{(n-1)\mu_s}{2} + (n-1)E_s \right)$$

so that:

$$\mu_{s} = \frac{2}{n-1} \left(n E_{\min} - E_{G} + \sum_{j=1, j \neq s}^{n} E_{j} - (n-1)E_{s} \right)$$
(6.37)

From equation 6.23,

$$E_{j}^{a} = \frac{\mu_{j} - \mu_{s}}{2} - (E_{s} - E_{s}^{a}) + E_{j}$$

thus for $\mu_j = 0, \ \forall j \in \{1, 2, ..., n\}, j \neq s$, the following is obtained: we have,

$$E_{j}^{a} = \frac{-\mu_{s}}{2} + (E_{s}^{a} - E_{s}) + E_{j}$$
(6.38)

6.1.2 Steps in solving the optimal energy allocation

From the derived equations above, to obtain the optimal energy allocations, we need to:

- 1. Choose a minimum energy allocation E_{min}
- 2. Let $E_s^a = E_{min}$
- 3. Compute μ_s from equation (6.37)
- Determine optimal allocation for each household using equation (6.38)

6.1.3 Verification of the optimal solution

Recall from equation 6.33 that $\mu_s \neq 0$. However, to satisfy the complementary slackness condition given in equations 6.18 and 6.20, μ_s must be shown to be greater than zero, i.e., ($\mu_s > 0$) according to equation 6.15. Note that the complementary slackness captures 2 things in the solving the optimal value of the optimisation process. One, it states that when E_{min} is less than E_{i}^{a} , μ must be equal to zero. Two, when E_{min} is equal to E_{i}^{a} , μ must be greater than zero. So, from equation (6.37), it follows that:

$$\frac{2}{n-1}(nE_{\min} - E_G + \sum_{j=1, j \neq s}^{n} E_j - (n-1)E_s) > 0$$
$$nE_{\min} - E_G + \sum_{j=1, j \neq s}^{n} E_j - (n-1)E_s > 0$$
(6.39)

Solving for E_{min} yields:

$$nE_{\min} > E_{G} + (n-1)E_{s} - \sum_{j=1, j \neq s}^{n} E_{j}$$

$$E_{\min} > \frac{1}{n} \left(E_{G} + (n-1)E_{s} - \sum_{j=1, j \neq s}^{n} E_{j} \right)$$
(6.40)

Moreover, from equation (6.12), $E_{\min} \leq E_i^a$, $\forall i \in \{1, 2, ..., n\}$. For i = s, $E_s^a = E_{\min}$,

$$E_{\min} < E_{i}^{a}, \forall i \in \{1, \dots, n\}, i \neq s$$

$$(6.41)$$

Therefore, by substituting equation 6.38 into 6.41, the following is obtained:

$$E_{j} + (E_{s}^{a} - E_{s}) - \frac{\mu_{s}}{2} > E_{min}$$

$$E_{j} - E_{s} - \frac{\mu_{s}}{2} + E_{s}^{a} > E_{min}$$
(6.42)

but $E_s^a = E_{min}$, therefore,

$$E_{j} - E_{s} - \frac{\mu_{s}}{2} + E_{min} > E_{min}$$

$$E_{j} - E_{s} - \frac{\mu_{s}}{2} > 0 \qquad (6.43)$$

Thus,

$$\mu_{s} < 2(E_{j} - E_{s}), \forall_{j} \in \{1, 2, \dots, n\}, j \neq s$$
(6.44)

Substituting equation (6.37) into equation (6.44), to obtain an upper bound for E_{min} , we have

$$\left(nE_{\min} - E_G + \sum_{j=1, j \neq s}^{n} E_j - (n-1)E_s \right) \frac{2}{n-1} < 2(E_i - E_s),$$

$$\forall i \in \{1, 2, \dots, n\}$$

$$nE_{\min} - E_G + \sum_{j=1, j \neq s}^{n} E_j - (n-1)E_s < (n-1)(E_i - E_s)$$

$$nE_{\min} - E_G + \sum_{j=1, j \neq s}^{n} E_j < (n-1)E_i$$
(6.45)

thus, for any j

$$E_{\min} < \frac{(n-1)E_{i} + E_{G} - \sum_{j=1, j \neq s}^{n} E_{j}}{n}, \forall i \in \{1, 2, \dots, n\}, i \neq s \quad (6.46)$$

Let

$$l = arg \min_{i \in \{1,2,\ldots,n\}, i \neq s} E_i,$$

(Note: l is defined as the index of the household with the second smallest ideal energy demand E_i , after household s.)

then, it suffices to write the set of (n - 1) equation in 6.46 as:

$$E_{\min} < \frac{1}{n} \left((n-1)E_1 + E_G - \sum_{j=1, j \neq s}^n E_j \right)$$

Also, recall from equation 6.40 that

$$E_{\min} > \frac{1}{n} \left(E_G + (n-1)E_s - \sum_{i=1, i \neq s}^n E_i \right)$$

Therefore, the minimum energy allocation to the household should be within the range as expressed by the inequality in 6.47 below:

$$E_{G} + (n-1)E_{s} - \sum_{j=1, j \neq s}^{n} E_{j} < nE_{min} < (n-1)E_{l} + E_{G} - \sum_{j=1, j \neq s}^{n} E_{j}$$
(6.47)

6.1.4 Validating the optimisation results using the forecasted results

In the previous section, the mathematical framework of the optimisation of energy allocation for each household is presented. Now, the equations derived from the objective function of the optimisation will be used to determine the optimal energy allocation of each household in the generation-constrained microgrids. Firstly, an example of how the equation works is presented.

Consider the following energy consumption values predicted by the MLP-based forecast model for the four selected households connected to the microgrid. These values represent the ideal utilisation of the four houses.

 $E_1 = 0.7845 \text{ kWh}$ $E_2 = 0.5075 \text{ kWh}$ $E_3 = 0.6394 \text{ kWh}$ $E_4 = 0.2914 \text{ kWh}$

For these 4 houses, the total energy allocation is 3.2 kWh and the smallest forecasted energy usage is $E_4 = 0.2914$ kWh, i.e., s = 4, with the second lowest forecasted energy usage being $E_2 = 0.5075$ kWh i.e., l = 2, and the summation of the forecasted energy usage excluding E_4 is

$$\sum_{j=1,j\neq s}^{n}E_{j}=1.9314\text{ kWh}$$

From the steps presented in section 6.1.1, in which the first step is to choose the minimum energy allocation E_{min} . However from equation 6.47, the value of E_s , which is the smallest amount of the predicted ideal demand, must be known before the range for the E_{min} is determined. So, from equation 6.47, the range of E_{min} is determined as follows:

$$\begin{split} E_{G} + (n-1)E_{s} &- \sum_{j=1, j \neq s}^{n} E_{j} < nE_{min} < (n-1)E_{1} + E_{G} - \sum_{j=1, j \neq s}^{n} E_{j} \\ &\left(3.2 + 3(0.2914) - 1.9314\right) \frac{1}{4} < E_{min} < \frac{1}{4} \left(3(0.5075) + 3.2 - 1.9314\right) \\ & 0.5402 < E_{min} < 0.7023 \text{ kWh} \end{split}$$

The value for E_{min} is chosen within the calculated range as shown above. This means that within this calculated range using equation 6.47, the E_{min} can be chosen arbitrarily and it will lead to the optimal solution. For example, E_{min} is arbitrarily chosen as 0.55 kWh, which is closed to the lower limit as calculated above (equation 6.48). By virtue of the steps described in Section 6.1.2, E_{min} is set equal to E_s^a which means,

$$E_4^a = 0.55 \text{ kWh}.$$

Then, the term μ_s is calculated using equation 6.37:

$$\mu_{s} = \frac{2}{n-1} \left(nE_{\min} - E_{G} + \sum_{j=1, j \neq s}^{n} E_{j} - (n-1)E_{s} \right)$$

$$\mu_{4} = \frac{2}{4-1} \left(4(0.55) - 3.2 + 1.9314 - 3(0.2914) \right)$$

$$\mu_{4} = 0.0381$$
(6.49)

which satisfy the condition stated in equation 6.44 that μ_s must be less than twice the difference between E_j and E_s as also shown below:

$$\mu_s < 2(\mathsf{E}_j - \mathsf{E}_s), \forall_j \in \{1, 2, \dots, n\}, j \neq s$$

For E_1 ,

$$\label{eq:masses} \begin{split} \mu_s &< 2(\mathsf{E}_1-\mathsf{E}_s) \\ \mu_s &< 2(0.7845-0.2914) \\ 0.0381 &< 0.9862, \text{check} \end{split}$$

For E_2 ,

$$\label{eq:masses} \begin{split} \mu_s &< 2(E_2-E_s) \\ \mu_s &< 2(0.5075-0.2914) \\ 0.0381 &< 0.4322, check \end{split}$$

For E₃,

$$\label{eq:masses} \begin{split} \mu_s &< 2(E_3-E_s) \\ \mu_s &< 2(0.6394-0.2914) \\ 0.0381 &< 0.6960, check \end{split}$$

Then the optimal energy allocation for the remaining households is calculated using equation 6.38

$$E_{j}^{a} = \frac{-\mu_{s}}{2} + (E_{s}^{a} - E_{s}) + E_{j}$$

$$E_{1}^{a} = \frac{-0.0381}{2} + (0.55 - 0.2914) + 0.7845$$

$$E_{1}^{a} = 1.0240 \text{ kWh}$$
(6.50)

$$E_{2}^{a} = \frac{-0.0381}{2} + (0.55 - 0.2914) + 0.5075$$

$$E_{2}^{a} = 0.7470 \text{ kWh}$$
(6.51)

$$E_3^a = \frac{-0.0381}{2} + (0.55 - 0.2914) + 0.6394$$

$$E_3^a = 0.8790 \text{ kWh}$$
(6.52)

To check if the results satisfy the constraints of the optimisation problem stated in section 6.1 given by the equations 6.9 and 6.10, the summation of optimal energy allocations is calculated and each allocation is compared to the minimum energy threshold for the households. The summation $(\sum_{i=1}^{n} E_i^a = E_G)$ of all the optimal energy allocation is 3.2 kWh which satisfies the first constraint. The optimal energy allocation for each household E_i^a is higher than the minimum energy threshold of E_{min} , which satisfies the second constraint. For this first example of the optimal energy allocation, the objective function J evaluates 0.2390 kWh using the optimal energy allocation compared to 0.3703 kWh for the equal energy allocation which corresponds to a reduction of 54.9% in terms of wastage of energy.

6.1.5 Energy threshold and its effect on the optimal energy allocation

Suppose now, the value of E_{min} is chosen arbitrarily as 0.69, which closer to the upper limit. From equation 6.37, μ_s is

$$\mu_{s} = \frac{2}{n-1} \left(nE_{\min} - E_{G} + \sum_{j=1, j \neq s}^{n} E_{j} - (n-1)E_{s} \right)$$

$$\mu_{4} = \frac{2}{4-1} \left(4(0.69) - 3.2 + 1.9314 - 3(0.2914) \right)$$

$$\mu_{4} = 0.4115$$
(6.53)

So, the optimal energy allocation for the remaining households is calculated using equation 6.38

$$E_{j}^{a} = \frac{-\mu_{s}}{2} + (E_{s}^{a} - E_{s}) + E_{j}$$

$$E_{1}^{a} = \frac{-0.4115}{2} + (0.69 - 0.2914) + 0.7845$$

$$E_{1}^{a} = 0.9773 \text{ kWh}$$
(6.54)

$$E_2^a = \frac{-0.4115}{2} + (0.69 - 0.2914) + 0.5075$$

$$E_2^a = 0.7003 \text{ kWh}$$
(6.55)

$$E_3^a = \frac{-0.4115}{2} + (0.69 - 0.2914) + 0.6394$$

$$E_3^a = 0.8322 \text{ kWh}$$
(6.56)

For this example, the summation of the optimal energy allocation is 3.1998 kWh which is approximately 3.2 kWh which satisfies the first constraint like the example 1. Each allocation for each household is greater than the minimum threshold (E_{min}) that is chosen from the calculated range given by the equation 6.47. The objective function J is equal to 0.2704 and when compared with equal allocation, a 36.9% reduction of energy waste is achieved with the proposed optimal energy allocation.

For the next example, E_{min} is chosen from the middle of the range. This time, E_{min} is 0.62 kWh. The examples shown above reflects how the minimum threshold for the energy consumption of households affects the total energy wastage. When the chosen value for energy threshold E_{min} is closer to the lower limit of equation 6.47, the reduction of the energy waste in terms of unused energy or deficit is higher compared when E_{min} is closer to the upper limit. For this reason, the value for the energy threshold E_{min} closer to the lower limit is preferred when choosing the energy threshold in order to maximise the energy usage of the available energy.

Intuitively, one may use ratio and proportion to allocate the available energy. The households can be given energy proportional to their ideal utilisation as predicted by MLP-based load forecast model. For example, given the predicted values in Section 6.1.4 and re-stated here, then according to ratio and proportion, the energy allocation for each household can be calculated as follows:

$$\mathsf{E}^{a}_{i} = \frac{\mathsf{E}_{i}}{\mathsf{E}_{\mathsf{T}}} \times \mathsf{E}_{\mathsf{G}} \tag{6.57}$$

where,

 E_T is the summation of the ideal energy usage of the households. E_G is the generated energy by the off-grid microgrid.

Ideal values:

$$E_1 = 0.7845 \text{ kWh}$$

 $E_2 = 0.5075 \text{ kWh}$
 $E_3 = 0.6394 \text{ kWh}$
 $E_4 = 0.2914 \text{ kWh}$

Then, the energy allocation for each household can be calculated using equation 6.57 as follows:

$E_T = E_1 + E_2 + E_3 + E_4 = 2.2228 \text{ kWh}$	(6.58)
$E_{G} = 3.2 kWh$	(6.59)
$E_1^{\mathfrak{a}} = \frac{E_1}{E_{T}} \times E_{G}$	
$=\frac{0.7845}{2.2228} \times 3.2$	
E ₁ ^a =1.1294 kWh	(6.60)
$E_2^{\mathfrak{a}} = \frac{E_2}{E_{T}} \times E_{G}$	
$=\frac{0.5075}{2.2228} \times 3.2$	
E ₂ ^a =0.7306 kWh	(6.61)
$E_3^{\mathfrak{a}} = \frac{E_3}{E_{T}} \times E_{G}$	
$=\frac{0.6394}{2.2228} \times 3.2$	
E ₃ ^a =0.9205 kWh	(6.62)
$E_4^{\mathfrak{a}} = \frac{E_4}{E_{T}} \times E_{G}$	
$=\frac{0.2914}{2.2228} \times 3.2$	
E ₄ ^a =0.4195 kWh	(6.63)

For this example, J is 0.26 using ratio and proportion. Using the proportionality method can be straight forward. However, this method does not consider the constraints that are important when dealing with an optimisation problem. Thus, this method is not used in solving the optimisation problem defined in this chapter (Section 6.1).

6.2 **DISCUSSION OF RESULTS**

In this section, the answers to the research questions are presented. Using the formulated equations from the previous sections, the optimal energy allocation for each household is calculated. The performance of the proposed optimal energy allocation is evaluated using performance evaluation indices such as root-mean-squared error (RMSE), mean absolute percentage error (MAPE), and mean squared error (MSE). Also, to determine whether the differences in the proposed optimal energy allocation between households are statistically significant, one-way analysis of variance (ANOVA) was performed.

Household	Ν	Mean	Standard	Standard	Min	Max
		(kWh)	Deviation	Error		
H5	288	0.951	0.130	0.008	0.326	1.153
H6	288	0.770	0.165	0.010	0.454	1.104
H7	288	0.936	0.073	0.004	0.506	1.123
H8	288	0.536	0.091	0.005	0.390	0.820
Total	1152	0.798	0.115	0.007	0.419	1.050

Table 6.1: Descriptive summary of the proposed optimal daily energy allocation (kWh) for each household in Village 2

6.2.1 Optimal and adaptive energy allocation: consumer profiles

RQ2: Can knowledge of consumer profiles aid in optimal and adaptive energy allocation?

In order to show that the consumer profile aids the optimal energy allocation, it needs to be established that the different households are allocated with different daily energy over several days (>180 days). Figure 6.1 shows the daily energy allocation of each household for 288 days. The mean values are 0.9513 kWh, 0.7698 kWh, 0.9362 kWh, and 0.5360 kWh for H5, H6, H7 and H8, respectively. Table 6.1 show the descriptive summary of the proposed optimal energy allocation. H5 has the highest mean, followed by H7 and H6 with H8 having the lowest mean. This confirms the summary of data presented in Section 4.5, wherein Village 2, H5 has the highest peak from the daily energy usage, and H8 has the lowest. According to the ANOVA test results, the significance value (p–value) is 2.98×10^{-268} which is below 0.05 and therefore, there is a statistically significant difference in the means of the optimal energy allocations for each household. This rejects the null hypothesis that the means for each household are the same. The difference in means, as shown in Figure 6.1 is because the household information was used in forecast model to obtain the ideal energy utilisation of the households which are then used for the allocation.

To know specifically which households optimal energy allocations are different from which households, a t-test was performed. Table 6.2 shows the results of the t-test for each pair. Results show that the difference in means is statistically significant for all household except for H₅ and H₇ where the p–value is 0.1134.

The p-value also suggests that the proposed energy allocation used well the information given and that using the households profiles as inputs aid in deriving an optimal and adaptive energy allocation.



Figure 6.1: Daily energy allocation for each household using the proposed optimal energy allocation scheme for 288 days with mean values of 0.9513 kWh, 0.7698 kWh, 0.9362 kWh, and 0.5360 kWh for H5, H6, H7 and H8, respectively

Group	Group	Mean Difference	Standard	p – value
(A)	(B)	(A-B), (kWh)	Error	
H5	H6	0.181	0.016	1.470×10^{-24}
	H_7	0.015	0.010	0.113
	H8	0.415	0.009	$9.170 \times 10^{-132**}$
H6	H7	-0.166	0.012	$3.050 \times 10^{-35**}$
	H8	0.234	0.014	$4.240 \times 10^{-46**}$
H8	H7	0.400	0.007	3.130×10^{-161}

Table 6.2: Multiple comparisons of the mean of the proposed optimal energy allocations for each household

** The mean difference is significant at the 0.01 level.

6.2.2 Energy efficiency under the limited generation capacity

*RQ*₃: *Can the energy allocation be optimised to improve the energy efficiency under the limited generation capacity?*

This section provides the answers to the RQ3. In order to show that the energy allocation can be optimised, it needs to be established that the proposed optimal allocation achieves a statistically significant error (in terms of of the objective function) than the equal allocation or with the proportional allocation method. This error corresponds to the energy wastage. Analysis of Variance (ANOVA) is used to determine the difference between the three methods. ANOVA compares the means of the groups and determines whether the means are statistically different from each other, as discussed in Section 3.3.1. ANOVA test the null hypothesis, that is $H_0 = \phi_1 = \phi_2 = \phi_3 = \phi_n$ where ϕ is the group means, and n is the number of groups to be compared. If the ANOVA test results show statistically significant, the alternate hypothesis H_A is accepted that the group means are statistically different from each other.

The objective function, $J = \sum_{i=1}^{n} (E_i^a - E_i)^2$ (as described in Section 6.1), is determined by calculating the aggregated difference of the squared error between the calculated optimal allocations and the ideal allocations which are obtained from the predicted values from Chapter 5. From the calculated values of the optimal energy allocations using the proposed mathematical framework presented in Sections 6.1 and 6.1.1, the energy allocation is calculated optimally for each household. Since the forecast model used the consumer profiles as inputs along with the historical energy usage data, the forecast model becomes adaptive. The consumer profiles enable the forecast model to process the data corresponding to each household. The given consumer profile serves as an identifier when the result for the individual households is needed. With this approach, predicting the next day energy consumption of each household is possible with a single forecast model.

The results show (e.g. Table 6.3 for a given day) that with the proposed optimal energy allocation, the energy wastage/deficit can be reduced from 0.24 kWh to 0.11 kWh which is a 54% decrease from the current equal energy allocation for each household. These results show that the objective function of the optimisation problem is minimised. Using the proportionality approach, the aggregated difference is 0.26 kWh which is higher than the total difference between the ideal and the equal allocations, which is equal to 0.24 kWh. This means that when the proportionality method is used in allocating the energy of the households, higher energy wastage can occur.

House	Ideal	Equal	Proportion	Optimum	Error1	Error2	Error3
No.	(I)	(Eq)	(Pr _i)	(E_i^a)	$(I - Eq)^2$	$(I - Pr_i^a)^2$	$(I - E^{a}_{i})^{2}$
H5	0.78	0.80	1.13	1.02	0.00	0.12	0.05
H6	0.51	0.80	0.73	0.75	0.01	0.05	0.00
H_7	0.64	0.80	0.92	0.88	0.00	0.08	0.01
H8	0.29	0.80	0.42	0.55	0.23	0.02	0.05
Error $\left(\sum_{i=1}^{n} (I - Eq)^2\right)$ 0.24							
Error2	$\left(\sum_{i=1}^{n} ($	$I - Pr_i)^2$	2)			0.26	
Error3	$\left(\sum_{i=1}^{n} ($	$I - E_i^a)^2$)				0.11

Table 6.3: Comparison of ideal energy usage and the energy allocations (equal, proportion, and optimal) of the households

Figure 6.2, shows the ANOVA results for the three methods of allocations. The different performance of the three allocation methods is described by the difference in their means which is statistically significant with p-value of 2.24×10^{-38} . For 288 days, the aggregated squared error difference between ideal energy usage and the equal allocation is 112.10 kWh² or \pm 10.59 kWh, with proportionality the total difference is 73.02 kWh² or \pm 8.55 kWh and with the proposed optimal allocation, it is 62.35 kWh² or \pm 7.90 kWh (shown in Table 6.4). Thus, using the proposed optimal allocations, the difference between the ideal energy usage is minimised further compared with the other allocation using the method of proportions.

Figure 6.3, shows the difference between the ideal usage of the households and the equal allocation as well as the proposed optimal allocation. The squared error of the ideal and the proposed allocation is preferred to be smaller than the current set up with the equal allocation. From the results shown in Figure 6.3 with 288 days of data for all households, the aggregated squared error between the ideal usage and the proposed optimal energy allocation is 62.35 kWh² or \pm 7.90 kWh compared with the existing equal energy allocation which is 112.10 kWh² or \pm 10.59 kWh. This difference is equivalent to 44.4% reduction of energy wastage/deficit and 14.6% reduction when compared with the results using the proportion method.

To determine whether there is a statistically significant difference between the two allocations (equal and the proposed optimal allocations), the ANOVA statistical test was performed. The difference in means of the two groups are statistically different with a mean squared error (MSE) of 1.074 and p-value of 6.61×10^{-31} and F-value of 137.3. The p-value suggests that the means of the two groups (equal allocations and the proposed optimal allocations) are different and rejects the null hypothesis that assumes that the means of the groups



Figure 6.2: ANOVA results for the 3 allocations (Equal, proportion, and optimal), with 288 days data. The difference between the means of the 3 groups are statistically significant with a p-value of 2.24×10^{-38} and F-value is 88.9.

Table 6.4: Results for the objective function J for 288 days

Allocations	Results J
Equal	112.10
Proportion	73.02
Optimal	62.35

are the same. This implies that with equal daily energy allocation, energy wastage/deficit is higher compare with the proposed optimal daily energy allocation.

Another test is also conducted to observe the performance of the proposed optimal energy allocation scheme in terms of mean absolute percentage error (MAPE) for each household. The calculated MAPE for each household are 6.2, 8.0, 5.8 and 10.4 for H1, H2, H3 and H4, respectively.

In summary, for 288 days, the results of the objective function J is shown in Table 6.4. The proposed optimal energy allocation minimised the energy wastage by 44.4% when compared with equal energy allocation.

The proposed energy allocation assumes that any changes in energy usage of the households are reflected in their past usage. If the households deliberately change their energy usage, this is perceived in the historical energy usage data and accounted in forecasting their



Figure 6.3: Anova test results between the two allocations with N = 288. The means of the two groups have statistically significant differences with p-value of 6.61×10^{-31} .



Figure 6.4: Absolute percent error (APE) for each household using the proposed optimal energy allocation. The MAPE for each household are 6.2, 8.0, 5.8 and 10.4 for H1, H2, H3 and H4, respectively.
ideal usage. Thus, the proposed energy allocation remains optimal according to their previous energy usage.

From the results of this research, the proposed optimal energy allocation can be used for other households connected in microgrids. The proposed energy allocation scheme would generate similar results when used for households that posed similar energy usage and household profiles.

6.3 CHAPTER SUMMARY

Given the two issues highlighted in Chapter 1 with the current set up where daily energy allocation is equal for all households, there is a need to re-allocate the available energy to ensure that the provision of the energy to each household is based on their ideal energy needs. Thus, the re-allocation is done to minimise the unused energy allocated to the households that are not used and maximise the utilisation of the available energy.

Optimal energy allocation is proposed by considering the limited generation capacity of the power systems and the basic electricity needs of each household. The basic electricity needs of each household are taken from the households forecasted energy usage based on their historical usage and the household demographic profile as presented in Chapter 5. The consumer profiles used in the development of the forecast model aid in deriving an optimal and adaptive energy allocation.

The optimal allocation is calculated using the KKT conditions such that the aggregate optimal energy allocation summed up to the available energy generated by the power system and each optimal energy allocation to the households is higher or equal to the minimum energy required by the households. This ensures that the utilisation of the available energy is maximised and the unused energy of some household in the current set up is distributed accordingly to other households that need more than their allocated energy without sacrificing the energy needs of any household.

The optimisation problem is defined in which the cost function is to minimise the squared difference between the optimal allocation and the predicted ideal energy utilisation of the households. The cost function is then subject to the constraints involving the available energy of the power system and the basic energy needs of the households.

The mathematical framework for calculating the optimal energy allocation is unique and new in this field. All the derived equations are confirmed and validated by checking the results using the predicted values from Chapter 5 and assessing if the results have met the stated constraints. The method can be applied to households connected to generation-constrained microgrids.

The optimisation problem is successfully solved, and the allocation can be done adaptively and optimally. The demographics may vary, but the model can dynamically recognise this change and interpret accordingly based on the data given to the model. Thus, the method is considered robust and adaptive.

As presented in this chapter, the means of the optimal energy allocations for each household are significantly different from each group. This signifies that the consumer profiles help the proposed allocation scheme to allocate the energy optimally and adaptively. The energy usage under a microgrid with limited generation capacity was improved by reducing the energy wastage/deficit. Using the proposed optimal energy allocation, the energy wastage/deficit is reduced to 62.3 kWh from 112.1 kWh when using an equal daily energy allocation for each household. That is equivalent to 44.4% reduction of energy wastage/deficit achieved in 288 days. With the proportion method, the energy wastage/deficit is reduced to 73.02 kWh. So when the results with the proposed optimal energy allocation are compared with the results using the proportion method, a 14.6% reduction in energy wastage/deficit is achieved.

The proposed optimal energy allocation is expected to generate similar results when used for other households with similar characteristics, such as their energy usage and household profile. This is regardless of the connectivity of the households to the power system. Whether households are connected to the traditional grid or microgrid (online or offline), the proposed energy allocation is expected to provide optimal energy allocation when all the conditions stated in this chapter are met.

7

CONCLUSION

This research proposed a solution to the In this thesis, the need for optimising the energy allocation of the households connected in off-grid microgrids with limited generation capacity is investigated. The problem defined in Chapter 1 as re-stated here with an equal amount of energy allocated daily and considering the limited generation capacity of the off-grid power system, two main issues are highlighted:

1.) Some households use up the allocated daily quota and experience power outage but may require more energy than what is currently provided to them.

2.) Some households do not use all the allocated energy daily and pay the same fixed monthly tariffs as the households that use all their allocated energy.

To address the above issues, an optimal daily energy allocation for each household in off-grid villages such as the Red Cross village in Daanbantayan, Cebu is proposed. Below are the main considerations of the proposed optimal energy allocation for households connected in off-grid microgrids with limited generation capacity, re-stated from Chapter 1:

- Case 1: If the energy allocation is more than the basic needs for a given household, they pay for something they do not use. Therefore, they want an allocation commensurate with their usage.
- 2.) Case 2: If the allocation is less than the desired utilisation for a given household, they would be dissatisfied with the amount of energy they have, as they would experience shortage every day. Therefore, they want to have an energy allocation that is within their usage.
- 3.) The allocations in cases 1 and 2 are not guaranteed to be same.
- 4.) Energy allocation is constrained within the limited generation capacity of the off-grid power system.

Chapter 2 presents the research background on microgrids, energy management system (EMS) and artificial neural networks, and the review on existing literatures related to this study. The literature review covers topics on energy management scheme of microgrids, such as load shedding, time-based operation or scheduling and daily energy allowance or allocation, EMS optimisation, load forecasting, factors affecting load consumption, and the importance of the research data as the baseline data for the development of load forecast model as well as the proposed optimal energy allocation.

Chapter 4 presents collection and processing of data needed for this research. This includes the development and deployment of the Philippines Micro-Off-Grid monitoring system used to gather the energy data and the field survey used to collect demographic profiles of the households. All these data are used in the development of the load forecast model. The PMOG data from Village 1 was dropped as the data available is only for 6 hours a day and the operation of the off-grid power system in this village can not be available for 24 hours due to its technical limitations. In developing the forecast model, the data needs to be representing whole day energy usage that is 24 hours and not part of it. Hence, only data from Village 2 were utilised in the development and training of the forecast model.

From the data gathered, it was found out that some households do not used all the given energy daily quota and some households experienced power outage. Survey data shows that 46 % of the total households from Village 2 experienced power outage daily and 54 % of the total households do not use all up their given energy quota daily. There are 59 households out of 128 total households experienced power outage daily. They want to have more energy quota if possible as they expressed their willingness to pay according to their daily usage. In order to maximise the available energy, the excess or unused energy quota of the remaining 69 households can is re-distributed to the 59 households that needs more energy. Hence the needs of the optimisation of daily energy allocation to each household.

Since the aim of the research is to provide an optimal energy allocation based on the ideal usage of the households, there are 2 things that need to be done. First is to predict the next day energy consumption of the households and second is to calculate the optimal energy allocation using KKT approach.

The first step is done by the Multilayer Perceptron (MLP) -based forecast model. During the development of the model, it was found out that the number of hidden layers and neurons as well as the delay window of inputs affects greatly the performance of the neural network in terms of RMSE. Several cases were considered in selecting the combination of the hidden layers HL, neurons N and delay window D that generate the least value of RMSE. For MLP-based forecast model, RMSE values continue to change as the input varies. The variation is random especially with the average RMSE during the K-fold validation, however, the least RMSE values were achieved when household profiles were added as inputs as presented in section 5.2. Different load forecast models such as RBFN, GPR and ARIMA have been investigated to determine the best performing models in predicting households energy consumption in the context of off-grid villages connected to off-grid power systems with limited generation capacity (as detailed in Chapter 5). Incorporating household profiles as inputs made it possible for the MLP-based forecast model to met the threshold of 15 % or less of the actual energy usage. The household profiles that also serve as indexed of the different profile in input matrix of neural network made it possible to use single model to forecast the next-day energy consumption of each household.

The second step is done by formulating the optimisation problem. A mathematical approach for optimal energy allocation is presented in Chapter 6 that is based on MLP-based forecast model that uses both historical energy data and households profiles to achieve a reasonable accuracy in forecasting. The cost function J is to minimise the difference between the forecasted values E_i using MLP-based forecast model and the calculated optimal energy allocation E_i^{α} with 2 constraints derived from the limited capacity of the off-grid power system and the minimum threshold of the households (refer to equation 6.8). Given the limited generation capacity of the off-grid power system, the first constraint is that aggregated optimal energy allocation must be equal to the available energy. In this case, the data is from Village 2, and the off-grid power system is using batteries, the total available energy is treated as equal to the summation of the existing allocation (equation 6.9). The second constraint is that the optimal energy allocation E_i^a must be equal or higher than the daily energy threshold of each household E_{min} (equation 6.10). These 2 equations, 6.9 and 6.10 are the equality and inequality constraints of the objective function J (equation 6.8). The inequality constraint allows the optimisation problem to be solved using KKT conditions as detailed in Chapter 6.

The work presented in this thesis has therefore focused on the development of the MLP-based load forecast model and the adaptive and optimal energy allocation for households connected in generation-constrained microgrids.

7.1 ANSWERS TO RESEARCH QUESTIONS

This thesis sets out to answer the following research questions:

RQ1: Can the household's daily energy consumption be forecast with reasonable accuracy?

Answer: Yes.

The accuracy of the proposed adaptive MLP-based load forecast model is important in predicting the different households daily energy consumption given the limited generation capacity of the off-grid power system. Section 5.13 shows the performance of the MLP-based load forecast model compared to other load forecasting techniques. The model's performance is measured by RMSE and the time needed to train the model. The best performance of the model is with a network with 3 hidden layers and 20 neurons and 7 input delays for historical energy data. This was identified via a grid search over the neural network parameters by optimising the predictive accuracy and training time. The experimental process is presented in Section 5.2.4.

The performance of the model is further improved by integrating the household profiles as inputs. The proposed MLP-based forecast model can forecast the next-day energy consumption of each household without the need of different model for each profile. This is possible as the model is developed and trained using different household profile as inputs. The model is able to recognise the different profiles accordingly. Although the performance of the proposed MLPbased forecast model maybe further improved by incorporating more than 4 households profiles and more than 2 years of historical data. Incorporating historical energy data at appliance level as input may also improve the performance of the model in forecasting. In this study, it was planned to have more PMOG system to be deployed and gather more energy data however, due to financial constraints, this plan was never materialised.

RQ2: Can knowledge of consumer profiles aid in optimal and adaptive energy allocation?

Answer: Yes.

Integrating the consumer profiles as inputs for the forecast model, enables the forecast model to predict the energy consumption of the households uniquely. As discussed in Chapter 6, the optimal energy allocation is based on the forecasted values by the MLP-based forecast model presented in Chapter 5. The optimal energy allocation is adaptive because the household profiles are incorporated in the load forecast model as inputs. Households profiles are included to the typical input variables such as historical energy data, temperature and calendar days, in developing a load forecast model.

Households profiles were created based on survey data. There are four consumer profiles identified based on the households' number of occupants, monthly income, number of appliance, frequency of use (in terms of hours for all appliances), number of children in school and with age less than 5 (who are expected to stay home during the day), and number of working family members. The identified consumer profiles are detailed in Section 4.8. Consumer profiles as inputs improves the accuracy of the forecast model. From 20.11 % error down to 13.58 % error without and with consumer profile as input, respectively. Results presented in Section 6.2.1 show that each household has unique energy allocations, based on the optimal algorithm after incorporating the households profiles in the forecast model. According to the ANOVA results, the difference between the means of the energy allocations for each household is statistically significant with p–value of 2.98×10^{-268} .

Given the results of the experiments and analysis of results, it can be concluded that the consumer profiles indeed aid in determining the optimal and adaptive energy allocation of the households. This approach is expected to work when applied to households with similar profile of households in Village 2 as presented in this study.

RQ3: Can the energy allocation be optimised to improve the energy efficiency under the limited generation capacity?

Answer: Yes.

MLP-based forecast model is proposed that uses both historical data and household demographic information to predict the household's day-ahead energy consumption. Using the forecasted values, the optimal energy allocation is then calculated using Karush-Kuhn-Tucker conditions from solving the optimisation problem (stated in Section 6.1) subject to two equality and inequality constraints related to the limited generation capacity of the microgrids and a certain guaranteed energy quota for each household. By providing an optimal energy allocation to each household, the energy wastage in terms of unused energy as well as energy deficit to some households is minimised.

The optimisation problem is defined with a main goal of reducing the difference between the allocated energy and the actual energy consumption in order to maximise the energy consumption in a generation-constrained microgrids. Using the proposed optimal energy allocation, the energy efficiency of off-grid microgrids with limited generation capacity is improved. In this study, the off-grid microgrid with limited generation capacity is said to be efficient when the energy wastage is minimum. The energy wastage is the unused daily energy allocation of the households. According to the simulated results, the energy wastage/deficit is reduced by 44.4% with the proposed optimal and adaptive energy allocation compared to the existing scheme that provides an equal energy allocation in 288 days, and 14.6% compared with a proportional allocation.

As detailed in Chapters 5 and 6, the proposed methodology of allocating energy to the households in generation-constrained microgrids that is adaptive and optimal is successfully investigated and validated in this thesis. The proposed methodology is a combination of load forecasting using MLP and optimisation using KKT approach. The set threshold of RMSE is achieved by employing MLP-based forecast model and incorporating the household profiles as input together with the historical data. The % error is improved by 32.5 % when household profile is included as input. Although this performance can be further improved with more household profiles and historical energy data. When the performance of the forecast model to predict the next-day energy consumption is improved, this also have a positive effect in calculating the optimal energy allocation of the households. The objective function of the optimisation problem defined in this study is to minimised the error difference between the forecasted value of the next-day energy consumption of the households and the calculated optimal energy allocation. When the forecasted values can be made more accurate by improving the performance of the MLP-based forecast model, the optimal energy allocation will also be more accurate as expressed in equation 6.38. Thus, minimising the objective function J. This proposed optimisation using KKT approach is applicable to any optimisation problem that satisfies the KKT conditions. The proposed methodology of determining the optimum energy allocation is applicable to other households in off-grid communities with similar household profile and off-grid power systems.

7.2 CONTRIBUTIONS TO KNOWLEDGE

From the answers to the research questions, the following are the contributions of this work:

- A generalised MLP-based load forecasting model that uses historical data and considers household demographic information to predict the household's day-ahead energy demand.
- A methodology for optimal energy allocation that is adaptive to individual households – optimal because it minimises the energy waste/deficit, and adaptive because it uses the households historical data and demographic information.

The allocation of energy quotas is considered as an optimisation problem [Tan+15; Tia+14], in which the objective is to minimise the collective energy waste as well as minimised energy deficit, in terms

of unused allocated energy, while ensuring the generation capacity is not exceeded. The historical usage are modelled and combined with other extracted features for demand prediction to derive an optimal energy allocation based on the forecasted ideal demand.

7.3 SIGNIFICANCE OF THE STUDY

Apart from the direct benefits to the selected villages where the research will be deployed, the work described in this thesis will be beneficial to other researchers interested in micro-grids and energy management systems for off-grid systems, as well as to the energy providers in the Philippines. The integration of the social aspect (household individual profile) into the load forecasting model along with the energy historical usage of each household increases the model's capacity for accuracy and thus generates real-world results.

The community is empowered by giving them an opportunity to use their limited energy supply more efficiently with less energy waste, and incorporate their individual requirements by considering households profile in developing the load forecasting model.

This work can be a good reference in planning and design considerations for energy management systems for off-grid implementations in remote areas.

7.4 FUTURE WORK

In this thesis, a methodology for optimal energy allocation that is adaptive to individual household is proposed. The following are the possible avenue to expand on the work in this thesis:

- The work can be further improved by including all the inherent losses on generation and distribution lines of the microgrid into the calculation of the energy allocation.
- 2. The work can be applied in smart grids that want to predict the next day load based on the historical data and the consumer profiles as laid out in this thesis and with possible inclusion of appliance level historical data.
- 3. Real-time updating of data for dynamic forecasting and energy allocation. Forecasting can be more efficient when the data is updated in real-time and the energy allocation is dynamically changing corresponding to the historical energy data.
- 4. The work can be used as the basis for energy management financial scheme that would ensure the sustainability of the off-

grid microgrid operation while maintaining the ideal energy usage of the households.

- 5. This work can be further improved by using Deep Neural Network architecture and/or any other hybrid models for load forecasting.
- 6. The performance of MLP-based forecast model can be further improved by using more than 4 household profiles and deploy PMOG systems to gather data for each profile.
- 7. Optimisation approach may be improved further by utilising other optimisation techniques such as sequential quadratic programming (SQP) and genetic algorithm (GA) that essentially minimise the objective function J.

This work is limited by the available data and the implementation of the proposed energy allocation scheme is possible with sufficient financial support and collaboration with the maker of energy management systems (EMS) of off-grid microgrids.

A

DEPLOYMENT DOCUMENTS

For data gathering, PMOG systems are installed to the selected households and a field survey were done. Attached here are the documents on the installation of PMOG system and the survey questionnaire.

A.1 PMOG SYSTEM RASPBERRY PI SET UP GUIDE

PMOG Raspberry PI EMS setup

This details the process of installing the PMOG software onto a Raspbian Install

Contents:

- Preparing the SD Card
- Initial configuration and package installation
- RTC setup
- Setup SSH
- Install PMOG code
- Setup AutoSSH

Preparing the SD Card

- 1. Download the latest copy of the Raspbian Lite from https://www.raspberrypi.org/downloads/raspbian/
- 2. Copy onto the SD Card (either use dd, or PiBaker)
- 3. Plug the SD card into the Pi (with keyboard/monitor attached), Apply Power
- 4. We should now be able to login to the PI (using standard pi/raspberry)

Initial Configuration and package installation

- 1. Update the source list and upgrade any packages
 - sudo apt-get update && sudo apt-get upgrade
- 2. Configure the raspberry pi sudo raspi-config
- 3. Select the following choices
 - Expand filesystem Change user password
 - Advance options

A3 set gpu to 0

A4 SSH set to yes

reboot

4. Install all the required packages

sudo apt-get install git mercurial subversion emacs python-smbus autossh oracle sudo apt-get install python-setuptools python-dev python-docutils sudo easy_install pyserial sudo easy_install requests

 Configure Hostnames sudo emacs /etc/hostname (Check that it is PMOG No.?) sudo emacs/etc/hosts (Is 127.0.1.1 PMOG No.?)

RTC setup

For the software and hardware installation of RTC to pi, refer to PiFace Real Time Clock user guide available at:

http://www.piface.org.uk/assets/piface_clock/PiFaceClockguide.pdf

Setup SSH

1. Generate SSH key

shh-keygen -t rsa

2. Copy key and ask someone with sudo access to add to pi user on cogentee (James/Ross)

cat .ssh/id_rsa.pub

3. Test the ssh connection by

ssh cogentee.coventry.ac.uk

Reply if necessary to prompt

Install PMOG Code

1. Get PMOG code from subversion

mkdir -/svn

cd ~/svn

svn co svn+ssh://cogentee.coventry.ac.uk/svn/PMOG

2. Move folder to opt

sudo mv PMOG /opt/PMOG

3. Make data directories

sudo mkdir /var/log/ch

4. Install python code

cd opt/PMOG sudo python setup.py develop

5. Install start-up scripts

cd /opt/PMOG/etc

sudo cp ch-CurrentCost /etc/init.d/

sudo upgrade-rc.d ch-currentcost defaults

sudo update-rc.d ch-currentcost enable

Setup AutoSSH

1. Copy auto-ssh to initi.d

cd /opt/PMOG/etc

sudo cp auto-ssh /etc/init.d/

sudo chmod 755 /etc/init.d/auto-ssh

2. Run the following in a shell, replacing the XXX with an unused number

Echo REMOTE_PORT=16xxx | sudo tee /etc/default/auto-ssh

3. Set to start at boot

sudo update-rc.d auto-ssh defaults

sudo update-rc.d auto-ssh enable

sudo reboot

4. Check if you can connect through cogentee

A.2 PMOG SYSTEMS INSTALLATION MANUAL

PMOG-CurrentCost Deployment Guide

A. Equipment

1 Server Used to collect and store data from the Wireless Sensor Network (WSN), and push data via 3g to the Cogent Computing Server. Composed of raspberry pi and the 3G/4G dongle with aerial. (Labelled as **PMOG1** ~ **PMOGN**)

1 Current Cost Node Senses the electricity consumption of the household includes Current Cost set and Current Cost Individual Appliance Monitor (IAM)

1 Router Used to access the server during installation

2 Ethernet Cables Used during installation

B. Installation

- 1. Map the house to take note where the raspberry pi with the dongle to be installed.
- 2. Locate the electric main panel/meter and note the desired point of installation for the raspberry pi and Current Cost display monitor.
- 3. Set up the Current Cost devices and the IAM following the instructions in the manual attached (make sure you get a reading on the display).

Note: Log the name of the appliances, (if possible with serial no.) plugged in with each individual appliance monitors for each household to the system monitoring log sheet.

- 4. Connect the Current Cost display monitor to the raspberry pi by plugging in the interface cable between the two.
- 5. Plug in the Router.
- 6. Using the Ethernet cables, connect both the pi and your laptop to the router
- Plug in the dongle to the raspberry pi. (Note the flashing light: Green connected to 4G network, blue – 3G network, RED light means not connected or no signal)
- 8. Plug in raspberry pi, check all lights come on.
- 9. To check if system is logging data, navigate to <u>http://<servername>.local/PMOG/</u> or

open a terminal window and run the following:

ssh cogentee ssh pi@servername.coventry.ac.uk

> (reply if necessary to prompt, when successfully logged in, do the following)

cd /tmp/ ls cat <filename>

(files containing data should appear) (to show the contents of the file, choose any file from the directory shown)

10. Make sure the node and devices are listed under sender and device column at the web page.

A.3 SURVEY QUESTIONNAIRE

Cogent Laboratory Coventry University, UK

Survey Questionnaire

Are you interested in participating in a "user-driven" energy project?

Yes
No

l 1)	<i>Demographic Profiling</i> Locatio <u>n:</u>		Date: Control No
2)	Name:		
3)	Role in the family:	6)	Educational Attainment:
	Father		Elementary
	Mother		High School
	Head of the family		College Level
4)	Others:Others:		Vocational/technology College graduate
5)	Gender:		Others
7)	House Occupancy: Single Couple	8)	Occupation: (Please tick one. If not on the list, please specify) Farmer
	working		Fisherman
	retired		Entreprenuer
	Family with children		 Teacher
	< 5 yrs. old		Brgy. Official/Worker
	5 -18 yrs. old		Others: (specify)
	> 18 yrs. old		
	Extended Family (with grandparents)		
	Others:		
9)	Total number of people living in the house:		-
10)	Number of children studying(or in school):		
11)	No. of family members who are working:		
12)	Monthly Salary/Income (₱)		
	Range		Range
	less than 2000		10001 - 15000
	2000 - 5000		15001 above
	5001 - 8000		Others, please specify:
	8001 - 10000		
13) 14)	Other source of income: Total Monthly/Daily income:		

Cogent Laboratory Coventry University, UK

Ene Wh	e rgy Consumption Profiling to is in the house?		During the	During the	
example:		Age	day	night	
	Josephine dela Cruz	30	No	6 PM-7 AM	
	Juan dela Cruz	32	8 AM - 5 PM	No	
1					
2					
3					
4					
5					
6					
7					
8					
9					
10					

2) Do you have appliances? (Please tick below and indicate how often do you use it)

		Everyday (Ave. hours of use)	once a week (Ave. hours of use)	Twice a week (Ave. hours of use)	Others, please specify
	Light bulbs and/or flourescent				
	TV				
	Refrigerator				
	Electric/ceiling fan				
	Electric kettle				
	Rice Cooker				
	Electric Stove				
	Others (List the other appliances you used at l	east once a week)		
	1				
	2				
	3				
	4				
	5				
3)	Monthly electric bill (₱)				
4)	Do you use smart phones?	No)		
5)	Do you have access to 24-hour energy?	Yes	s (Proceed to Section I s (Proceed to questior	II 6	
6)	If you will have 24-hour access to energy consumption?	gy, what wou	uld change in	your	

Ш

1)

	Cog Coven	gent Laboratory Itry University, UK
 1.)	User-empowerment profiling Please rank the following according to your priorities, with 10 as the highest and 1 as the lowest. (Please place the numbered smileys on the space provided)	
	Water	employment/job
	Sanitation	Community Livelihood program
	Road	Health Facilities (i.e. clinic/hospital)
	electricity	education/school
Gene	Housing	flood/typhoon protection

Cogent Laboratory Coventry University, UK

 How aware are you of the environmental impact of energy use? (Please encircle below : 5 - highest, 1 - lowest)



3.) How motivated are you to save energy? (Please encircle below : 5 - highest, 1 - lowest)



4.) How engaged are you in your local community?

(Please encircle below: 5 - Always, 4 - Often 3- Sometimes, 2 -Occationally , 1 - Never)

4a. I participate in every activity in the local community



4b. I take part on the decision making for my local community welfare, development and etc.



5.) How influential are you at changing how your community uses energy? (Please encircle below : 5 - highest, 1 - lowest)



A.4 DEPLOYMENT TECHNICAL REPORT

This document is available upon request.

B

FIELD WORK: PMOG SYSTEMS INSTALLATION

There were 10 PMOG systems installed in the selected households and microgrid at the two villages; Village 1 and Village 2. These two villages are both powered by an off-grid power system with limited generation capacity. Each village has 5 PMOG systems, 1 for monitoring the generation side and 4 for the households representing the majority of the households' profile. Households were selected after the survey was done where the demographic information was already gathered. The number of households with the same profile was considered in the selection process. Parameters such as number of appliances, number of occupancies and total household's monthly income were considered in the selection process. The number of households with the same number of those three parameters was tallied. Then one household was selected from the four most numbers of households with the same profile to represent them. Figure B.1 shows the actual PMOG system installed in the households.

The CT jaw sensor of the PMOG was clamped to the main electric panel of the household, and the wireless transmitter was fixed on the wall (as shown in Figure B.1). The sensor is now ready to gather data as the aggregate energy consumption of the household. To gather the data at the appliance level, the individual appliance monitor was deployed. Each appliance was attached with an IAM to determine the amount of energy used. IAM was plugged into the wall socket, and the selected load (appliance) was plugged into the IAM.

CT jaw sensor and each IAM were paired with the Envi display monitor. The wireless communication between the sensors and the Envi display monitor was confirmed by checking the display monitor whether it was showing energy usage of the households from CT jaw sensor and appliances from IAMs before fixing all the wires on the wall. Internet connection was then checked and confirmed during the installation process by checking the webpage in real-time to make sure that the system was working properly before fixing the devices in the household. The webpage, which is running from the remote server, is expected to update the displayed data every 20 minutes after every hour. When the webpage is not updated, PMOG system may need to be checked to confirm its status. The PMOG system is designed to store data in its local server. The CT jaw sensors and the appliance monitors sensed the data every 6 seconds and transmitted it to the



PMOG system installed at the power plant (generation side)



Figure B.1: PMOG systems deployed in the villages collecting energy data on both the off-grid power systems (generation side) and the households with individual appliance monitors (IAMs). CC monitor and logged in to the Rpi storage. The PMOG system, through its internet dongle, transmits the sensed data from its local server (household) to the remote server (Cogent laboratory) every 20 minutes past hour.

Data logging at the webpage created for the PMOG system was checked every day to monitor the performance of all sensors. The PMOG systems were expected to send data every hour if they were working properly. The last transmission can be checked at the webpage. When the last the transmission was successful, the webpage automatically updates and displays the most current data received. If not, and the last transmission is already one day past, the person-incharge in the Philippines was contacted to check with the household what causes the missing transmissions and resolve the issues remotely when necessary. Common causes of missing transmission include network problems such as no internet connection, and PMOG systems were not working and turned off.

Figure B.2 is the actual webpage of the PMOG monitoring system. The webpages display sender, device number, server time for both local and remotes servers, and the readings from IAMs, CT jaw sensor and built-in temperature sensor. Appliances monitored by the IAMS were assigned to a specific channel display in the Envi display monitor in which the devices number is displayed on the webpage. Envi display monitor can display individual energy consumption of the appliances through IAMs with channel number from 1 to 9 with channel o reserved for CT jaw sensor. The transmitted data is retrieved from the remote server designated to the PMOG system by accessing the server from any computer connected to the internet. The data is in the form of .csv file that can be viewed by a Microsoft excel. The data includes the date and time, electricity usage (power (W)), temperature (degree C), device number, and the server number of the nodes.

is - genepale	ncia@gmail.com - G	imail ×	cogentee.co	wentry.ac.uk/PMOG	/
Jode I	2 onort	Interface			
ioue i	xeport 1	litterrace			
					Power_Usage (W)
PMOG10	0	2016-10-14 10:01:52	2016-10-14 11:01:52	27.80	19.00
PMOG1	0	2016-10-14 10:04:02	2016-10-14 11:04:02	26.00	8930.00
PMOG2	0	2016-10-14 10:04:33	2016-10-14 11:04:33	35.40	31.00
PMOG2	1	2016-10-14 10:04:33	2016-10-14 11:04:33	35.40	9.00
PMOG3	0	2016-10-04 09:57:59	2016-10-04 10:57:59	33.80	76.00
PMOG3	1	2016-10-04 09:57:56	2016-10-04 10:57:56	33.80	0.00
PMOG3	2	2016-10-04 09:57:57	2016-10-04 10:57:57	33.80	0.00
PMOG4	0	2016-10-14 10:04:10	2016-10-14 11:04:10	28.20	183.00
PMOG4	1	2016-10-14 10:04:07	2016-10-14 11:04:07	28.20	5.00
PMOG5	0	2016-10-13 14:04:21	2016-10-13 15:04:21	31.30	31.00
PMOG6	0	2016-10-14 10:03:14	2016-10-14 11:03:14	27.10	1296.00
MOG7	0	2016-10-14 10:06:34	2016-10-14 11:06:34	27.80	60.00
MOG7	1	2016-10-12 14:24:59	2016-10-12 15:24:59	28.40	0.00
PMOG8	0	2016-10-08 16:35:12	2016-10-08 17:35:12	28.10	7.00
PHOCO	0	2016-10-14 04-56-11	2016-10-14 05:56:11	28.10	2.00

last loaded: 2016-10-14 10:26:52

Figure B.2: PMOG system node report interface. The webpage displays the newest transmitted data to the remote server. The data include the date and time, device number, and PMOG number aside from the main data which are the temperature and power usage of the household

C

RESEARCH DATA

For this research, data refers to the PMOG data and survey data as described in Chapter 4. These data are stored and can be accessed at Cogentee repository [Pal19]. Both raw data and pre-processed data are available at this webpage: http://cogentee.coventry.ac.uk/ gene/re-searchdata/)

D

ETHICS APPROVAL

This research has undergone the ethical approval process of Coventry University. Attached here are the certificates, and application documents such as the participants information sheet and consent form.



Certificate of Ethical Approval

Applicant:

Gene Fe Palencia

Project Title:

Wireless Sensor Networks To Enable User-Driven Control Of Micro-Power Generation Devices: Baseline Survey

This is to certify that the above named applicant has completed the Coventry University Ethical Approval process and their project has been confirmed and approved as Medium Risk

Date of approval:

25 April 2016

Project Reference Number:

P42996



Certificate of Ethical Approval

Applicant:

Gene Fe Palencia

Project Title:

Wireless Sensor Networks To Enable User-Driven Control Of Micro-Power Generation Devices

This is to certify that the above named applicant has completed the Coventry University Ethical Approval process and their project has been confirmed and approved as Medium Risk

Date of approval:

25 July 2017

Project Reference Number:

P60318



Certificate of Ethical Approval

Applicant:

Gene Fe Palencia

Project Title:

WirelessSensor NetworksToEnableUser-DrivenControl Of Micro- Power Generation Devices

This is to certify that the above named applicant has completed the Coventry University Ethical Approval process and their project has been confirmed and approved as Medium Risk

Date of approval:

22 August 2018

Project Reference Number:

P72455



PARTICIPANT INFORMATION SHEET

Title of the Project: Wireless sensor networks to enable user-driven control of micropower generation devices – Baseline Survey

This research project aims to investigate the local community's power generation and utilisation of energy generated by the PV plant and to collect primary data that serves as a baseline for each load's power consumption. We are trying to establish a good understanding of the needs and aspirations of the end-user that may affect the energy supply and demand within the local community. The proposed project will undertake a range of monitoring and evaluation activities such as pre-post surveys, interviews with audio or video recordings and pictures.

Your participation in the pre-post surveys and interviews will be a great help in the study. You will not be identified in any of the analyses, reports or publications we produce and we will safeguard your confidentiality.

Participation is entirely voluntary and your help is very much appreciated. If at any time you want to withdraw from the study without prejudice you have only to let us know. In that event, please contact Gene (details below).

This study has been cleared by the ethics committees of Coventry University. You are free to discuss your participation in this study with the project staff (contactable on phone numbers or email below). If you would like to speak to an officer of the University not involved in the study, you may contact the Ethics Officer on $+44\ 2477658278$.

Researchers Contact Details:

Dr. Anh Tran Engr. Gene Fe P. Palencia Faculty of Engineering, Computing & Environment Coventry University

Phone: +44 2477 658299 (Anh) +44 7435 350757 (Gene) Email: <u>anh.tran@coventry.ac.uk</u> (Dr. Anh) <u>palencig@uni.coventry.ac.uk</u> (Gene)



CONSENT FORM

Title of the Project: Wireless sensor networks to enable user-driven control of micropower generation devices – Baseline Survey

- I have read and understand the attached participant information sheet and by signing below, I consent to participate in this study. I have retained a copy of this sheet for future reference if required.
- I understand that this research project will entail surveys, interviews which will involve images, audio or video recordings for the process of understanding the current conditions of my local community with respect to its energy generation and utilisation.
- I understand that any information taken in note form, pictures, orally or by video will be treated with the utmost confidentiality.
- I understand that participation in the research is voluntary, and I can withdraw at any time without giving a reason.
- I understand that I may choose not to answer particular questions, without being obliged to withdraw completely from the research.
- I understand if I wish to withdraw from the research, I can do so at any time without prejudice.
- I understand that the research has been approved by the Coventry University's Ethical Review Committee.
- I have voluntarily agreed to participate in this research project.

Participant's Name (Please Print)
Participant's Signature
Date:
Witnessed by (Please Print)
Witness's Signature
Researcher's Signature

Researchers Contact Details:

Dr. Anh Tran Engr. Gene Fe P. Palencia Faculty of Engineering, Computing & Environment Coventry University

Phone: +44 2477 658299 (Anh) +44 7435 350757 (Gene) Email: <u>anh.tran@coventry.ac.uk</u> (Dr. Anh) <u>palencig@uni.coventry.ac.uk</u> (Gene)

BIBLIOGRAPHY

- [Abe+20] Jhud Mikhail Aberilla, Alejandro Gallego-Schmid, Laurence Stamford, and Adisa Azapagic. "Design and environmental sustainability assessment of small-scale offgrid energy systems for remote rural communities". In: *Applied Energy* 258.October 2019 (2020), p. 114004. ISSN: 03062619. DOI: 10.1016/j.apenergy.2019.114004. URL: https://doi.org/10.1016/j.apenergy.2019.114004 (cit. on p. 14).
- [AS15] Helene Ahlborg and Martin Sjostedt. "Small-scale hydropower in Africa: Socio-technical designs for renewable energy in Tanzanian villages". In: *Energy Research and Social Science* 5 (2015), pp. 20–33. ISSN: 22146296. DOI: 10.1016/j.erss.2014.12.017. URL: http://dx.doi.org/10.1016/j.erss.2014.12.017 (cit. on p. 15).
- [AR20] Seyed Ehsan Ahmadi and Navid Rezaei. "A new isolated renewable based multi microgrid optimal energy management system considering uncertainty and demand response". In: International Journal of Electrical Power and Energy Systems 118.December 2019 (2020), p. 105760. ISSN: 01420615. DOI: 10.1016/j.ijepes.2019.105760. URL: https://doi.org/10.1016/j.ijepes.2019.105760 (cit. on p. 12).
- [AAM20] Ahmed S. Alahmed and Mohammad M. Al-Muhaini.
 "An intelligent load priority list-based integrated energy management system in microgrids". In: *Electric Power Systems Research* 185.May (2020), p. 106404. ISSN: 03787796.
 DOI: 10.1016/j.epsr.2020.106404. URL: https://doi.org/10.1016/j.epsr.2020.106404 (cit. on p. 19).
- [ARP16] Ugonna R. Anuebunwa, Haile Selassie Rajamani, and Prashant Pillai. "Identification of realistic scheduling opportunities and base loads in smart homes using dynamic pricing and standard deviation of load profiles". In: Asia-Pacific Power and Energy Engineering Conference, APPEEC 2016-December (2016), pp. 222–226. ISSN: 21574847. DOI: 10.1109/APPEEC.2016.7779501 (cit. on p. 2).
- [Bak+17] Nur Najihah Abu Bakar, Mohammad Yusri Hassan, Mohamad Fani Sulaima, Mohamad Na im Mohd Nasir, and
Aziah Khamis. "Microgrid and load shedding scheme during islanded mode: A review". In: *Renewable and Sustainable Energy Reviews* 71.June 2015 (2017), pp. 161–169. ISSN: 18790690. DOI: 10.1016/j.rser.2016.12.049. URL: http://dx.doi.org/10.1016/j.rser.2016.12.049 (cit. on p. 19).

- [BGK15] Arjun Baliyan, Kumar Gaurav, and Sudhansu Kumar. "A Review of Short Term Load Forecasting using Artificial Neural Network Models". In: *Procedia - Procedia Computer Science* 48.Iccc (2015), pp. 121–125. ISSN: 1877-0509. DOI: 10.1016/j.procs.2015.04.160. URL: http://dx.doi. org/10.1016/j.procs.2015.04.160 (cit. on p. 5).
- [Baso7] Z A Bashir. "Short-term Load Forecasting using Artificial Neural Network based on Particle Swarm Optimization Algorithm". In: Departemen of Electrical & Computer Engineering Dallhousie University On page.s (2007), pp. 272– 275. DOI: 10.1109/CCECE.2007.74 (cit. on p. 29).
- [BK17] Merve Bedir and Emre C. Kara. "Behavioral patterns and profiles of electricity consumption in dutch dwellings". In: *Energy and Buildings* 150 (2017), pp. 339–352. ISSN: 03787788. DOI: 10.1016/j.enbuild.2017.06.015. URL: http://dx.doi.org/10.1016/j.enbuild.2017.06.015 (cit. on pp. 31, 32, 37).
- [Bi+o5] Weixing Bi, Xugang Wang, Zheng Tang, and Hiroki Tamura. "Avoiding the local minima problem in backpropagation algorithm with modified error function". In: *IEICE Transactions on Fundamentals of Electronics, Communications and Computer Sciences* E88-A.12 (2005), pp. 3645–3653. ISSN: 17451337. DOI: 10.1093/ietfec/e88-a.12.3645 (cit. on p. 28).
- [Bor+17] Kianoosh G. Boroojeni, M. Hadi Amini, Shahab Bahrami, S. S. Iyengar, Arif I. Sarwat, and Orkun Karabasoglu. "A novel multi-time-scale modeling for electric power demand forecasting: From short-term to medium-term horizon". In: *Electric Power Systems Research* 142 (2017), pp. 58–73. ISSN: 03787796. DOI: 10.1016/j.epsr.2016.08. 031. URL: http://dx.doi.org/10.1016/j.epsr.2016.08. 031 (cit. on pp. 25, 26).
- [BK07] Ralf Brand and Andrew Karvonen. "The ecosystem of expertise: complementary knowledges for sustainable development". In: *Sustainability: Science, Practice, & Policy* 3.1 (2007), pp. 21–31. ISSN: 15487733 (cit. on p. 14).

- [Bye19] Logan Byers. Global Power Plant Database. https: //github.com/wri/global - power - plant - database. 2019 (cit. on p. 33).
- [Car+17] G. Carpinelli, F. Mottola, D. Proto, and P. Varilone. "Minimizing unbalances in low-voltage microgrids: Optimal scheduling of distributed resources". In: *Applied Energy* 191 (2017), pp. 170–182. ISSN: 03062619. DOI: 10.1016/j.apenergy.2017.01.057. URL: http://dx.doi.org/10.1016/j.apenergy.2017.01.057 (cit. on p. 21).
- [CMP14] Guido Carpinelli, Fabio Mottola, and Daniela Proto. "Optimal scheduling of a microgrid with demand response resources". In: *IET Generation, Transmission and Distribution* 8.12 (2014), pp. 1891–1899. ISSN: 17518687. DOI: 10.1049/iet-gtd.2013.0758 (cit. on p. 21).
- [Cha+16a] Young Tae Chae, Raya Horesh, Youngdeok Hwang, and Young M. Lee. "Artificial neural network model for forecasting sub-hourly electricity usage in commercial buildings". In: *Energy and Buildings* 111 (2016), pp. 184–194. ISSN: 03787788. DOI: 10.1016/j.enbuild.2015.11.045. URL: http://dx.doi.org/10.1016/j.enbuild.2015.11. 045 (cit. on p. 29).
- [Cha+16b] Young Tae Chae, Raya Horesh, Youngdeok Hwang, and Young M. Lee. "Artificial neural network model for forecasting sub-hourly electricity usage in commercial buildings". In: *Energy and Buildings* 111 (2016), pp. 184–194. ISSN: 03787788. DOI: 10.1016/j.enbuild.2015.11.045. URL: http://dx.doi.org/10.1016/j.enbuild.2015.11. 045 (cit. on pp. 31, 49, 73).
- [CD14] T. Chai and R. R. Draxler. "Root mean square error (RMSE) or mean absolute error (MAE)? -Arguments against avoiding RMSE in the literature". In: *Geoscientific Model Development* 7.3 (2014), pp. 1247–1250. ISSN: 19919603. DOI: 10.5194/gmd-7-1247-2014 (cit. on p. 76).
- [Chaoo] Chris Chatfield. *Time-Series Forecasting*. 2000. ISBN: 1584880635. DOI: 10.1198/tech.2002.s718 (cit. on p. 26).
- [Che17] Yi Tui Chen. "The factors affecting electricity consumption and the consumption characteristics in the residential sector—a case example of Taiwan". In: *Sustainability* (*Switzerland*) 9.8 (2017). ISSN: 20711050. DOI: 10.3390/ su9081484 (cit. on pp. 6, 31, 32, 37).

- [Chi+16] Hsiao-Dong Chiang, Li-Guo Chen, Rong-Peng Liu, and Na Dong. "Group-based chaos genetic algorithm and non-linear ensemble of neural networks for short-term load forecasting". In: *IET Generation, Transmission & Distribution* 10.6 (2016), pp. 1440–1447. ISSN: 1751-8687. DOI: 10.1049/iet-gtd.2015.1068. URL: http://digitallibrary.theiet.org/content/journals/10.1049/ietgtd.2015.1068 (cit. on p. 30).
- [Chi+08] V. Chin, Z. Y. Dong, T. K. Saha, J. Ford, and J. Zhang.
 "Adaptive and optimal under frequency load shedding". In: 2008 Australasian Universities Power Engineering Conference. 2008, pp. 1–6 (cit. on p. 20).
- [Chi+15] Hamed Chitsaz, Hamid Shaker, Hamidreza Zareipour, David Wood, and Nima Amjady. "Short-term electricity load forecasting of buildings in microgrids". In: *Energy* & Buildings 99 (2015), pp. 50–60. ISSN: 0378-7788. DOI: 10.1016/j.enbuild.2015.04.011. URL: http://dx.doi. org/10.1016/j.enbuild.2015.04.011 (cit. on p. 24).
- [Cog15] Cogent Laboratory, Coventry University. Cogent-House: Data Sets from home monitoring Wireless Sensor Network. 2015. URL: http://cogentee.coventry.ac.uk/datasets/ Cogent-House-Data.htmlf (cit. on p. 50).
- [Cur] *Current Cost Products: EnviR and Transmitter*. http://www. currentcost.com/products.html. Accessed: 5th May 2020 (cit. on p. 51).
- [D'A+17] F. D'Agostino, S. Massucco, F. Silvestro, A. Fidigatti, F. Monachesi, and E. Ragaini. "Low voltage microgrid islanding through adaptive load shedding". In: Conference Proceedings 2017 17th IEEE International Conference on Environment and Electrical Engineering and 2017 1st IEEE Industrial and Commercial Power Systems Europe, EEEIC / I and CPS Europe 2017 (2017). DOI: 10.1109/EEEIC.2017. 7977861 (cit. on p. 19).
- [DW17] Hannah Daly and Molly Walton. WEO-2017 Special Report: Energy Access Outlook. Tech. rep. International Energy Agency, 2017. URL: https://www.iea. org/publications/freepublications/publication/ WE02017SpecialReport{_}EnergyAccessOutlook.pdf (cit. on p. 1).
- [DM13] Rai Dan and Mehran Mesbahi. "Optimal power generation and load management for off-grid hybrid power

systems with renewable sources via ..." In: *Energy Conversion and Management* 73.September 2013 (2013), pp. 234–244. DOI: 10.1016/j.enconman.2013.04.039 (cit. on pp. 15, 24).

- [Deb+17] Chirag Deb, Fan Zhang, Junjing Yang, Siew Eang Lee, and Kwok Wei Shah. "A review on time series forecasting techniques for building energy consumption". In: *Renewable* and Sustainable Energy Reviews 74.February (2017), pp. 902– 924. ISSN: 18790690. DOI: 10.1016/j.rser.2017.02.085. URL: http://dx.doi.org/10.1016/j.rser.2017.02.085 (cit. on p. 26).
- [DM17] Ghulam Mohi Ud Din and Angelos K. Marnerides. "Short term power load forecasting using Deep Neural Networks". In: 2017 International Conference on Computing, Networking and Communications, ICNC 2017 (2017), pp. 594–598. DOI: 10.1109/ICCNC.2017.7876196 (cit. on pp. 72, 95).
- [Dud16] Grzegorz Dudek. "Neural networks for pattern-based short-term load forecasting: A comparative study". In: *Neurocomputing* 205 (2016), pp. 64–74. ISSN: 18728286. DOI: 10.1016/j.neucom.2016.04.021. URL: http://dx.doi. org/10.1016/j.neucom.2016.04.021 (cit. on pp. 5, 24, 26).
- [EAR20] Javad Ebrahimi, Mohammad Abedini, and Mohammad Mahdi Rezaei. "Optimal scheduling of distributed generations in microgrids for reducing system peak load based on load shifting". In: Sustainable Energy, Grids and Networks 23 (2020), p. 100368. ISSN: 23524677. DOI: 10.1016/ j.segan.2020.100368. URL: https://doi.org/10.1016/ j.segan.2020.100368 (cit. on pp. 2, 19).
- [ESMK11] Mehdi Eghbal, Tapan Kumar Saha, and Nadali Mahmoudi-Kohan. "Utilizing demand response programs in day ahead generation scheduling for micro-grids with renewable sources". In: 2011 IEEE PES Innovative Smart Grid Technologies, ISGT Asia 2011 Conference: Smarter Grid for Sustainable and Affordable Energy Future (2011), pp. 1–6. DOI: 10.1109/ISGT-Asia.2011.6167148 (cit. on p. 24).
- [Ene12] Philippines Department of Energy. Summary Report : 2012 DOE Microgrid Workshop. 2012. URL: http://energy.gov/ sites/prod/files/2012MicrogridWorkshopReport09102012. pdf (cit. on pp. 1, 2, 11).

- [EPBN20] Danny Espín-Sarzosa, Rodrigo Palma-Behnke, and Oscar Núñez-Mata. "Energy management systems for microgrids: Main existing trends in centralized control architectures". In: *Energies* 13.3 (2020), pp. 1–32. ISSN: 19961073. DOI: 10.3390/en13030547 (cit. on p. 17).
- [FA14] Muhammad Usman Fahad and Naeem Arbab. "Factor Affecting Short Term Load Forecasting". In: *Journal of Clean Energy Technologies* 2.4 (2014), pp. 305–309. ISSN: 1793821X. DOI: 10.7763/jocet.2014.v2.145 (cit. on pp. 24, 25).
- [FHC16] Sarah Feron, Harald Heinrichs, and Raul R. Cordero.
 "Sustainability of rural electrification programs based on off-grid photovoltaic (PV) systems in Chile". In: *Energy, Sustainability and Society* 6.1 (2016). ISSN: 21920567. DOI: 10.1186/s13705-016-0098-4. URL: http://dx.doi.org/ 10.1186/s13705-016-0098-4 (cit. on p. 14).
- [Fra+11] D. Frame, K. Tembo, M. J. Dolan, S. M. Strachan, and G. W. Ault. "A community based approach for sustainable off-grid PV systems in developing countries". In: *IEEE Power and Energy Society General Meeting* (2011), pp. 1–6. ISSN: 19449925. DOI: 10.1109/PES.2011.6039593 (cit. on p. 14).
- [GZ16] Hossam A. Gabbar and Aboelsood Zidan. "Optimal scheduling of interconnected micro energy grids with multiple fuel options". In: Sustainable Energy, Grids and Networks 7 (2016), pp. 80–89. ISSN: 23524677. DOI: 10.1016/j.segan.2016.06.006. URL: http://dx.doi.org/10.1016/j.segan.2016.06.006 (cit. on p. 21).
- [GG15] Carlos Gamarra and Josep M. Guerrero. "Computational optimization techniques applied to microgrids planning: A review". In: *Renewable and Sustainable Energy Reviews* 48 (2015). ISSN: 18790690. DOI: 10.1016/j.rser.2015.04.025 (cit. on p. 16).
- [GFS12] M. Giroletti, M. Farina, and R. Scattolini. "A hybrid frequency/power based method for industrial load shedding". In: *International Journal of Electrical Power and Energy Systems* 35.1 (2012), pp. 194–200. ISSN: 01420615. DOI: 10.1016/j.ijepes.2011.10.013. URL: http://dx.doi.org/10.1016/j.ijepes.2011.10.013 (cit. on p. 20).

- [Gu+14] Wei Gu, Wei Liu, Junpeng Zhu, Bo Zhao, Zaijun Wu, Zhao Luo, and Jie Yu. "Adaptive decentralized underfrequency load shedding for Islanded smart distribution networks". In: *IEEE Transactions on Sustainable Energy* 5.3 (2014), pp. 886–895. ISSN: 19493029. DOI: 10.1109/TSTE. 2014.2310291 (cit. on pp. 13, 15, 19).
- [GS+16] O. Guerra-Santin, N. Romero Herrera, E. Cuerda, and D. Keyson. "Mixed methods approach to determine occupants' behaviour Analysis of two case studies". In: *Energy and Buildings* 130 (2016), pp. 546–566. ISSN: 03787788. DOI: 10.1016/j.enbuild.2016.08.084. URL: http://dx.doi.org/10.1016/j.enbuild.2016.08.084 (cit. on p. 32).
- [Gup17] Vikas Gupta. "An Overview of Different Types of Load Forecasting Methods and the Factors Affecting the Load Forecasting". In: *International Journal for Research in Applied Science and Engineering Technology* V.IV (2017), pp. 729– 733. DOI: 10.22214/ijraset.2017.4132 (cit. on pp. 18, 24, 25, 44).
- [HB12] Omar Hafez and Kankar Bhattacharya. "Optimal planning and design of a renewable energy based supply system for microgrids". In: *Renewable Energy* 45 (2012), pp. 7–15. ISSN: 09601481. DOI: 10.1016/j.renene.2012.01.087. URL: http://dx.doi.org/10.1016/j.renene.2012.01.087 (cit. on p. 12).
- [Han15] Vanessa Hans. Red Cross, partners to unveil village for Yolanda survivors. http://www.fpua.org.ph/2015/11/ 10/red - cross - partners - to - unveil - village - for yolanda-survivors/. 2015 (cit. on pp. 2, 3).
- [Har+15] Keishiro Hara, Michinori Uwasu, Yusuke Kishita, and Hiroyuki Takeda. "Determinant factors of residential consumption and perception of energy conservation: Time-series analysis by large-scale questionnaire in Suita, Japan". In: *Energy Policy* 87 (2015), pp. 240–249. ISSN: 03014215. DOI: 10.1016/j.enpol.2015.09.016. URL: http://dx.doi.org/10.1016/j.enpol.2015.09.016 (cit. on p. 6).
- [HD14] Nikos Hatziargyriou and Thomas Degner. Microgrid : architectures and control. 2014. ISBN: 9781118720684 1118720687 9781118720653 1118720652. URL: http: //site.ebrary.com/id/10815834 (cit. on p. 12).

- [Her+12] Luis Hernández, Carlos Baladrón, Javier M. Aguiar, Lorena Calavia, Belén Carro, Antonio Sánchez-Esguevillas, Diane J. Cook, David Chinarro, and Jorge Gómez. "A study of the relationship between weather variables and electric power demand inside a smart grid/smart world framework". In: Sensors (Switzerland) 12.9 (2012), pp. 11571–11591. ISSN: 14248220. DOI: 10.3390/s120911571 (cit. on p. 25).
- [HA12] George William Hong and Naoya Abe. "Sustainability assessment of renewable energy projects for off-grid rural electrification: The Pangan-an Island case in the Philippines". In: *Renewable and Sustainable Energy Reviews* 16.1 (2012), pp. 54–64. ISSN: 13640321. DOI: 10.1016/j.rser. 2011.07.136. URL: http://dx.doi.org/10.1016/j.rser. 2011.07.136 (cit. on p. 1).
- [HM12] R. Hooshmand and M. Moazzami. "Optimal design of adaptive under frequency load shedding using artificial neural networks in isolated power system". In: *International Journal of Electrical Power and Energy Systems* 42.1 (2012), pp. 220–228. ISSN: 01420615. DOI: 10.1016/j.ijepes.2012.04.021. URL: http://dx.doi.org/10.1016/j.ijepes.2012.04.021 (cit. on p. 20).
- [HSW89] Kurt Hornik, Maxwell Stinchcombe, and Halbert White. "Multilayer feedforward networks are universal approximators". In: *Neural networks* 2.5 (1989), pp. 359–366 (cit. on p. 27).
- [Hsi15] Yu-hsiang Hsiao. "Household Electricity Demand Forecast Based on Context Information and User Daily Schedule Analysis From Meter Data". In: *IEEE Transactions on Industrial Informatics* 11.1 (2015), pp. 33–43. DOI: 10.1109/ TII.2014.2363584. URL: http://ieeexplore.ieee.org/ document/6926785/ (cit. on pp. 6, 25, 31, 32, 44, 49, 72).
- [HCC11] Cheng Ting Hsu, Hui Jen Chuang, and Chao Shun Chen.
 "Adaptive load shedding for an industrial petroleum cogeneration system". In: *Expert Systems with Applications* 38.11 (2011), pp. 13967–13974. ISSN: 09574174. DOI: 10.1016/j.eswa.2011.04.204. URL: http://dx.doi.org/10.1016/j.eswa.2011.04.204 (cit. on p. 20).
- [Hu+17] Shan Hu, Da Yan, Siyue Guo, Ying Cui, and Bing Dong.
 "A survey on energy consumption and energy usage behavior of households and residential building in urban China". In: *Energy and Buildings* 148 (2017), pp. 366–378.

ISSN: 03787788. DOI: 10.1016/j.enbuild.2017.03.064. URL: http://dx.doi.org/10.1016/j.enbuild.2017.03. 064 (cit. on pp. 13, 69).

- [Hun+12] David Hunter, Hao Yu, Michael S. Pukish, Janusz Kolbusz, and Bogdan M. Wilamowski. "Selection of proper neural network sizes and architectures-A comparative study". In: *IEEE Transactions on Industrial Informatics* 8.2 (2012), pp. 228–240. ISSN: 15513203. DOI: 10.1109/TII. 2012.2187914 (cit. on p. 30).
- [IRE17] IRENA. Accelerating Renewable Mini-Grid Deployment : A study on the Philippines. 2017. ISBN: 9789292600303 (cit. on pp. 1, 4, 11, 12, 14, 18, 24).
- [IRE18] IRENA. "Off-grid renewable energy solutions". In: /publications/2018/Jul/Off-grid-Renewable-Energy-Solutions 2015 (2018), pp. 1–20. URL: http://www.irena.org/ publications/2018/Jul/Off-grid-Renewable-Energy-Solutions (cit. on pp. 13, 15, 23, 24).
- [Ima+18] Mahmood Hosseini Imani, M. Jabbari Ghadi, Sahand Ghavidel, and Li Li. "Demand Response Modeling in Microgrid Operation: a Review and Application for Incentive-Based and Time-Based Programs". In: *Renewable and Sustainable Energy Reviews* 94.June (2018), pp. 486–499. ISSN: 18790690. DOI: 10.1016/j.rser.2018.06.017 (cit. on pp. 18, 21).
- [Int14] International Renewable Energy Agency. *Renewables Readiness Assessment: The Philippines*. 2014, pp. 1–72.
 ISBN: 9789292600037. DOI: 10.1007/s11367-015-0960-1 (cit. on pp. 1, 14, 24).
- [JS05] Joanna Jedrzejowicz and Andrzej Szepietowski. Neural Information Processing. Ed. by Irwin King, Laiwan Chan, Jun Wang, and DeLiang Wang. Vol. 3618. Springer, 2005.
 ISBN: 9783540464846 (cit. on p. 28).
- [Jia+15] Linni Jian, Yanchong Zheng, Xinping Xiao, and C. C. Chan. "Optimal scheduling for vehicle-to-grid operation with stochastic connection of plug-in electric vehicles to smart grid". In: *Applied Energy* 146 (2015), pp. 150–161. ISSN: 03062619. DOI: 10.1016/j.apenergy.2015.02.030. URL: http://dx.doi.org/10.1016/j.apenergy.2015.02.030 (cit. on p. 21).

- [KB96] Iebeling Kaastraa and Milton Boydb. "Designing a neural network for forecasting financial time series 29". In: *Neurocomputing* 10.3 (1996), pp. 215–236. URL: https://doi. org/10.1016/0925-2312(95)00039-9 (cit. on p. 29).
- [KSB14] R. Kanimozhi, K. Selvi, and K. M. Balaji. "Multi-objective approach for load shedding based on voltage stability index consideration". In: *Alexandria Engineering Journal* 53.4 (2014), pp. 817–825. ISSN: 11100168. DOI: 10.1016/j. aej.2014.09.005. URL: http://dx.doi.org/10.1016/j. aej.2014.09.005 (cit. on p. 20).
- [KI06] F. Katiraei and M. R. Iravani. "Power Management Strategies for a Microgrid With Multiple Distributed Generation Units". In: *IEEE Transactions on Power Systems* 21.4 (2006), pp. 1821–1831. ISSN: 0885-8950. DOI: 10.1109/TPWRS.2006. 879260 (cit. on p. 12).
- [KRF13] Amir Kavousian, Ram Rajagopal, and Martin Fischer. "Determinants of residential electricity consumption: Using smart meter data to examine the effect of climate, building characteristics, appliance stock, and occupants' behavior". In: *Energy* 55 (2013), pp. 184–194. ISSN: 03605442. DOI: 10.1016/j.energy.2013.03.086. arXiv: arXiv: 1108.2234v1. URL: http://dx.doi.org/10.1016/j.energy.2013.03.086 (cit. on p. 25).
- [Kha+18] Aziah Khamis, Hussain Shareef, Azah Mohamed, and Zhao Yang Dong. "A load shedding scheme for DG integrated islanded power system utilizing backtracking search algorithm". In: *Ain Shams Engineering Journal* 9.1 (2018), pp. 161–172. ISSN: 20904479. DOI: 10.1016/j.asej. 2015.10.001. URL: https://doi.org/10.1016/j.asej. 2015.10.001 (cit. on p. 19).
- [Kip+20] Mark Kipngetich Kiptoo, Mohammed Elsayed Lotfy, Oludamilare Bode Adewuyi, Abdul Conteh, Abdul Motin Howlader, and Tomonobu Senjyu. "Integrated approach for optimal techno-economic planning for high renewable energy-based isolated microgrid considering cost of energy storage and demand response strategies". In: *Energy Conversion and Management* 215.October 2019 (2020), p. 112917. ISSN: 01968904. DOI: 10.1016/j.enconman.2020. 112917. URL: https://doi.org/10.1016/j.enconman.2020.
- [Kir+09] Charles Kirubi, Arne Jacobson, Daniel M. Kammen, and Andrew Mills. "Community-Based Electric Micro-Grids

Can Contribute to Rural Development: Evidence from Kenya". In: *World Development* 37.7 (2009), pp. 1208–1221. ISSN: 0305750X. DOI: 10.1016/j.worlddev.2008.11.005. URL: http://dx.doi.org/10.1016/j.worlddev.2008.11. 005 (cit. on p. 13).

- [KK15] Toru Kobayakawa and Tara C. Kandpal. "Analysis of electricity consumption under a photovoltaic micro-grid system in India". In: *Solar Energy* 116 (2015), pp. 177–183. ISSN: 0038092X. DOI: 10.1016/j.solener.2015.04.001. URL: http://dx.doi.org/10.1016/j.solener.2015.04.001 (cit. on p. 16).
- [Koh95] Ron Kohavi. "A Study of Cross-Validation and Bootstrap for Accuracy Estimation and Model Selection". In: International Joint Conference of Artificial Intelligence June (1995) (cit. on p. 75).
- [Kon+17] Weicong Kong, Zhao Yang Dong, David J. Hill, Fengji Luo, and Yan Xu. "Short-Term Residential Load Forecasting based on Resident Behaviour Learning". In: *IEEE Transactions on Power Systems* 33.1 (2017), pp. 1–1. ISSN: 0885-8950. DOI: 10.1109/TPWRS.2017.2688178. arXiv: TSG.2017.2753802 [10.1109]. URL: http://ieeexplore. ieee.org/document/7887751/ (cit. on pp. 25, 32, 33).
- [Kua+16] K. Kuang, Y. Zhang, B. Zhou, C. Li, Y. Cao, and L. Li.
 "A review of renewable energy utilization in islands". In: *Renewable Sustainable Energy Review* 59 (2016), pp. 504–513 (cit. on p. 12).
- [LBH15] Yann Lecun, Yoshua Bengio, and Geoffrey Hinton. "Deep learning". In: *Nature* 521.7553 (2015), pp. 436–444. ISSN: 14764687. DOI: 10.1038/nature14539 (cit. on pp. 27, 28).
- [LC16] Dasheng Lee and Chin Chi Cheng. "Energy savings by energy management systems: A review". In: *Renewable and Sustainable Energy Reviews* 56 (2016), pp. 760–777. ISSN: 18790690. DOI: 10.1016/j.rser.2015.11.067 (cit. on p. 13).
- [LL18] Xiao Li and Zhijun Li. "Micro-grid resource allocation based on multi-objective optimization in cloud platform". In: Proceedings of the IEEE International Conference on Software Engineering and Service Sciences, ICSESS 2017-November (2018), pp. 509–512. ISSN: 23270594. DOI: 10.1109/ICSESS.2017.8342966 (cit. on p. 13).

- [LLC09] Chi Jie Lu, Tian Shyug Lee, and Chih Chou Chiu. "Financial time series forecasting using independent component analysis and support vector regression". In: *Decision Support Systems* 47.2 (2009), pp. 115–125. ISSN: 01679236. DOI: 10.1016/j.dss.2009.02.001. URL: http://dx.doi.org/10.1016/j.dss.2009.02.001 (cit. on p. 26).
- [MS97] Saumen Majumdar and P. R. Shukla. "Electric load fore-casting using artificial neural networks and deficit management". In: *Energy Sources* 19.8 (1997), pp. 771–782. ISSN: 00908312. DOI: 10.1080/00908319708908889 (cit. on p. 26).
- [Man+o6] Paras Mandal, Tomonobu Senjyu, Naomitsu Urasaki, and Toshihisa Funabashi. "A neural network based severalhour-ahead electric load forecasting using similar days approach". In: 28 (2006), pp. 367–373. DOI: 10.1016/j. ijepes.2005.12.007 (cit. on pp. 32, 49).
- [Man+16] Stefano Mandelli, Jacopo Barbieri, Riccardo Mereu, and Emanuela Colombo. "Off-grid systems for rural electrification in developing countries: Definitions, classification and a comprehensive literature review". In: *Renewable* and Sustainable Energy Reviews 58 (2016), pp. 1621–1646. ISSN: 13640321. DOI: 10.1016/j.rser.2015.12.338. URL: http://linkinghub.elsevier.com/retrieve/pii/ S1364032115017219 (cit. on p. 14).
- [Mar+13] Mousa Marzband, Andreas Sumper, Albert Ruiz-álvarez, José Luis Domínguez-García, and Bogdan Tomoiagâ. "Experimental evaluation of a real time energy management system for stand-alone microgrids in day-ahead markets". In: *Applied Energy* 106 (2013), pp. 365–376. ISSN: 03062619. DOI: 10.1016/j.apenergy.2013.02.018 (cit. on pp. 2, 22).
- [Mar+16] Mousa Marzband, Ebrahim Yousefnejad, Andreas Sumper, and José Luis Domínguez-García. "Real time experimental implementation of optimum energy management system in standalone Microgrid by using multi-layer ant colony optimization". In: *International Journal of Electrical Power and Energy Systems* 75 (2016), pp. 265–274. ISSN: 01420615. DOI: 10.1016/j.ijepes.2015.09.010. URL: http://dx.doi.org/10.1016/j.ijepes.2015.09.010 (cit. on pp. 2, 22).

- [Mas43] Abraham. H. Maslow. "A theory of human motivation." In: *Psychological Review* 50.4 (1943), pp. 370–396. DOI: 10. 1037/h0054346 (cit. on p. 49).
- [MMS16a] Mohammadreza Mazidi, Hassan Monsef, and Pierluigi Siano. "Incorporating price-responsive customers in dayahead scheduling of smart distribution networks". In: *Energy Conversion and Management* 115 (2016), pp. 103–116. ISSN: 01968904. DOI: 10.1016/j.enconman.2016.02.030. URL: http://dx.doi.org/10.1016/j.enconman.2016.02. 030 (cit. on p. 22).
- [MMS16b] Mohammadreza Mazidi, Hassan Monsef, and Pierluigi Siano. "Robust day-ahead scheduling of smart distribution networks considering demand response programs". In: Applied Energy 178 (2016), pp. 929–942. ISSN: 03062619. DOI: 10.1016/j.apenergy.2016.06.016. URL: http: //dx.doi.org/10.1016/j.apenergy.2016.06.016 (cit. on p. 21).
- [Maz+14] Mohammadreza Mazidi, Alireza Zakariazadeh, Shahram Jadid, and Pierluigi Siano. "Integrated scheduling of renewable generation and demand response programs in a microgrid". In: *Energy Conversion and Management* 86 (2014), pp. 1118–1127. ISSN: 01968904. DOI: 10.1016/j. enconman.2014.06.078. URL: http://dx.doi.org/10. 1016/j.enconman.2014.06.078 (cit. on pp. 19, 21, 24).
- [Men15a] Lexuan Meng. Hierarchical Control for Optimal and Distributed Operation of Microgrid Systems. Aalborg University Press, 2015. ISBN: 9788792846662 (cit. on pp. 13, 16).
- [Men15b] Lexuan Meng. Hierarchical Control for Optimal and Distributed Operation of Microgrid Systems. Aalborg University Press, 2015. ISBN: 9788792846662 (cit. on p. 16).
- [Men+15] Lexuan Meng, Tomislav Dragicevic, Javier Roldan-Perez, Juan C. Vasquez, and Josep M. Guerrero. "Modeling and Sensitivity Study of Consensus Algorithm-Based Distributed Hierarchical Control for DC Microgrids". In: *IEEE Transactions on Smart Grid* (2015), pp. 1–12. ISSN: 19493053. DOI: 10.1109/TSG.2015.2422714 (cit. on p. 15).
- [Men+16] Lexuan Meng, Xin Zhao, Fen Tang, Mehdi Savaghebi, Tomislav Dragicevic, Juan C. Vasquez, and Josep M. Guerrero. "Distributed Voltage Unbalance Compensation in Islanded Microgrids by Using a Dynamic Consensus Algorithm". In: IEEE Transactions on Power Electronics 31.1

(2016), pp. 827–838. ISSN: 08858993. DOI: 10.1109/TPEL. 2015.2408367 (cit. on p. 18).

- [Mil+15] Clark A Miller, Carlo Altamirano-Allende, Nathan Johnson, and Malena Agyemang. "The social value of midscale energy in Africa: Redefining value and redesigning energy to reduce poverty". In: *Energy Research and Social Science* 5 (2015), pp. 67–69. ISSN: 22146296. DOI: 10.1016/j.erss.2014.12.013. URL: http://dx.doi.org/10.1016/j.erss.2014.12.013 (cit. on p. 15).
- [Moh+18] H. Mohamad, A. I.Md Isa, Z. Mat Yasin, N. A. Salim, and N. N.A.Mohd Rahim. "Optimal load shedding technique for an islanding distribution system by using Particle Swarm Optimization". In: 3rd International Conference on Power Generation Systems and Renewable Energy Technologies, PGSRET 2017 2018-January (2018), pp. 154–158. DOI: 10.1109/PGSRET.2017.8251819 (cit. on pp. 13, 20).
- [Moo+19] Jihoon Moon, Sungwoo Park, Seungmin Rho, and Eenjun Hwang. "A comparative analysis of artificial neural network architectures for building energy consumption forecasting". In: *International Journal of Distributed Sensor Networks* 15.9 (2019), p. 155014771987761. ISSN: 1550-1477. DOI: 10.1177/1550147719877616 (cit. on pp. 5, 25, 32, 49).
- [Mor+16] Javier Moriano, Francisco Javier Rodríguez, Pedro Martín, Jose Antonio Jiménez, and Branislav Vuksanovic. "A new approach to detection of systematic errors in secondary substation monitoring equipment based on short term load forecasting". In: *Sensors (Switzerland)* 16.1 (2016), pp. 1–14. ISSN: 14248220. DOI: 10.3390/s16010085 (cit. on pp. 6, 72).
- [MSS20] Roozbeh Morsali, Gokul Sidarth, and Mehdi Seyedmahmoudian. "A relaxed constrained decentralised demand side management system of a community-based residential microgrid with realistic appliance models". In: *Applied Energy* 277.August (2020), p. 115626. ISSN: 0306-2619. DOI: 10.1016/j.apenergy.2020.115626. URL: https: //doi.org/10.1016/j.apenergy.2020.115626 (cit. on p. 13).
- [Nas+14] Vahidreza Nasirian, Student Member, Seyedali Moayedi, Student Member, and Ali Davoudi. "Distributed Cooperative Control of DC Microgrids". In: 8993.4 (2014), pp. 1– 29. ISSN: 0885-8993. DOI: 10.1109/TPEL.2014.2324579 (cit. on p. 18).

- [Och13] Alexander Ochs. Combatting Climate Change & Energy Poverty in the Philippines Current Situation : Key Problems. Tech. rep. June. 2013 (cit. on p. iii).
- [Oli+14] Daniel E. Olivares et al. "Trends in microgrid control". In: *IEEE Transactions on Smart Grid* 5.4 (2014), pp. 1905–1919.
 ISSN: 19493053. DOI: 10.1109/TSG.2013.2295514 (cit. on pp. 12, 15).
- [Pal16] Gene Fe P Palencia. "Off-grid Communities : Deployment Report". In: (2016), pp. 1–9 (cit. on p. 11).
- [Pal19] Gene Fe Palencia. PMOG (Enery) and Survey Data. http: //cogentee.coventry.ac.uk/~gene/researchdata/. Version 1.0. 2019 (cit. on pp. 59, 60, 151).
- [PJSo6] Alfonso Palmer, Juan José Montaño, and Albert Sesé.
 "Designing an artificial neural network for forecasting tourism time series". In: *Tourism Management* 27.5 (2006), pp. 781–790. ISSN: 02615177. DOI: 10.1016/j.tourman. 2005.05.006 (cit. on pp. 30, 73).
- [PB17] Sibarama Panigrahi and H S Behera. "Engineering Applications of Artificial Intelligence A hybrid ETS ANN model for time series forecasting". In: *Engineering Applications of Artificial Intelligence* 66.September (2017), pp. 49–59.
 ISSN: 0952-1976. DOI: 10.1016/j.engappai.2017.07.007.
 URL: http://dx.doi.org/10.1016/j.engappai.2017.07.
 007 (cit. on p. 26).
- [Par+91] D. C. Park, M. A. El-Sharkawi, R. J. Marks, L. E. Atlas, and M. J. Damborg. "Electric load forecasting using an artificial neural network". In: *IEEE Transactions on Power Systems* 6.2 (1991), pp. 442–449. ISSN: 0885-8950. DOI: 10. 1109/59.76685 (cit. on p. 26).
- [Per+12] Emanuele Graciosa Pereira, Jadir Nogueira Da Silva, Jofran L. De Oliveira, and Cassio S. MacHado. "Sustainable energy: A review of gasification technologies". In: *Renewable and Sustainable Energy Reviews* 16.7 (2012), pp. 4753–4762. ISSN: 13640321. DOI: 10.1016/j.rser. 2012.04.023 (cit. on p. 14).
- [Phi13] Philippines Statistics Authority. Electricity is the most common source of energy used by households. 2013. URL: https: //psa.gov.ph/content/electricity-most-commonsource-energy-used-households (cit. on p. 1).

- [PNS15] S. Ali Pourmousavi, M. Hashem Nehrir, and Ratnesh K. Sharma. "Multi-Timescale Power Management for Islanded Microgrids Including Storage and Demand Response". In: *IEEE Transactions on Smart Grid* 6.3 (2015), pp. 1185–1195. ISSN: 19493053. DOI: 10.1109/TSG.2014.2387068 (cit. on p. 18).
- [RKS95] Heggere S. Ranganath, Derek E. Kerstetter, and S. Richard F Sims. "Self partitioning neural networks for target recognition". In: *Neural Networks* 8.9 (1995), pp. 1475–1486. ISSN: 08936080. DOI: 10.1016/0893-6080(95)00087-9 (cit. on p. 29).
- [RK15] Muhammad Qamar Raza and Abbas Khosravi. "A review on artificial intelligence based load demand forecasting techniques for smart grid and buildings". In: *Renewable* and Sustainable Energy Reviews 50 (2015), pp. 1352–1372. ISSN: 18790690. DOI: 10.1016/j.rser.2015.04.065. URL: http://dx.doi.org/10.1016/j.rser.2015.04.065 (cit. on pp. 24, 29, 30).
- [RM+14] Rémy Rigo-Mariani, Bruno Sareni, Xavier Roboam, and Christophe Turpin. "Optimal power dispatching strategies in smart-microgrids with storage". In: *Renewable and Sustainable Energy Reviews* 40 (2014), pp. 649–658. ISSN: 13640321. DOI: 10.1016/j.rser.2014.07.138. URL: http: //dx.doi.org/10.1016/j.rser.2014.07.138 (cit. on p. 16).
- [Rol11] Simon Rolland. Rural Electrification with Renewable Energy Technologies: Quality Standards and Business Models. Tech. rep. Alliance for Rural Electrification, 2011, pp. 1–56. URL: https://www.ruralelec.org/sites/default/files/ are{_}technological{_}publication{_}0.pdf (cit. on p. 13).
- [RS16] Fernando Roxas and Andrea Santiago. "Alternative framework for renewable energy planning in the Philippines". In: *Renewable and Sustainable Energy Reviews* 59 (2016), pp. 1396–1404. ISSN: 13640321. DOI: 10.1016/j.rser.2016.01.084. URL: http://linkinghub. elsevier . com / retrieve / pii / S1364032116001143 (cit. on p. 1).
- [RNK16] Seunghyoung Ryu, Jaekoo Noh, and Hongseok Kim. "Deep neural network based demand side short term load forecasting". In: 2016 IEEE International Conference on Smart Grid Communications, SmartGrid-

Comm 2016 (2016), pp. 308–313. ISSN: 19961073. DOI: 10.1109/SmartGridComm.2016.7778779 (cit. on pp. 72, 94, 95).

- [SSP11] Alireza Saffarian and Majid Sanaye-Pasand. "Enhancement of power system stability using adaptive combinational load shedding methods". In: *IEEE Transactions on Power Systems* 26.3 (2011), pp. 1010–1020. ISSN: 08858950.
 DOI: 10.1109/TPWRS.2010.2078525 (cit. on pp. 13, 20).
- [San+19] Athila Quaresma Santos, Renato Machado Monaro, Denis Vinicius Coury, and Mario Oleskovicz. "A new real-time multi-agent system for under frequency load shedding in a smart grid context". In: *Electric Power Systems Research* 174.October 2018 (2019), p. 105851. ISSN: 03787796. DOI: 10.1016/j.epsr.2019.04.029. URL: https://doi.org/ 10.1016/j.epsr.2019.04.029 (cit. on p. 20).
- [Sap+18] N. M. Sapari, H. Mokhlis, Javed Ahmed Laghari, A. H.A. Bakar, and M. R.M. Dahalan. "Application of load shedding schemes for distribution network connected with distributed generation: A review". In: *Renewable and Sustainable Energy Reviews* 82.September 2017 (2018), pp. 858–867. ISSN: 18790690. DOI: 10.1016/j.rser.2017.09.090. URL: http://dx.doi.org/10.1016/j.rser.2017.09.090 (cit. on pp. 14, 19).
- [SKN11] Martina Schafer, Noara Kebir, and Kirsten Neumann. "Research needs for meeting the challenge of decentralized energy supply in developing countries". In: *Energy for Sustainable Development* 15.3 (2011), pp. 324–329. ISSN: 09730826. DOI: 10.1016/j.esd.2011.07.001. URL: http: //dx.doi.org/10.1016/j.esd.2011.07.001 (cit. on p. 14).
- [Sch+14] Daniel Schnitzer, Deepa Shinde Lounsbury, Juan Pablo Carvallo, Ranjit Deshmukh, Jay Apt, and Daniel M Kammen. *Microgrids for Rural Electrification : A critical review* of best practices based on seven case studies. Tech. rep. 2014 (cit. on p. 14).
- [Shi+15] Wenbo Shi, Xiaorong Xie, Chi Cheng Chu, and Rajit Gadh.
 "Distributed Optimal Energy Management in Microgrids". In: *IEEE Transactions on Smart Grid* 6.3 (2015), pp. 1137–1146. ISSN: 19493053. DOI: 10.1109/TSG.2014.2373150 (cit. on pp. 15, 16, 18).

- [Shu+19] H. Shuai, J. Fang, X. Ai, J. Wen, and H. He. "Optimal Real-Time Operation Strategy for Microgrid: An ADP-Based Stochastic Nonlinear Optimization Approach". In: *IEEE Transactions on Sustainable Energy* 10.2 (2019), pp. 931–942.
 ISSN: 1949-3029. DOI: 10.1109/TSTE.2018.2855039 (cit. on p. 13).
- [SW12] Ango Sobu and Guohong Wu. "Optimal operation planning method for isolated micro grid considering uncertainties of renewable power generations and load demand". In: 2012 IEEE Innovative Smart Grid Technologies -Asia, ISGT Asia 2012 (2012), pp. 1–6. DOI: 10.1109/ISGT-Asia.2012.6303271 (cit. on p. 15).
- [Sol+15] Bharatkumar V. Solanki, Akash Raghurajan, Kankar Bhattacharya, and Claudio A. Ca??izares. "Including smart loads for optimal demand response in integrated energy management systems for isolated microgrids". In: *IEEE Transactions on Smart Grid* (2015). ISSN: 19493053. DOI: 10.1109/TSG.2015.2506152 (cit. on p. 25).
- [Sun+16] D. B. Sun, H. Xiao, X. Wang, J. J. Liu, X. Wang, X. Q. Jin, J. Wang, and X. K. Xie. "Energy consumption quota management of Wanda commercial buildings in China". In: *IOP Conference Series: Earth and Environmental Science* 40.1 (2016). ISSN: 17551315. DOI: 10.1088/1755-1315/40/ 1/012014 (cit. on pp. 2, 22).
- [SH17] Kaiyu Sun and Tianzhen Hong. "A simulation approach to estimate energy savings potential of occupant behavior measures". In: *Energy and Buildings* 136 (2017), pp. 43–62.
 ISSN: 03787788. DOI: 10.1016/j.enbuild.2016.12.010.
 URL: http://dx.doi.org/10.1016/j.enbuild.2016.12.010
 010 (cit. on p. 32).
- [TJ16] V. Tamilselvan and T. Jayabarathi. "A hybrid method for optimal load shedding and improving voltage stability". In: *Ain Shams Engineering Journal* 7.1 (2016), pp. 223–232. ISSN: 20904479. DOI: 10.1016/j.asej.2015.11.003. URL: http://dx.doi.org/10.1016/j.asej.2015.11.003 (cit. on p. 20).
- [Tan+15] Jie Tang, Daniel K C So, Senior Member, Emad Alsusa, Senior Member, Khairi Ashour Hamdi, Senior Member, and Arman Shojaeifard. "Resource Allocation for Energy Efficiency Optimization in Heterogeneous Networks". In: *IEEE Journal on Selected Areas in Communications* 33.10 (2015), pp. 2104–2117 (cit. on p. 132).

- [Tan+13] Junjie Tang, Junqi Liu, Ferdinanda Ponci, and Antonello Monti. "Adaptive load shedding based on combined frequency and voltage stability assessment using synchrophasor measurements". In: *IEEE Transactions on Power Systems* 28.2 (2013), pp. 2035–2047. ISSN: 08858950. DOI: 10.1109/TPWRS.2013.2241794 (cit. on p. 20).
- [TS16] Akin Tascikaraoglu and Borhan M Sanandaji. "Shortterm residential electric load forecasting : A compressive spatio-temporal approach". In: *Energy & Buildings* 111 (2016), pp. 380–392. ISSN: 0378-7788. DOI: 10.1016/ j.enbuild.2015.11.068. URL: http://dx.doi.org/10. 1016/j.enbuild.2015.11.068 (cit. on p. 25).
- [Tay10] James W. Taylor. "Triple seasonal methods for short-term electricity demand forecasting". In: European Journal of Operational Research 204.1 (2010), pp. 139–152. ISSN: 03772217. DOI: 10.1016/j.ejor.2009.10.003. URL: http://dx.doi. org/10.1016/j.ejor.2009.10.003 (cit. on p. 26).
- [TMM06] James W. Taylor, Lilian M. de Menezes, and Patrick E. McSharry. "A comparison of univariate methods for forecasting electricity demand up to a day ahead". In: *International Journal of Forecasting* 22.1 (2006), pp. 1–16. ISSN: 01692070. DOI: 10.1016/j.ijforecast.2005.06.006 (cit. on p. 26).
- [Tero6] V. V. Terzija. "Adaptive underfrequency load shedding based on the magnitude of the disturbance estimation". In: *IEEE Transactions on Power Systems* 21.3 (2006), pp. 1260–1266. ISSN: 0885-8950. DOI: 10.1109/TPWRS.2006.879315 (cit. on p. 20).
- [Tia+14] Hao Tian, Xiaohui Yuan, Bin Ji, and Zhihuan Chen. "Multi-objective optimization of short-term hydrothermal scheduling using non-dominated sorting gravitational search algorithm with chaotic mutation". In: *Energy Conversion and Management* 81 (2014), pp. 504–519. ISSN: 01968904. DOI: 10.1016/j.enconman.2014.02.053. URL: http://dx.doi.org/10.1016/j.enconman.2014.02.053 (cit. on p. 132).
- [Tia18] Zhao Tianyang. Energy management systems for hybrid AC/DC microgrids. https://github.com/Matrixeigs/ energy_management_systeme. 2018 (cit. on p. 33).
- [Uls+15] Kirsten Ulsrud, Tanja Winther, Debajit Palit, and Harald Rohracher. "Village-level solar power in Africa: Accelerating access to electricity services through a socio-technical

design in Kenya". In: *Energy Research and Social Science* 5 (2015), pp. 34–44. ISSN: 22146296. DOI: 10.1016/j.erss. 2014.12.009. URL: http://dx.doi.org/10.1016/j.erss. 2014.12.009 (cit. on p. 15).

- [Uni12] United Nations Development Programme. *Towards an* 'Energy Plus' Approach for the Poor: An Agenda for Action for Asia and the Pacific. 2012, p. 112. ISBN: 9789746802994 (cit. on p. 2).
- [Wel+17] Shirantha Welikala, Chinthaka Dinesh, Mervyn Parakrama B. Ekanayake, Roshan Indika Godaliyadda, and Janaka Ekanayake. "Incorporating Appliance Usage Patterns for Non-Intrusive Load Monitoring and Load Forecasting". In: *IEEE Transactions on Smart Grid* (2017), pp. 1–14. ISSN: 19493053. DOI: 10.1109/TSG.2017.2743760 (cit. on p. 25).
- [Rto] What is R? https://www.r-project.org/about.html. Accessed: 10th May 2020 (cit. on p. 59).
- [Xin+12] Yajuan Xin, Shilei Lu, Neng Zhu, and Wei Wu. "Energy consumption quota of four and five star luxury hotel buildings in Hainan province, China". In: *Energy and Buildings* 45 (2012), pp. 250–256. ISSN: 03787788. DOI: 10. 1016/j.enbuild.2011.11.014. URL: http://dx.doi.org/ 10.1016/j.enbuild.2011.11.014 (cit. on pp. 23, 31, 34).
- [Xu+17] Xin Xu, Hengxu Zhang, Changgang Li, Yutian Liu, Wei Li, and Vladimir Terzija. "Optimization of the Event-Driven Emergency Load-Shedding Considering Transient Security and Stability Constraints". In: *IEEE Transactions on Power Systems* 32.4 (2017), pp. 2581–2592. ISSN: 08858950. DOI: 10.1109/TPWRS.2016.2619364 (cit. on p. 12).
- [Xu17] Y. Xu. "Robust Finite-Time Control for Autonomous Operation of an Inverter-Based Microgrid". In: *IEEE Transactions on Industrial Informatics* 13.5 (2017), pp. 2717–2725.
 ISSN: 1551-3203. DOI: 10.1109/TII.2017.2693233 (cit. on p. 13).
- [Xu+16] Yin Xu, Chen-Ching Liu, Ying Chen, and Haixiang Gao. "Dynamic load shedding for an islanded microgrid with limited generation resources". In: *IET Generation, Transmission & Distribution* 10.12 (2016), pp. 2953–2961. ISSN: 1751-8687. DOI: 10.1049/iet-gtd.2015.1452. URL: http: //digital-library.theiet.org/content/journals/10. 1049/iet-gtd.2015.1452 (cit. on pp. 2, 18, 20).

- [ZJS14] Alireza Zakariazadeh, Shahram Jadid, and Pierluigi Siano. "Smart microgrid energy and reserve scheduling with demand response using stochastic optimization". In: *International Journal of Electrical Power and Energy Systems* 63 (2014), pp. 523–533. ISSN: 01420615. DOI: 10.1016/j. ijepes.2014.06.037. URL: http://dx.doi.org/10.1016/ j.ijepes.2014.06.037 (cit. on p. 16).
- [ZDM12] Daniel Zelazo, Ran Dai, and Mehran Mesbahi. "An energy management system for off-grid power systems". In: *Energy Systems* 3.2 (2012), pp. 153–179. ISSN: 18683967. DOI: 10.1007/s12667-012-0050-4 (cit. on pp. 12, 15, 16).
- [ZB16] By Bowen Zhang and John Baillieul. "Control and Communication Protocols Based on Packetized Direct Load Control in Smart Building Microgrids". In: 104.4 (2016) (cit. on p. 12).
- [Zha+18] Cuo Zhang, Yan Xu, Zhao Yang Dong, and Kit Po Wong.
 "Robust Coordination of Distributed Generation and Price-Based Demand Response in Microgrids". In: *IEEE Transactions on Smart Grid* 9.5 (2018), pp. 4236–4247. ISSN: 19493053. DOI: 10.1109/TSG.2017.2653198 (cit. on p. 18).
- [Zha+16] Jingrui Zhang, Yihong Wu, Yiran Guo, Bo Wang, Hengyue Wang, and Houde Liu. "A hybrid harmony search algorithm with differential evolution for day-ahead scheduling problem of a microgrid with consideration of power flow constraints". In: *Applied Energy* 183 (2016), pp. 791–804. ISSN: 03062619. DOI: 10.1016/j.apenergy.2016.09.035. URL: http://dx.doi.org/10.1016/j.apenergy.2016.09.035 (cit. on p. 21).
- [ZXT12] Jing Zhao, Yajuan Xin, and Dingding Tong. "Energy consumption quota of public buildings based on statistical analysis". In: *Energy Policy* 43 (2012), pp. 362–370. ISSN: 03014215. DOI: 10.1016/j.enpol.2012.01.015. URL: http://dx.doi.org/10.1016/j.enpol.2012.01.015 (cit. on pp. 2, 13, 19, 23, 24).
- [Zhe+11] J. H. Zheng, Y. Wang, Z. Wang, S. Zhu, X. Wang, and Shen Xinwei. "Study on microgrid operation modes switching based on eigenvalue analysis". In: 2011 International Conference on Advanced Power System Automation and Protection. Vol. 1. 2011, pp. 445–450. DOI: 10.1109/APAP.2011.
 6180443 (cit. on p. 12).